# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

#### CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report: July 28, 2004
Date of Earliest Event Reported: July 26, 2004

# **BOISE CASCADE CORPORATION**

(Exact name of registrant as specified in its charter)

Delaware1-505782-0100960(State or other jurisdiction of incorporation or organization)(Commission File incorporation or organization)(I.R.S. Employer identification No.)

1111 West Jefferson Street
P.O. Box 50
Boise, Idaho
(Address of principal executive offices)

**83728-0001** (Zip Code)

#### 208/384-6161

(Registrant's telephone number, including area code)

#### Item 5. Other Events.

On July 26, 2004, Boise Cascade Corporation announced that it had reached a definitive agreement to sell its paper, forest products, and timberland assets for approximately \$3.7 billion to Madison Dearborn Partners, Inc., a private equity investment firm located in Chicago, Illinois.

A copy of the Asset Purchase Agreement dated July 26, 2004, is attached as Exhibit 2.

The foregoing description of the transaction is not complete. For further information, please refer to the Asset Purchase Agreement that is filed with this Form 8-K.

# Item 7. <u>Financial Statements and Exhibits.</u>

(c) Exhibits.

Exhibit 2 Asset Purchase Agreement dated July 26, 2004, between Boise Cascade Corporation, Boise Southern Company, Minidoka Paper

Company and Forest Products Holdings, L.L.C., and Boise Land & Timber Corp.

Exhibit 99 Transcript of July 26, 2004, Webcast and Conference Call

# Item 9. <u>Regulation FD Disclosure.</u>

Boise hosted a webcast and conference call on Monday, July 26, 2004, at 10 a.m. (Eastern Daylight Time) at which it discussed this announcement. A transcript of the webcast and conference call is attached as Exhibit 99.

### SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

BOISE CASCADE CORPORATION

By <u>/s/ Karen E. Gowland</u>

Karen E. Gowland

Vice President and Corporate Secretary

Date: July 28, 2004

# EXHIBIT INDEX

Number	Description
2	Asset Purchase Agreement dated July 26, 2004, between Boise Cascade Corporation, Boise Southern Company, Minidoka Paper Company and Forest Products Holdings, L.L.C., and Boise Land & Timber Corp.
99	Transcript of July 26, 2004, Webcast and Conference Call
	3

#### ASSET PURCHASE AGREEMENT

This Agreement is made and entered into as of this 26th day of July, 2004, by and among Boise Cascade Corporation, a Delaware corporation ("Parent"), Boise Southern Company, a Louisiana general partnership ("Southern"), Minidoka Paper Company, a Delaware corporation ("Minidoka") (collectively, "Sellers" and, individually, a "Seller") and Forest Products Holdings, L.L.C., a Delaware limited liability company ("Holdings") and Boise Land & Timber Corp., a Delaware corporation and indirect wholly-owned Subsidiary of Holdings ("Timber Co.").

#### RECITALS

WHEREAS Sellers own, directly or through certain Subsidiaries and Affiliates, certain business operations in which they own and operate timberlands, convert timber produced on such lands and timber grown on the lands of others to paper products, packaging and newsprint products and building products and distribute paper products, packaging and newsprint products and building products;

WHEREAS Holdings and certain Permitted Affiliate Purchasers have agreed with Sellers to purchase the paper operations, packaging and newsprint operations and building products manufacturing and distribution operations of Parent and its Subsidiaries, including their headquarters operations which manage and administer their paper, packaging and newsprint, building products manufacturing and distribution, and timberlands businesses; and

WHEREAS Timber Co. has agreed with certain of the Sellers to purchase the Timberlands (as hereinafter defined) and such Sellers have agreed to accept the Timber Notes in partial payment of the purchase price for the Timberlands;

NOW THEREFORE, the parties do hereby agree as follows:

# Article 1 Defined Terms

- 1.1 <u>Definitions</u>. The terms set forth below shall have the meaning indicated below whenever used in this agreement:
- "Accounting Principles" means GAAP, applied in a manner consistent with the accounting practices and procedures used in the preparation of the Historical Financial Statements.
- "Accounts Receivable" means the obligations to make payment to any of Sellers or any of their Retained Subsidiaries by all Persons who have purchased products or services in the operation of the Business prior to Closing, including Trade Intercompany Accounts. Accounts Receivable does not include any such receivables which have been sold by Parent to its receivables

securitization facility; provided that Parent shall cease making sales of its receivables generated by the Business to its receivables securitization facility not less than 45 days prior to the date set for Closing.

- "ABO" shall have the meaning given it in Section 11.4.2.1.
- "Accumulated Benefit Obligation" means such term as defined under Statement of Financial Accounting Standards No. 87.
- "Additional Consideration Agreement" means the agreement bearing such title between Parent and Holdings provided for in Section 3.5.
- "Additional Financial Statements" shall have the meaning given it in Section 10.13.
- "Affiliates" means, with respect to any Person, any Persons directly or indirectly controlling, controlled by or under common control with, such other Person as of the date on which, or at any time during the period for which, the determination of affiliation is being made. For the purpose of this definition, "control" means (i) the ownership or control of 50% or more of the equity interest in any Person, or (ii) the ability to direct or cause the direction of the management or affairs of a Person, whether through the direct or indirect ownership of voting interests, by contract or otherwise.
- "Allocation" shall have the meaning given it in Section 3.4.
- "Amended Paper Sales Agreement" means the agreement between Parent (to be assigned to Holdings at Closing) and Boise Office Solutions to be entered into as of the Closing pursuant to Section 18.2.
- "Ancillary Agreements" means, with respect to any party, each additional agreement which such party or any of such party's Subsidiaries is obligated to enter into pursuant to the terms of this agreement, including, without limitation, the Shareholder Agreements, the Additional Consideration Agreement, the Amended Paper Sales Agreement, the Mutual Administrative Services Agreement, the Timber Notes, and the Aviation Services Agreements.
- "Assets" means all of the assets owned (including all leasehold and licensee interests in any assets leased or licensed) by any of Sellers or any of the Retained Subsidiaries and (a) located at any of the Facilities or the Timberlands, or (b) exclusively or primarily related to, or used exclusively or primarily in, the conduct of the Business, whether tangible or intangible, real or personal, including the Transferred Cash, the Realty, Timberlands, Leases, Fixtures and Equipment, Inventory, Accounts Receivable, Contracts, Permits and Licenses, Parent Intellectual Property, IP Licenses, the Boise Marks, Transferred Records, stock of the Transferred Subsidiaries, and Miscellaneous Assets. Notwithstanding the foregoing, the "Assets" do not include any of the Retained Assets.
- "Assumed Liabilities" means the liabilities and obligations of Sellers to be assumed by Holdings or a Permitted Affiliate Purchaser pursuant to Section 5.1.
- "Audited Financial Statements" shall have the meaning given it in Section 10.13.

"Aviation Services Agreements" means the agreements between Holdings and Parent relating to operation and cost sharing for the Business's aviation operations after Closing, in form and substance substantially similar to that set forth on Schedule D-1A and D-1B.

"Boise Marks" means (i) the trade names Boise, Cascade, Boise Cascade, and combinations of such names with other words and phrases, including, without limitations, the trademarks and business descriptions Boise Building Solutions and Boise Paper Solutions; (ii) Parent's tree in a circle logo, and its "Boise" logo; and (iii) Parent's New York Stock Exchange trading symbol, BCC, and all rights related to the assets described in clauses (i) - (iii) foregoing.

"Boise Office Solutions" means Boise Cascade Office Products Corporation, a Delaware corporation, and OfficeMax, Inc., an Ohio corporation, and their respective direct and indirect Subsidiaries to the extent engaged in the office products distribution business conducted by such parties, and in each case shall include such Person as the name thereof may be changed after Closing after giving effect to the covenants of Parent in Section 10.6 hereof.

"BOS Paper Sales Agreement" means the agreement titled Paper Purchase Agreement Term Sheet dated April 28, 2004, between Parent and Boise Office Solutions.

"Business" means the forest products and paper manufacturing and distribution businesses that have been conducted by Sellers and their Subsidiaries prior to the Closing, including, without limitation: (1) their timberlands operations; (2) their paper manufacturing, conversion, distribution and sales operations; (3) their building solutions manufacturing and conversion operations; (4) their building solutions distribution and sales operations, (5) their packaging and newsprint manufacturing, conversion, distribution and sales operations, (6) Parent's aviation operations (other than as specifically included in Retained Assets), (7) all business activities currently conducted at the facility locations set forth on Schedule D-2, (8) all business activities contemplated as part of the business to be sold as depicted in the Historical Financial Statements and the management presentations delivered by Parent and its management to Holdings and its Affiliates, (9) their transportation operations; and (10) the Transferred Headquarters Operations. For greater certainty, the "Business" does not include: (a) any portion of the operations described in the immediately preceding sentence that was sold, transferred, discontinued or terminated prior to the date hereof; (b) Boise Office Solutions, (c) any line of business of Parent or its Retained Subsidiaries not described in the preceding sentence, including any such line of business that was sold, transferred, discontinued or terminated prior to the date hereof, (d) the architectural products business division of Parent and its Subsidiaries, or (e) any of the Excluded Staff Functions.

"Cap" means two hundred forty-eight million nine hundred thousand dollars (\$248,900,000).

"CERCLA" shall have the meaning given it in Section 17.9.

"Chosen Courts" shall have the meaning given it in Section 19.11.

3

"Closing" means the simultaneous conveyance of the Assets to Holdings or its designated Permitted Affiliate Purchasers by Sellers, payment by Holdings to Sellers of the Non-Timber Purchase Price and the cash portion of the Timberlands Purchase Price, issuance of the Timber Notes, Reinvestment by Parent or one of its Subsidiaries in Operating Holding Co. and by Parent Sub in Timber Co., and delivery by the parties of the other items called for in this Agreement at "Closing".

"Closing Balance Sheet" means the consolidated balance sheet for the Operating Unitsto be prepared by Holdings pursuant to Section 4.2.1.

"Closing Date" shall have the meaning given it in Section 14.1.

"Closing Net Working Capital" means the Net Working Capital of the Operating Units, determined in accordance with the Accounting Principles, as of immediately prior to Closing (but assuming consummation of the transactions contemplated hereby).

"Closing Statement" shall have the meaning given it in Section 4.2.1.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Commitment Letters" shall have the meaning given it in Section 8.8.

"Company Superior Proposal" shall have the meaning given it in Section 10.11.

"Company Takeover Proposal" shall have the meaning given it in Section 10.11.

"Confidential Information" shall have the meaning given it in Section 19.18.

"Confidentiality Agreement" shall mean the letter agreement dated January 21, 2004, between Madison Dearborn Partners, Inc. and Goldman, Sachs & Co. on behalf of Parent.

"Consent" shall mean any consent, approval, authorization, waiver, permit, grant, franchise, concession, agreement, license, exemption or order of, registration, certificate, declaration or filing with, or report or notice to any Person, including any Governmental Authority, including those identified pursuant to Sections 7.3, 7.14, and 7.15.

"Contracts" means all agreements, contracts, timber deeds, cutting contracts, purchase orders, refund and other arrangements, IP Licenses, incentive agreements, commitments, collective bargaining agreements, and other agreements to which any of the Sellers or any of their respective Subsidiaries are a party or to which any of the Assets are subject, and which are related to the Business, except to the extent included in the Retained Assets.

"Credit Enhancement Banks" means one or more banks selected by Timber Co. with the consent of Parent (such consent not to be unreasonably withheld) to provide the standby letters of credit required by Section 4.1 hereof as credit enhancement for the Timber Notes.

"Current Assets" means, with respect to the Operating Units, as of any date, the amounts (other than cash or cash equivalents) required to be recorded as current assets on a balance sheet for the

4

"Current Liabilities" means, with respect to the Operating Units, as of any date, the amounts required to be recorded as current liabilities on a balance sheet for the Operating Units prepared in accordance with the Accounting Principles and as set forth on Schedule D-3.

"Debt Commitment Letters" shall have the meaning given it in Section 8.8.

"Deductible" means twenty million seven hundred thousand dollars (\$20,700,000).

"Deeds" shall have the meaning given it in Section 9.2.

"Defined Contribution Plans" means the Boise Cascade Corporation Savings and Supplemental Retirement Plan, the Boise Cascade Corporation Retirement Savings Plan, and the Boise Cascade Corporation Qualified Employee Savings Trust.

"Disclosed" means, with respect to Parent and where referenced in any representation, warranty, covenant, or agreement of Parent herein, a specific disclosure made by Parent to Holdings contained in Parent's Disclosure Memorandum, and, with respect to Holdings and where referenced in any representation, warranty, covenant, or agreement of Holdings herein, a specific disclosure made by Holdings to Parent in Holdings' Disclosure Memorandum, in each case with specific identification of the representation, warranty, covenant or agreement for which such disclosure is made.

"Disclosure Memorandum" means, in respect of each of Parent or Holdings, the memorandum delivered by such party of even date herewith providing specific Disclosures referred to in this Agreement.

"Elections" means, with respect to the Transferred Subsidiaries, the elections to be made by Sellers and Holdings, or a Permitted Affiliate Purchaser, pursuant to Section 338(h)(10) of the Code, as described in Section 12.1 hereof.

"Employees" means all persons employed by any of Sellers or any of their respective Subsidiaries whose duties are wholly or primarily related to or involved in the conduct of the Business. Employees does not include persons assigned exclusively or primarily to the Excluded Staff Functions nor does it include the specific additional individuals listed in Schedule D-4.

"Encumbrance" shall mean Financial Encumbrances, easements, covenants, use restrictions, zoning restrictions and other third-party rights affecting any Asset.

"Environmental Assessment" has the meaning given it in Section 9.3.

"Environmental Deductible" means five million dollars (\$5,000,000).

"Environmental Law" shall mean any applicable federal, state, local, common or foreign law, statute, ordinance, rule, regulation, code, order, judgment, decree, injunction or other legal requirement relating to (i) the protection of the environment (including air, water vapor, surface

5

water, groundwater, drinking water supply, surface or subsurface land), or (ii) the exposure to, or the use, storage, recycling, treatment, generation, transportation, processing, handling, labeling, protection, release or disposal of, hazardous materials, substances or wastes, or workplace safety or health, in each case as the same may be in effect from time to time on or prior to the Closing Date; <u>provided</u> that for purposes of clauses (A), (B), (C), (F) and (G) of the definition of "Retained Environmental Liabilities," "Environmental Law" shall include all of the foregoing legal requirements as the same may be in effect from time to time prior to, on, or after the Closing Date.

"Equipment Leases" means all leases held by either of the Sellers or any of the Retained Subsidiaries as lessee of Fixtures and Equipment, except to the extent included in the Retained Assets.

"Equity Commitment Letter" shall have the meaning given it in Section 8.8.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any entity that, together with Parent or any of its subsidiaries, is treated as a single employer under Section 414 of the Code.

"Estimated Non-Timber Purchase Price" shall have the meaning given it in Section 4.2.

"Excluded Staff Functions" means the staff functions in Parent's Headquarters Facility which are listed in Schedule D-5.

"Facility" means any discrete location owned, leased or licensed by Sellers or any of their Subsidiaries related to or in connection with the conduct of the Business, including the locations set forth on Schedule D-2, but excluding any Retained Assets. When the term Facility is combined with a geographic location, it is intended to refer solely to the specific Facility at such geographic location. Headquarters Facility means Parent's headquarters building at 1111 West Jefferson Street, Boise, Idaho and the additional properties owned or leased by it in the immediate vicinity thereof.

"Facility Leases" means all leases of real property in respect of any Facility.

"Financial Encumbrances" means liens, charges, mortgages, security interests, encumbrances, or other interests or rights in any portion of the Assets which subject such Asset to payment of any obligation or which relate to claims and litigation included in the Assumed Liabilities; provided that "Financial Encumbrances" shall not include (a) the lien for property Taxes not yet due and payable, and (b) mechanics, materialman's, and warehouseman's liens and other similar statutory liens arising in the ordinary course of the conduct of the Business which are not yet perfected because payment of the amount secured is not due.

"Fixtures and Equipment" means all furniture, fixtures, furnishings, machinery, vehicles, equipment (including computer hardware, computer terminals, network servers, and research and development equipment) and other tangible personal property owned by any of Sellers or the Retained Subsidiaries and that are (a) located at any of the Facilities or the Timberlands or (b)

exclusively or primarily related to, or used exclusively or primarily in the conduct of, the Business, in each case except to the extent included in the Retained Assets.

"Former Facility" means a facility or property previously owned or operated by any of Sellers or any of their respective Subsidiaries or their respective predecessors in the conduct of the Business, other than the Facilities.

"Former Timberlands" means any real property owned or operated at any time prior to Closing by any of Sellers or any of their respective Subsidiaries or their respective predecessors for the purpose of growing and harvesting timber, which is not included in the Timberlands due to the prior sale or transfer of such lands.

"GAAP" means United States generally accepted accounting principles, consistently applied.

"Governmental Authority" means any nation or government, any state, province or other political subdivision thereof, and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any government authority, agency, department, board, commission or instrumentality of the United States, any state of the United States or any political subdivision thereof or of the nations of Canada, Brazil, United Kingdom, Chile, or the People's Republic of China.

"Governmental Authorizations" means all licenses, permits, franchises, certificates of occupancy, other certificates and other authorizations and approvals required to carry on the Business as currently conducted under the applicable laws, ordinances or regulations of any Governmental Authority.

"Harvest Plan" means Parent's actual and projected timber harvest plans posted to the on-line database during the month of June 2004 entitled "Harvest Plan Update" under the file names "6-x-3-14", with the letter "x" denoting the region number assigned by Parent.

"Hazardous Substances" means (i) petroleum, petroleum byproducts and any petroleum fractions; (ii) materials which contain any substance defined as a hazardous or toxic substance or words of similar meaning or effect under the following United States federal statutes and their state counterparts, as well as such statutes' implementing regulations: the Hazardous Materials Transportation Act, the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation and Liability Act, the Clean Water Act, the Safe Drinking Water Act, the Atomic Energy Act, the Toxic Substances Control Act, the Federal Insecticide, Fungicide, and Rodenticide Act, and the Clean Air Act; and (iii) any other material, substance or waste as to which liability or standards of conduct may be imposed pursuant to Environmental Laws.

"Headquarters Facility" has the meaning given it in the definition of "Facility."

"Historical Accounting Procedures" means the accounting principles and procedures used by Parent in preparing the Historical Financial Statements.

"Historical Financial Statements" has the meaning given to such term in Section 7.6.

7

"Holdings" shall mean Forest Products Holdings, L.L.C., a Delaware limited liability company.

"Holdings General Severance Policy" shall have the meaning given it in Section 11.8.

"Holdings Indemnified Parties" shall have the meaning given it in Section 17.2.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" means with respect to the Business, on any date, each of the following obligations of Seller and its Subsidiaries in connection with the Business, except to the extent included in Retained Liabilities: (i) any obligation for borrowed money or issued in substitution for or exchange of indebtedness for borrowed money, (ii) any obligation evidenced by any note, bond, debenture or other debt security, (iii) any obligation for the deferred purchase price of property or services with respect to which any such Person is liable, contingently or otherwise, as obligor or otherwise (other than trade payables and other current liabilities incurred in the ordinary course of business), including any amounts owed by any such Person under any noncompetition or consulting Contract that was entered into in connection with an acquisition or sale of any portion of a business, (iv) any commitment by which any such Person assures a creditor against loss (including, without limitation, contingent reimbursement obligations with respect to letters of credit and performance bonds) exclusive of (x) letters of credit and performance bonds to the extent they provide standby assurance for obligations which are included in the adjustments to the Non-Timber Purchase Price provided for in Section 3.2.2, (y) letters of credit and performance bonds to the extent they provide standby credit support for Contracts, Leases and Governmental Authorizations that are assumed by Holdings, and (z) performance bonds to the extent posted as security for lien claims included in Transferred Litigation, (v) any obligation guaranteed in any manner by any such Person (other than warranties made to customers in the ordinary course of business, but including guarantees in the form of an agreement to repurchase or reimburse), (vi) any obligations under capitalized or synthetic leases (as determined in accordance with GAAP) with respect to which any such Person is liable, contingently or otherwise, as obligor, guarantor or otherwise, or with respect t

"Indemnification Claim Notice" shall have the meaning given it in Section 17.6.

"Indemnified Environmental Liabilities" means all liabilities arising under Environmental Laws relating to or arising from (A) any breach of the representations and warranties set forth in Section 7.11; (B) any Straddle Environmental Regulatory Issue (solely to the extent arising from pre-Closing operations or activities); and (C) any Straddle Hazardous Substance Release (solely to the extent arising from pre-Closing operations or activities).

"Indemnified Party" shall have the meaning given it in Section 17.6.

"Indemnifying Party" shall have the meaning given it in Section 17.6.

"Independent Accountants" means the accounting firm identified as such in Section 4.2.3.

8

"Intellectual Property" means (except to the extent included in the Retained Assets) the following intellectual property (and the rights associated therewith) exclusively or primarily related to, or used by Sellers and its Subsidiaries exclusively or primarily in the conduct of, the Business: trademarks, service marks, stock exchange trading symbols, brand names, certification marks, trade dress, assumed names, Internet domain names, trade names and other indications of origin, the goodwill associated with the foregoing and registrations in any jurisdiction of, and applications in any jurisdiction to register, the foregoing, including any extension, modification or renewal of any such registration or application; patents, applications for patents (including divisionals, continuations, continuations-in-part and renewal applications), and any renewals, extensions or reissues thereof, in any jurisdiction; patent disclosures and inventions (whether or not patentable and whether or not reduced to practice); non-public information, trade secrets and confidential information and rights in any jurisdiction to limit the use or disclosure thereof by any Person; copyrightable works and registrations or applications for registration of copyrights in any jurisdiction, and any renewals or extensions thereof; software; any similar intellectual property or proprietary rights; and any claims or causes of action arising out of or related to any infringement or misappropriation of any of the foregoing occurring before or after the Closing.

"Interest Rate" means a rate of interest equal to LIBOR plus 300 basis points.

"Inventory" means all goods owned by any of the Sellers or any of the Retained Subsidiaries and held by them for resale in connection with the conduct of the Business, together with all raw materials, work in process, finished products, office supplies, storeroom inventory, spare parts and equipment, wrapping, supply and packaging items owned by any of them in connection with the conduct of the Business.

"IP Licenses" means all licenses and other agreements held by any of Sellers or any of the Retained Subsidiaries allowing them to use the intellectual property of others, including without limitation, software licenses.

"Knowledge," "Known", or any similar phrase means the actual knowledge, after due inquiry, of one or more of the following management employees of Parent and its Subsidiaries as of the date hereof and the Closing Date: Chairman and Chief Executive Officer; Senior Vice President and Chief Financial Officer; Senior Vice President and General Counsel; Vice President and Controller; Senior Vice President and General Manager, Boise Paper Solutions; Senior Vice President and General Manager, Boise Building Solutions, Distribution.

"KPMG" shall have the meaning given it in Section 10.13.

"LC Commitment Letters" shall have the meaning given it in Section 4.1.2.

"Leased Timberlands" shall have the meaning given it in the definition of "Timberlands."

"Leases" means the Equipment Leases, the Facility Leases, and the Timberlands Leases.

9

"LIBOR" shall mean the London interbank offered rate per annum for 30-day periods for deposits in dollars, determined in the manner provided for such rate under the definitive loan documentation entered into by Holdings in connection with the debt financing contemplated by the Debt Commitment Letters. For purposes of this Agreement, the LIBOR will be determined on the date when interest begins to accrue pursuant to the terms of this Agreement and will be redetermined on the last day of each 30-day period thereafter during which interest is being accrued.

"Litigation and Claims Reserve" means, as of Closing, an aggregate reserve for liabilities relating to litigation and other claims constituting Transferred Litigation, which will be equal to the sum of the individual reserves set forth opposite each item of Transferred Litigation listed on Schedule D-6, as such individual reserves may be adjusted solely pursuant to this definition and the second paragraph of Section 7.21. From the date hereof until the Closing, each of the individual reserves set forth on Schedule D-6 will be increased (in the case of any increase in the expected liability relating to the item of Transferred Litigation corresponding to such individual reserve) or decreased (in the event the item of Transferred Litigation corresponding to such individual reserve is settled or resolved in the ordinary course of business and ceases to be an Assumed Liability) in accordance with Parent's historical practices and methods for establishing and adjusting book reserves relating to litigation and other claims liabilities.

"Losses" shall have the meaning given it in Section 17.2.

"Matching Offer" shall have the meaning given it in Section 10.11.

"Material Adverse Change" and "Material Adverse Effect" mean a change or an effect that is or would reasonably be expected to be materially adverse to the value of the Assets or the Business taken as a whole or materially adverse to the business, financial condition or results of operations or business prospects of the Business taken as a whole; provided that none of the following shall be considered a Material Adverse Change or a Material Adverse Effect: (1) any general market, economic or political conditions (it being understood that "political conditions" would not include the effects of any terrorist attack) affecting the economy or market generally, or normal cyclical fluctuations affecting the forest products or paper manufacturing and distribution industry; (2) any change or effect resulting from compliance with the terms and conditions of this Agreement; or (3) any change in accounting requirements or principles or applicable laws, rules or regulations of general application that becomes effective after the date of this agreement. The scope of this definition of "Material Adverse Change" and "Material Adverse Effect" shall in no way be construed to be applicable to or to limit in any respect the determination of "Material Adverse Effect" or "Material Adverse Change" as used in the Commitment Letters, or the agreements and indentures contemplated thereby, or by the lenders thereunder with respect to the conditions precedent to such lenders' obligation to consummate the financing contemplated by the Debt Commitment Letters.

"MDP" means Madison Dearborn Capital Partners IV, L.P., and its affiliated entities.

"Minidoka" means Minidoka Paper Company, a Delaware corporation.

10

"Miscellaneous Assets" means assets, rights and properties of any of the Sellers or the Retained Subsidiaries not included in any other category of Assets which are exclusively or primarily related to, or used exclusively or primarily in the conduct of, the Business, except to the extent included in Retained Assets.

"Mutual Administrative Services Agreement" means the Agreement between Parent and Holdings to be entered into as of the Closing pursuant to Section 18.1.

"Net Working Capital" means, with respect to the Operating Units, as of any date, the excess (whether positive or negative) of Current Assets over Current Liabilities.

- "Noncompete Period" shall have the meaning given it in Section 10.12.
- "Non-Timber Assets" shall mean all of the Assets, save and except the Timberlands.
- "Non-Timber Purchase Price" means the amount payable by Holdings and/or its Permitted Affiliate Purchasers to Sellers for the Non-Timberlands Business, as determined pursuant to Section 3.1.
- "Non-Timberlands Business" means that portion of the Business which does not include the Timberlands Business.
- "Notice Period" shall have the meaning given it in Section 17.6.
- "Objection Statement" shall have the meaning given it in Section 4.2.2.
- "Operating Holding Co." means BCC Holdings, L.L.C., a Delaware limited liability company. Operating Holding Co. is a Permitted Affiliate Purchaser.
- "Operating Units" means the following operating divisions of Parent and its Subsidiaries, as generally understood in the historical operation of the Business: Boise Building Solutions, Manufacturing; Boise Building Solutions, Distribution; and Boise Paper Solutions.
- "Outside Date" shall mean November 12, 2004.
- "Owned Timberlands" shall have the meaning given it in the definition of "Timberlands."
- "Parent" means Boise Cascade Corporation, a Delaware corporation, and shall include such Person to which the name thereof may be changed after Closing after giving effect to the covenants of Parent in Section 10.6 hereof.
- "Parent Confidential Information" shall have the meaning given it in Section 19.18.
- "Parent Indemnified Parties" shall have the meaning given it in Section 17.4.
- "Parent Intellectual Property" means (except to the extent included in the Retained Assets) the Intellectual Property owned by any of the Sellers or any of the Retained Subsidiaries and related to, or used by any of them in their conduct of or operation of, the Business or the Assets.

11

- "Parent Plans" means each "employee benefit plan" as defined in Section 3(3) of ERISA, and each pension, retirement, profit sharing, stock bonus, stock option, stock purchase, severance, retention, change of control, deferred compensation, medical, post-employment welfare, dental, vision, vacation, life insurance, sick pay or disability plan, program, policy or arrangement or any material bonus, incentive, death benefit or other employee benefit plan, program, policy or arrangement maintained or contributed to by any of the Sellers, any of the Transferred Subsidiaries or any of the Retained Subsidiaries for any of the Employees or with respect to which any Transferred Subsidiary has any liability.
- "Parent Preliminary Tax Payment" shall have the meaning given it in Section 12.2.3.
- "Parent Sub" means a Delaware corporation to be organized by Parent prior to Closing which shall be a wholly-owned Subsidiary of Parent.
- "Parent Timberlands Purchase Price" shall have the meaning given it in Section 3.4.
- "Pension Plans" means: (i) the Boise Cascade Corporation Pension Plan for Salaried Employees; (ii) Boise Cascade Corporation WCIW Pension Plan "A"; (iii) Boise Cascade Corporation Pension Plan "B"; (iv) Boise Cascade Corporation Pension Plan "C"; (v) Boise Cascade Corporation Pension Plan "P"; and (vi) the Boise Cascade Corporation Pension Plan for Hourly Employees at International Falls, Minnesota.
- "Permits and Licenses" means all permits, licenses, and other governmental authorizations held by any of Sellers or any of the Retained Subsidiaries to conduct any aspect of the Business.
- "Permitted Affiliate Purchaser" means Operating Holding Co., Timber Co., Timber LLC and each of the other corporations, limited liability companies, or other business organizations listed on Schedule D-7.
- "Permitted Encumbrances" shall have the meaning given it in Section 7.16.2.
- "Perpetual Timber Rights" shall have the meaning given it in the definition of "Timberlands."
- "Person" means an individual, a corporation, a partnership, a limited liability company, an association, a trust or other entity or organization.
- "Pre-Closing Environmental Regulatory Issue" means any failure to comply with Environmental Laws by the Business or at any Facility which failure to comply occurred prior to the Closing Date and did not continue after the Closing Date as a result of the operation of the Business by Purchasers in violation of Permits and Licenses or other regulatory requirements under Environmental Laws immediately after the Closing.
- "Pre-Closing Hazardous Substance Release" means any Release of Hazardous Substances or any condition of contamination on, at, or migrating from any Facility, which Release or condition of contamination occurred or was caused prior to the Closing Date, and which Release did not continue as a result of, or which condition of contamination was not exacerbated by (excluding, for the avoidance of doubt, any exacerbation arising from post-Closing passive migration of such Hazardous Substances), the operation of the Business by Purchasers in violation of Permits and

12

Licenses or other regulatory requirements under Environmental Laws immediately after the Closing.

- "Pre-Closing Returns" shall have the meaning given it in Section 12.2.1.
- "Pre-Closing Workers' Compensation Policy" has the meaning given it in Section 11.9.
- "Proceeding" shall have the meaning given it in Section 17.6.
- "Purchaser Payment Performance Obligations" shall have the meaning given it in Section 10.15.2.
- "Purchasers" means Holdings, Timber Co., and the other Permitted Affiliate Purchasers designated by Holdings to acquire portions of the Assets.
- "Realty" means all real property owned by Sellers or their Retained Subsidiaries which is exclusively or primarily related to, or used exclusively or primarily in the conduct of, the Business, including real property of which the address is listed on Schedule D-8; provided that "Realty" does not include the Timberlands nor any Retained Asset.
- "REC" shall have the meaning given it in Section 9.3.
- "Regulation S-X" shall have the meaning given it in Section 10.13.
- "Regulation S-X Financial Statements" shall have the meaning given it in Section 10.13.
- "Reinvestment" means Parent's purchase of the securities of Operating Holding Co. and Parent Sub's purchase of the securities of Timber Holdings, in each case as provided for in Article 6.
- "Reinvestment Amount" means one hundred seventy-five million dollars (\$175,000,000), which is the aggregate cash consideration to be paid by Parent and Parent Sub for the Reinvestment.
- "Release" shall have the meaning given to it in the Comprehensive Environmental Response, Compensation and Liability Act, 42 USC 9601 et seq.
- "Required Consents" shall have the meaning given it in Section 15.1.3.
- "Retained Assets" means the assets, rights, and properties of any of Sellers listed on Schedule D-9.
- "Retained Environmental Liabilities" means all liabilities arising under Environmental Laws relating to or arising from (A) any operations or businesses conducted by Sellers or the Transferred Subsidiaries other than the Business; (B) any property, facility or location associated with the Business other than the Facilities or the Timberlands; (C) any Former Facility or Former Timberlands; (D) any Pre-Closing Hazardous Substance Release; (E) any Pre-Closing Environmental Regulatory Issue; (F) the discharge, disposal or arrangement for the disposal of

13

Hazardous Substances prior to Closing onto or into any location other than the Facilities; (G) events or conditions set forth on Schedule D-10 hereto; or (H) any matter so designated pursuant to Section 9.3.2.2 hereof. Any liability falling within the foregoing definition shall be deemed to be a Retained Environmental Liability for all purposes under this Agreement, regardless of whether it is Disclosed in the Parent's Disclosure Memorandum or would also fall within the definition of "Indemnified Environmental Liabilities."

- "Retained HBU Lands" means the 6,000 acres of timberlands in Idaho previously identified by Parent for sale in its Idaho higher and better use land sales program and included on Schedule D-9 as a Retained Asset.
- "Retained Liabilities" means the liabilities and obligations of Sellers and its Subsidiaries which are to be retained by Sellers pursuant to Section 5.3. Any liability falling within the foregoing definition shall be deemed to be a Retained Liability for all purposes under this Agreement, regardless of whether it is Disclosed in Parent's Disclosure Memorandum.
- "Retained Litigation" means all claims and litigation, pending or threatened, against any of Sellers or any of their respective Subsidiaries relating to the Business other than the Transferred Litigation, including any claim or litigation which (i) is based on the plaintiff's exposure to asbestos (including silica or mixed dust if part of or included with an asbestos claim); (ii) arises out of the conduct of the Business at any location other than the Facilities; (iii) arises out of any activities or relationships of any of Sellers or any of their respective Subsidiaries which did not arise from their conduct of the Business; or (iv) is listed in Schedule D-11.
- "Retained Records" means (whether in paper or electronic form in all cases) all files and records of any of Sellers or any Retained Subsidiary relating to or arising out of any aspect of their business and affairs not included in the Business or the Assets, including, without limitation, financial records, Tax Returns, dispute settlements, transactional records, realty abstracts and other realty records, minute books and other corporate records of any of Sellers or any Retained Subsidiary or former subsidiary of Sellers or any Retained Subsidiary.
- "Retained Subsidiaries" means the corporations and other business entities identified as such on Schedule D-12.
- "Sellers" means, collectively, Parent, Southern and Minidoka.
- "Shareholder Agreements" means the agreements between Parent and Holdings concerning their respective investments in Operating Holding Co., and between Holdings and Parent Sub concerning their respective investments in Timber Holdings, in each case as provided for in Section 6.3.
- "Southern" means Boise Southern Company, a Louisiana General Partnership composed of Parent and Minidoka, and shall include such Person to which the name thereof may be changed after Closing after giving effect to the covenants of Parent in Section 10.6 hereof.
- "Southern Timberlands Purchase Price" shall have the meaning given it in Section 3.4.

"Straddle Environmental Regulatory Issue" means any failure to comply with Environmental Laws by the Business or at any Facility prior to the Closing where such failure to comply continues as a result of the operation of the Business by Purchasers in violation of Permits and Licenses or other regulatory requirements under Environmental Laws immediately after the Closing.

"Straddle Hazardous Substance Release" means any Release of Hazardous Substances or condition of contamination on, at, or migrating from any Facility prior to Closing, where such Release continues as a result of, or such conditions of contamination are exacerbated by (excluding, for the avoidance of doubt, any exacerbation arising from post-Closing passive migration of such Hazardous Substances), the operation of the Business by Purchasers in violation of Permits and Licenses or other regulatory requirements under Environmental Laws immediately after the Closing.

"Straddle Period" means any taxable period that begins before and ends after the Closing Date.

"Straddle Period Return" means any Tax Return for a Straddle Period.

"Stub Period Financial Statements" shall have the meaning given it in Section 10.13.

"Subsequent Financial Statements" shall have the meaning given it in Section 10.13.

"Subsidiary" means, with respect to any Person, any other Person of which (i) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a partnership, limited liability company, association or other business entity, a majority of the partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a partnership, limited liability company, association or other business entity if such Person or Persons shall be allocated a majority of partnership, limited liability company, association or other business or shall be or control the managing director or general partner of such partnership, association or other business entity.

"Surveys" shall have the meaning given it in Section 9.1.

"Tax" and "Taxes" means all federal, state, local or foreign taxes, including but not limited to income, gross receipts, windfall profits, goods and services, value added, severance, property, production, sales, use, license, excise, franchise, employment, estimated, environmental, minimum, withholding or other taxes of any kind, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties.

"Tax Authority" means, with respect to any Tax, the governmental entity or political subdivision thereof that imposes such Tax, and the agency (if any) charged with the collection of such Taxes for such entity or subdivision.

15

"Tax Benefit" means the amount, if any, by which a Person's net Tax liability is actually reduced (including any related interest actually received from a Tax Authority in connection therewith, net of any Tax imposed thereon), after taking into account the tax effects of the Loss at issue and any corresponding indemnification payments.

"Tax Claim" shall have the meaning given it in Section 12.5.1.

"Tax Indemnification Obligation" shall have the meaning given it in Section 12.2.3.

"Tax Indemnified Party" shall have the meaning given it in Section 12.5.1.

"Tax Indemnifying Party" shall have the meaning given it in Section 12.5.1.

"Tax Period" means, with respect to any Tax, the period for which the Tax is reported as provided under applicable Tax laws.

"Tax Return" means, with respect to any Tax, any information return with respect to such Tax, any report, statement, declaration or document required to be filed under the applicable Tax law in respect of such Tax, any claim for refund of Taxes paid, and any amendment or supplement to any of the foregoing.

"Timber Co." means Boise Land & Timber Corp., a Delaware corporation and wholly owned subsidiary of Timber Holdings. Timber Co. is a Permitted Affiliate Purchaser with respect to the Timberlands.

"Timber Holdings" means Boise Land & Timber Holdings Corp., a Delaware corporation and subsidiary of Holdings.

"Timber LLC" means Boise Land & Timber, L.L.C., a Delaware limited liability company and wholly owned subsidiary of Timber Co., which as of the date hereof is treated as a disregarded entity and as a division of Timber Co. for federal income tax purposes. Timber LLC is a Permitted Affiliate Purchaser with respect to the Timberlands. At Holdings' option, and consistent with the LC Commitment Letters, Timber LLC may consist of two such Delaware limited liability companies, in order to accommodate multiple Credit Enhancement Banks, and collectively such limited liability companies will perform the obligations of Timber LLC hereunder.

"Timber Notes" shall have the meaning given it in Section 4.1.

"Timber Supply Agreement" means a contract or timber deed obligating Sellers or any of the Transferred Subsidiaries to purchase standing timber, to pay for standing timber already conveyed to them under a timber deed, or to purchase delivered logs.

"Timberlands" means all real property owned or leased by either of Parent or Southern for the purpose of growing and harvesting timber, including: (i) real property owned by Parent or Southern described in Schedule D-13A (the "Owned Timberlands"); (ii) leases of land for the purpose of growing and harvesting timber which are described in Schedule D-13B (the "Timberlands Leases" and the "Leased Timberlands"); and (iii) the perpetual timber rights

- "Timberlands Business" means that portion of the Business of growing and harvesting timber conducted by Sellers and its Subsidiaries with the Timberlands.
- "Timberlands Leases" shall have the meaning given it in the definition of Timberlands.
- "Timberlands Purchase Price" means the amount to be paid by Timber Co. to Parent and Southern for the Timberlands determined in accordance with Section 3.1.
- "Title Commitments" shall have the meaning given it in Section 9.1.
- "Title Company" shall have the meaning given it in Section 9.1.
- "Title Policies" shall have the meaning given it in Section 9.1.
- "Trade Intercompany Accounts" means bona fide trade intercompany accounts payable and accounts receivable (as determined in accordance with GAAP) that are incurred for the provision of goods and services in the ordinary course of business.
- "Transferred Cash" means (i) the lock boxes, deposit accounts, payroll accounts, and payables accounts of Sellers and the Transferred Subsidiaries exclusively or primarily related to, or used exclusively or primarily in, the conduct of the Business, and the cash balances thereof, and (ii) an amount of cash and cash equivalents that will be specified by Holdings to Parent in writing not less than 10 days prior to Closing.
- "Transferred Headquarters Operations" shall mean all of the business operations, support functions, and other activities conducted by Parent and its Subsidiaries at the Headquarters Facility (including all assets of Sellers and its Subsidiaries located at the Headquarters Facility and any other assets of Sellers and its Subsidiaries exclusively or primarily related to, or used exclusively or primarily in, the Headquarters Facility or the conduct of such activities), except for the Excluded Staff Functions.
- "Transferred Insurance Policies" shall have the meaning given it in Section 10.10.
- "Transferred Litigation" means the claims and litigation, pending or threatened, against any of Sellers which arose out of their conduct of the Business and which are set forth on Schedule D-15 to this Agreement.
- "Transferred Records" means all files and records of any of Sellers and their Subsidiaries generated by them in the course of their conduct of the Business, whether in paper or electronic form, exclusive of "Retained Records."
- "Transferred Subsidiaries" shall mean the corporations and other business entities identified as such on Schedule D-12 to this Agreement.
- "Transfer Taxes" shall have the meaning given it in Section 12.3.

17

"Unfunded Pension Plans" means (i) the Boise Cascade Corporation Supplemental Early Retirement Plan for Executive Officers; and (ii) the Boise Cascade Corporation Supplemental Pension Plan.

"Workers Compensation Reserve" means, as of Closing, an aggregate reserve for liabilities relating to workers compensation claims, which will be equal to the sum of (x) the IBNR reserve set forth on Schedule D-16, and (y) the claims reserve set forth on Schedule D-16, in each case as such reserves may be adjusted solely pursuant to this definition. From the date hereof until the Closing, the IBNR reserve described in clause (x) above will be adjusted on a semi-annual basis as required by GAAP in accordance with Parent's historical practices and methods for establishing and adjusting IBNR reserves relating to workers compensation claims. From the date hereof until the Closing, the claims reserve described in clause (y) above will be increased (in the case of any increase in the expected liability relating to any of the workers compensation claims included in such claims reserve are settled or resolved in the ordinary course of business and cease to be an Assumed Liability) in accordance with Parent's historical practices and methods for establishing and adjusting book reserves relating to workers compensation claim arising from the most recent month-end prior to the date hereof until the Closing Date, an individual reserve will be established for such claim, which reserve will be determined in accordance with Parent's historical practices and methods for establishing and adjusting book reserves relating to workers compensation claims and will be reasonably acceptable to Holdings, and such individual reserve will be added to the claims reserve described in clause (y) above.

"2004 Capital Spending Authorizations" means the amounts for capital expenditures set forth on Schedule D-17.

"2004 Retention Bonus Plan" means the retention bonus plan and the retention agreements for officers and nonofficers in the form set forth in Schedule D-18 of Parent's Disclosure Memorandum. Schedule D-18 also includes a chart setting forth the aggregate amounts of bonuses to be awarded pursuant to this plan in 2004 for nonofficers and for executive officers (in each case, broken out and subtotaled by salary range).

"2004 Severance Agreements" means the severance agreements for executive officers of Parent in the form set forth in Schedule D-19 of Parent's Disclosure Memorandum. Schedule D-19 sets forth the identity of each executive officer to enter into such a severance agreement and the term and amount of severance pay or benefits applicable to each such executive officer.

# 1.2 <u>Interpretive Provisions</u>.

- 1.2.1 The words "hereof", "herein", and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "including" means "including without limitation."
  - 1.2.2 The terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa.

18

1.2.3 The terms "dollars" and "\$" shall mean United States dollars.

2.1 <u>Purchase and Sale of Assets.</u> At Closing, Sellers shall, and shall cause each of the Retained Subsidiaries to, sell, convey, assign and deliver, free and clear of all Encumbrances (other than Permitted Encumbrances), the Non-Timber Assets to Holdings or one or more of its designated Permitted Affiliate Purchasers, and Holdings or one or more of its designated Permitted Affiliate Purchasers shall purchase and accept the Non-Timber Assets from Sellers and such Retained Subsidiaries. In addition, at Closing, Parent and Southern shall sell, convey, assign and deliver, free and clear of all Encumbrances (other than Permitted Encumbrances), the Timberlands to Timber Co., and Timber Co. shall purchase and accept the Timberlands from Parent and Southern.

#### Article 3 Timberland Purchase Price and Non-Timber Purchase Price

- 3.1 <u>Timberlands and Non-Timber Purchase Prices</u>. The Non-Timber Purchase Price to be paid by Holdings and/or the Permitted Affiliate Purchasers to Sellers shall be two billion twenty-five million dollars (\$2,025,000,000), as adjusted pursuant to terms of this Agreement. The Timberlands Purchase Price to be paid by Timber Co. to Parent and Southern for the Timberlands shall be one billion six hundred fifty million dollars (\$1,650,000,000), as adjusted pursuant to the provisions of this Agreement.
  - 3.2 <u>Non-Timber Purchase Price Adjustments</u>. The Non-Timber Purchase Price shall be adjusted as follows:
- 3.2.1 It shall be reduced by the amount (which, in the case of standby letters of credit and performance bonds, will be the amount that would be required to be set forth on a balance sheet prepared in accordance with the Accounting Principles) of Indebtedness assumed by Purchasers at Closing.
  - 3.2.2 It shall be reduced by the following amounts:
    - 3.2.2.1 the Workers Compensation Reserve.
    - 3.2.2.2 the Litigation and Claims Reserve.
- 3.2.2.3 an amount equal to nineteen million seven hundred thousand dollars (\$19,700,000), which represents the amount to compensate Holdings for assuming certain obligations to provide post-employment welfare benefits to Transferred Employees who retire after the Closing Date, as described in Article 11.
- 3.2.3 It shall be increased by the amount, if any, by which the Closing Net Working Capital of the Operating Units is greater than five hundred forty-eight million nine hundred thousand dollars (\$548,900,000), or shall be decreased by the amount, if any, by which the Closing Net Working Capital of the Operating Units is less than four hundred ninety-eight million nine hundred thousand dollars (\$498,900,000).

19

- 3.2.4 It shall be increased by the amount of cash and cash equivalents included in clause (ii) of the definition of Transferred Cash and transferred to Holdings at Closing.
  - 3.3 [Reserved.]
  - 3.4 <u>Allocation of Purchase Price</u>.
- 3.4.1 Allocation of Non-Timber Purchase Price. Holdings shall determine and prepare, and Sellers shall agree to (which agreement shall not be unreasonably withheld), the allocation of the Non-Timber Purchase Price and the Assumed Liabilities that are considered assumed liabilities for income Tax purposes (other than liabilities directly related to the Timberlands which are assumed by Timber Co.) among the Non-Timber Assets to be purchased hereunder which allocation shall be finalized as soon as practicable after the Closing, but not later than six months after Closing, but shall be adjusted to take account of any post-closing Non-Timber Purchase Price adjustments (together with the allocation described in Section 3.4.2, the "Allocation"). The Allocation shall be made in accordance with Section 1060 of the Code and applicable Treasury regulations. Except to the extent such action or inaction would cause any Person to be in violation of the final determination of any Taxing Authority, each of Holdings and Sellers shall (i) be bound by the Allocation for purposes of determining any Taxes, (ii) prepare and file, and cause their affiliates to prepare and file, their Tax Returns on a basis consistent with the Allocation, and (iii) take no position, and cause their affiliates to take no position, inconsistent with the Allocation on any applicable Tax Return or in any proceeding before any Taxing Authority or otherwise. In the event that the Allocation is disputed by any Taxing Authority, the party receiving notice of the dispute shall promptly notify the other party hereto concerning the dispute and shall keep the other party informed regarding the dispute and its resolution.
- Allocation of Timberlands Purchase Price. The parties agree that the Timberlands Purchase Price (together with any Assumed Liabilities directly related to the Timberlands which are considered liabilities for income tax purposes and are assumed by Timber Co.) shall be allocated in its entirety to the Timberlands. Assets other than Timberlands utilized in the Timberlands Business or Assumed Liabilities directly related to the Timberlands Business will be allocated for Tax purposes as a part of the allocation provided for in Section 3.4.1, it being the intention of the parties that the Timberlands Purchase Price will be paid solely as consideration for the transfer of the Timberlands. The Timberlands Purchase Price shall be allocated between Parent (the "Parent Timberlands Purchase Price") and Southern (the "Southern Timberlands Purchase Price") in proportion to Parent's reasonable estimate of the relative fair value of their respective shares of the Timberlands. Such estimate shall be prepared in consultation with Timber Co.'s timberlands consultant, Forest Systems, LLC. Such allocation shall be provided to Timber Co within 15 days after the date hereof. If Holdings or Timber Co. disagrees with such allocation, the parties shall meet and negotiate in good faith in respect of such issue, and if they are unable to reach agreement, they shall jointly retain Peters Forest Resources as an independent forestry consultant and instruct such consultant to determine the allocation of the Timberlands Purchase Price between Parent and Southern based on the relative fair value of the Timberlands owned by Parent and Southern. Such consultant's conclusion shall be final and binding on the parties and shall be respected in further allocation provided for in the balance of this Section. Holdings shall determine and prepare, and Sellers shall agree to (which

20

agreement shall not be unreasonably withheld), the allocation of the Parent Timberlands Purchase Price among the Timberlands to be purchased hereunder from Parent, and of the Southern Timberlands Purchase Price among the Timberlands to be purchased hereunder from Southern, which allocation shall be finalized as soon as practicable after the Closing, but not later than six months after Closing, but shall be adjusted to take account of any post-closing Timberlands Purchase Price adjustments (together with the allocation described in Section 3.4.1, the "Allocation"). The Allocation shall be made in accordance with Section 1060 of the Code and applicable Treasury regulations. Except to the extent such action or inaction would cause any Person to be in violation of the final determination of any Taxing Authority, each of Holdings, Timber Co., Parent and Southern shall (i) be bound by the Allocation for purposes of determining any Taxes, (ii) prepare and file, and cause their affiliates to prepare and file, their Tax Returns on a basis consistent with the Allocation, and (iii) take no position, and cause their affiliates to take no

position, inconsistent with the Allocation on any applicable Tax Return or in any proceeding before any Taxing Authority or otherwise. In the event that the Allocation is disputed by any Taxing Authority, the party receiving notice of the dispute shall promptly notify the other party hereto concerning the dispute and shall keep the other party informed regarding the dispute and its resolution.

3.5 Additional Consideration. As additional mutual consideration for the Non-Timber Assets Parent and Holdings shall, at Closing, enter into the Additional Consideration Agreement in form and substance substantially similar to the form attached as Schedule 3.5.

#### Article 4 Terms of Payment

The Timberlands Purchase Price and the Non-Timber Purchase Price shall be paid as follows:

# 4.1 <u>Timberlands Purchase Price</u>.

- 4.1.1 An aggregate of \$5,000,000 of the Timberlands Purchase Price shall be payable to Parent and Southern in cash at Closing. Such cash portion of the Timberlands Purchase Price shall be allocated between Parent and Southern in proportion to the relative amounts of the Parent Timberlands Purchase Price and the Southern Timberlands Purchase Price, respectively.
- 4.1.2 The balance of the Timberlands Purchase Price shall be payable in one or more promissory notes (the "Timber Notes") issued by Timber LLC to each of Parent and Southern in an amount for Parent equal to the Parent Timberlands Purchase Price less the portion thereof paid to Parent in cash pursuant to Section 4.1.1, and an amount for Southern equal to the Southern Timberlands Purchase Price less the portion thereof paid to Southern in cash pursuant to Section 4.1.1, and which shall be in form and substance mutually satisfactory to Parent and Holdings and substantially consistent with the LC Commitment Letters (as defined below). Each Timber Note shall be secured by a standby letter of credit issued by a Credit Enhancement Bank in an amount not less than the aggregate principal amount of such Timber Note and for a term which exceeds the maturity of such Timber Note by at least 95 days. The standby letter of credit and related documentation shall be fully assignable in connection with any transfer, sale, assignment or pledge of the Timber Notes and shall otherwise be substantially consistent with the structure and terms described in the commitment letters attached hereto as Schedule 4.1(b)

21

(the "LC Commitment Letters") and reasonably acceptable in form and substance to Parent. The interest rate on the Timber Notes shall be a fixed stated rate of interest set forth in the Timber Notes, which shall, as described in the Section entitled "Timber Installment Notes" in Exhibit A to the LC Commitment Letters, be equal to (x) the interest rate for the "Collateral Notes" described in the Section entitled "Collateral Notes" in Exhibit A to the LC Commitment Letters, less (y) the sum of the cost of the letter of credit and an additional percentage rate intended to cover the cost of administering the Timber Notes during their term; provided that, at Holdings' option, the Timber Notes may provide for the payment of such interest rate without reduction for the costs of the letter of credit, and in such case Parent shall agree in writing with the Credit Enhancement Bank that it will upon receipt of each such interest payment pay the Credit Enhancement Bank directly the costs of the letter of credit.

- 4.1.3 Any reduction in the Timberlands Purchase Price pursuant to any provision of this Agreement (including Section 9.3.2.1 or 10.2.1.2 or Article 13) shall operate to reduce the cash portion of the Timberlands Purchase Price prior to any reduction in the aggregate amount of Timber Notes. Any such reduction shall be allocated between Parent and Southern in proportion to the relative amounts of the Parent Timberlands Purchase Price and the Southern Timberlands Purchase Price prior to such reduction.
- 4.2 Non-Timber Purchase Price. The Non-Timber Purchase Price shall be paid in cash to Parent, as agent for Sellers, at Closing by wire transfer of immediately available funds to a bank account designated by Parent. Such payment shall be made on the basis of Parent's good faith estimate of the Non-Timber Purchase Price (the "Estimated Non-Timber Purchase Price"). Such estimate shall be made by Parent, and communicated to Holdings not less than 10 business days prior to Closing. Parent shall estimate the closing payment on the basis of the latest financial data reasonably available (which shall not be more than 60 days old). Such calculation shall include an estimate of all adjustments to the Non-Timber Purchase Price required by this Agreement. During and after its preparation of the Estimated Non-Timber Purchase Price, Parent shall give Holdings and its representatives reasonable access to the information required to calculate, and otherwise being used to calculate, and the personnel (including outside accountants) involved in calculating, the Estimated Non-Timber Purchase Price. If Holdings disagrees with Parent's calculation of the Estimated Non-Timber Purchase Price, but if they cannot resolve their differences prior to Closing despite such reasonable best efforts, then the Estimated Non-Timber Purchase Price actually paid at the Closing shall be Parent's determination thereof.
- 4.2.1 Within 60 days following the Closing, Holdings shall, at its expense and with reasonable cooperation from Sellers' employees, prepare, or cause to be prepared, and deliver to Parent a consolidatedbalance sheet for the Operating Units as of immediately prior to (but assuming consummation of) the Closing (the "Closing Balance Sheet"). The Closing Balance Sheet shall be prepared from the books and records of the Business in accordance with the Accounting Principles and as set forth on Schedule D-3. The Closing Balance Sheet shall be accompanied by a statement derived from the Closing Balance Sheet (the "Closing Statement") setting forth Holdings' calculation of Closing Net Working Capital and the Non-Timber Purchase Price in accordance with the terms hereof.

22

- 4.2.2 Parent and its accountants shall have 30 days after delivery by Holdings of the Closing Balance Sheet and the Closing Statement to review such financial statements. During such 30-day period, Parent and its representatives shall have full access to all information used by Holdings, and all personnel (including outside accountants) involved, in preparing the Closing Balance Sheet, the Closing Statement and in calculating the Closing Net Working Capital and the Non-Timber Purchase Price, including work papers of Holdings' accountants (subject to Sellers executing any waivers or indemnifications reasonably required by Holdings' accountants related to review of such work papers). If, as a result of such review, Parent determines that the Closing Balance Sheet, or Closing Net Working Capital or the Non-Timber Purchase Price, was not prepared in accordance with the requirements of this Agreement, it shall, prior to the expiration of such 30 day period, provide Holdings with a written statement of its objections (an "Objection Statement"), setting forth a specific description of the basis of each objection and the adjustments to the Closing Balance Sheet or the calculations of Closing Net Working Capital or the Non-Timber Purchase Price which Parent seeks to have made. Holdings shall then have 30 days to review and respond to the Objection Statement. In reviewing any Objection Statement, Holdings and its accountants shall have full access to all information used by the Sellers, and all personnel (including outside accountants) involved, in preparing the Objection Statement, including work papers of Sellers' accountants (subject to Holdings executing any necessary waivers or indemnifications reasonably required by Sellers' accountants related to review of such work papers).
- 4.2.3 The parties shall attempt in good faith to reach an agreement with respect to all disputed items set forth in Parent's Objection Statement. All discussions in connection with such attempts shall be governed by Rule 408 of the Federal Rules of Evidence or any similar state or local rule of evidence that may be applicable to any settlement discussions, and all settlement discussions shall be inadmissible in any proceeding before or submission to the Independent

Accountants resulting pursuant to this Section 4.2. If the parties are unable to resolve all of their disagreements with respect to any of such items within 45 days following the delivery of the Objection Statement, they shall refer their remaining differences to the Boise office of Deloitte & Touche LLP (the "Independent Accountants"), who shall, acting as experts and not as arbitrators, determine in accordance with this Agreement a resolution of the items referred to them. The parties shall each direct the Independent Accountants to use their best efforts to render their determination within 30 days after such submission. The Independent Accountants' determination shall be conclusive and binding upon the parties with respect to the items referred and shall not be subject to appeal except for manifest error. The fees and disbursements of the Independent Accountants shall be borne equally by Parent and Holdings. The parties shall make readily available to the Independent Accountants all relevant books and records of the Business and any work papers generated by the parties (or their accountants) in their preparation and review of the Closing Balance Sheet, the Closing Statement and the Objection Statement together with any other items reasonably requested by the Independent Accountants.

4.2.4 In the event that no Objection Statement is delivered to Holdings during the 30 day period provided for in Section 4.2.2 or upon resolution of all issues raised by such Objection Statement pursuant to the procedures of Section 4.2.3, the Closing Balance Sheet and Closing Statement shall be deemed final (with the changes, if any, required by the outcome of the objection process).

23

4.2.5 Within ten days after the Closing Balance Sheet is deemed final and the calculations of Closing Net Working Capital and Non-Timber Purchase Price have been finally determined, (i) in the event that the Estimated Non-Timber Purchase Price is less than the Non-Timber Purchase Price (as finally determined), Holdings shall, or shall cause one or more of its Subsidiaries to, make payment to Parent in an amount equal to such shortfall or (ii) in the event that the Estimated Non-Timber Purchase Price is greater than the Non-Timber Purchase Price (as finally determined), Parent shall, or shall cause one or more of its Subsidiaries to, make payment to Holdings in an amount equal to such excess. Such payment, together with interest thereon from the date of Closing to the date of payment at the Interest Rate, shall be made by wire transfer of immediately available funds to a bank account of the party to receive the payment designated by such party.

# Article 5 Liabilities

- 5.1 <u>Assumed Liabilities</u>. At the Closing, Holdings and/or one or more Permitted Affiliate Purchasers shall assume all of the Assumed Liabilities (as defined below). Assumed Liabilities directly relating to the Timberlands, including the obligations of Parent and Southern under the Timberlands Leases, shall be assumed by Timber Co. The "Assumed Liabilities" shall consist of all debts, liabilities, commitments or obligations of Sellers exclusively or primarily related to their conduct of the Business at the Facilities, other than the Retained Liabilities, including, without duplication, the following (and in the case of items described in 5.1.1 through 5.1.6, subject to the limitations stated therein):
- 5.1.1 All undischarged current liabilities and obligations of Sellers which arose in the ordinary course of and are related to the operation of the Business prior to the Closing (including Trade Intercompany Accounts) and which are recorded pursuant to the Accounting Principles on the Closing Balance Sheet and included in the calculation of Closing Net Working Capital, but only to the extent of the dollar amount reflected therein;
- 5.1.2 All liabilities and obligations of Sellers arising from and after the Closing under the Contracts, Leases and Governmental Authorizations that are identified, or by the express terms of this Agreement not required to be identified, in Parent's Disclosure Memorandum (other than any Contracts, Leases and Governmental Authorizations that are listed or described on Schedule 5.3 as a Retained Liability), but only to the extent such Contracts, Leases and Governmental Authorizations are assigned to Holdings or a Permitted Affiliate Purchaser and excluding any liability or obligation arising out of or in connection with any breach thereof occurring prior to, at, or as a result of the Closing;
- 5.1.3 Liabilities and obligations relating to the employee benefit plans, workers' compensation, and Transferred Employees in accordance with, but only to the extent required by, Article 11;
  - 5.1.4 The obligations of Sellers in respect of the Transferred Litigation;
- 5.1.5 Straddle Environmental Regulatory Issues and Straddle Hazardous Substance Releases (including Straddle Environmental Regulatory Issues and Straddle Hazardous Substance Releases Disclosed on Schedule 7.11 of Parent's Disclosure

24

Memorandum), in either case solely to the extent arising as a result of the operation of the Business by Purchasers in violation of Permits and Licenses or other regulatory requirements under Environmental Laws immediately after Closing, but excluding any Retained Environmental Liabilities; and

5.1.6 Liabilities and obligations of the Business solely to the extent expressly agreed to be assumed by Holdings and its Subsidiaries under the terms of this Agreement and the Ancillary Agreements.

At Closing, Holdings shall and/or shall cause one or more of its Subsidiaries to, execute any instrument reasonably requested by Parent to reflect such assumption. After the Closing, Holdings shall, or shall cause one or more of its Subsidiaries to, fully perform and discharge the Assumed Liabilities.

# 5.2 [Reserved.]

- 5.3 Nonassumption of Other Liabilities. Except as set forth in Section 5.1 of this Agreement, Holdings and its Subsidiaries do not, and will not be obligated to, assume any other debt, obligation, liability, commitment or duty of any of Sellers or of any of their respective Subsidiaries of any form or nature, absolute or contingent, known or unknown (the "Retained Liabilities"). Without limiting the generality of the foregoing, and notwithstanding anything to the contrary in Section 5.1, neither Holdings nor any of its Subsidiaries shall assume or become liable for any of the Retained Liabilities set forth on Schedule 5.3. At Closing, Parent shall and shall cause its Subsidiaries to execute any instrument reasonably requested by Holdings to reflect Parent's continued retention of the Retained Liabilities.
- 5.4 <u>Consents to Assignment</u>. Parent shall use its best efforts to obtain any required consent to the assignment of the Contracts, Leases, and Governmental Authorizations prior to the Closing; provided that Parent shall not be obligated or permitted (without Holdings' written consent) to agree to any change in the terms of the transferred arrangement which is material in the context of the particular item being transferred. Holdings shall reasonably cooperate with Parent to obtain such consents, but shall not be obligated to make any payment to obtain any consent to assignment. With respect to the specific Contracts, Leases and Governmental Authorizations, other than the Required Consents, if Parent is unable to obtain consent to the assignment prior to Closing, the Contract, Lease or Governmental Authorization in question shall be deemed a Retained Asset and any obligation or liability arising therefrom shall be excluded from the Assumed

Liabilities. The parties shall use their best efforts to arrange and enter into mirror image purchase and sale arrangements, subleases, sublicenses, service agreements and other work-around arrangements in respect of any Contract or Lease so excluded; <u>provided</u> that to the extent, despite the exercise of best efforts, such arrangements cannot be entered into or would not place Holdings and its Subsidiaries in substantially the same economic position as if such Contract, Lease or Governmental Authorization were able to be assigned, Parent will indemnify Holdings and its Subsidiaries for the aggregate amount of any costs, expenses, losses, or diminution or loss of value resulting from the exclusion of any such Contract, Lease or Governmental Authorization.

25

#### Article 6 Reinvestment By Parent and Parent Sub

At Closing, Parent and Parent Sub shall reinvest the Reinvestment Amount in the Business on the terms and conditions set forth in this Article 6.

- Subsidiary of Parent, its Series A Common Units and its Series B Common Units. The consideration payable by Parent for the Series A Common Units shall be \$36,367,347 and the consideration payable by Parent for the Series B Common Units shall be \$60,061,224. Subject only to the following sentence, the price per Series B Common Unit paid by Parent or such wholly-owned Subsidiary of Parent shall be the same as that paid by Holdings for its Series B Common Units. Operating Holding Co. shall be entitled to issue equity instruments (including units, other equity interests, or options on units or equity interests) to Holdings to reflect management equity issued at Holdings so that both MDP and Parent proportionately bear any dilution caused by such management equity. At the time of issuance, the Series B Common Units acquired by Parent shall constitute less than 20% of the outstanding Series B Common Units and the aggregate of all equity interests acquired by Parent shall be less than 50% of the aggregate equity interests of Operating Holding Co then outstanding. The consideration payable for the Series A Common Units and Series B Common Units to be acquired by Parent or its wholly-owned Subsidiary shall be paid by Parent at Closing by wire transfer of immediately available funds to an account designated by Holdings on behalf of Operating Holding Co.
- Issuances of Timber Holdings Capital Stock. Holdings shall cause its subsidiary, Timber Holdings, the parent company of Timber Co., to issue to Parent Sub, its Series A Common Stock, and its Series B Common Stock. The consideration payable by Parent Sub for the Series A Common Stock shall be \$29,632,653 and the consideration payable by Parent for the Series B Common Stock shall be \$48,938,776. Subject only to the following sentence, the capital stock acquired by Parent will be issued at the same price per share of Series B Common Stock ) as the Series B Common Stock in Timber Holdings acquired by Holdings at the Closing, subject only to adjustment in the case of the Series B Common Stock for management equity issued at Holdings. Timber Holdings shall be entitled to issue equity instruments (including common or preferred stock or options on common or preferred stock) to Holdings to reflect management equity issued at Holdings so that both MDP and Parent Sub proportionately bear any dilution caused by such management equity. At the time of issuance, the Series B Common Stock acquired by Parent Sub shall constitute less than 20% of the outstanding Series B Common Units and the aggregate of all equity interest acquired by Parent Sub shall be less than 50% of the aggregate equity interests of Timber Holdings then outstanding. The consideration payable for the Series A Common Stock and Series B Common Stock to be acquired by Parent Sub shall be paid by Parent Sub at Closing by wire transfer of immediately available funds to an account designated by Holdings on behalf of Timber Holdings.
- 6.3 <u>Equity Agreements</u>. Holdings and Parent shall enter into a securityholders agreement in respect of Operating Holding Co., and Holdings and Parent Sub shall enter into a stockholders agreement in respect of Timber Holdings, in form and substance substantially similar to the form attached as Schedule 6.3.

26

# Article 7 Sellers' Representations and Warranties

Sellers jointly and severally represent and warrant to Purchasers that:

- 7.1 Organization and Qualification. Parent, Minidoka, and each of the Transferred Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware (or in the case of certain of the Transferred Subsidiaries, the jurisdiction indicated after their name on Schedule D-12). Southern is a partnership organized under the laws of Louisiana which is composed of Parent and Minidoka. Sellers and each Transferred Subsidiary have all requisite power and authority to own and operate the Assets and to carry on the Business as currently conducted. Sellers and each of the Transferred Subsidiaries are duly qualified to do business and are in good standing as a foreign entity in all jurisdictions in which such qualification is required to permit them to own their respective assets and to conduct their portion of the Business as it is currently being conducted. Parent Sub is a wholly-owned Subsidiary of Parent that does not own, lease or license any of the Timberlands.
- 7.2 <u>Corporate Authorization</u>. Each Seller and each Transferred Subsidiary have all necessary power and authority to execute and deliver this Agreement and each of the Ancillary Agreements to which it is a party, and to perform their respective obligations hereunder and thereunder. The execution, delivery, and performance by each Seller and each Transferred Subsidiary of this Agreement and each of the Ancillary Agreements to which it is a party have been duly and validly authorized, and no additional authorization or Consent is required in connection with the execution, delivery, and performance by them of this Agreement and each of the Ancillary Agreements to which it is a party.
- 7.3 Consents and Approvals. Except as Disclosed on Schedule 7.3 of Parent's Disclosure Memorandum, no Consent is required to be obtained by any Seller or any Transferred Subsidiary from, and no notice or filing is required to be given by any Seller or any Transferred Subsidiary to, or made by any Seller with, its shareholders or any Governmental Authority. Except as specifically set forth in Schedule 7.14(c) or 7.15(b) of Parent's Disclosure Memorandum, no Consent is required to be obtained by any Seller or any Transferred Subsidiary from, and no notice or filing is required to be given to, any counterparty under any Contract or Lease listed, or required to be listed, on Schedule 7.14(a) or 7.15(a) of Parent's Disclosure Memorandum, in connection with the execution, delivery, and performance by any Seller or any Transferred Subsidiary of this Agreement and each of the Ancillary Agreements to which it is a party.
- 7.4 <u>Non-contravention</u>. Except as Disclosed on Schedule 7.4 of Parent's Disclosure Memorandum, the execution, delivery and performance by any Seller and/or any Transferred Subsidiary of this Agreement and each of the Ancillary Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, does not and will not (i) violate any provision of the certificate of incorporation or bylaws of Parent or Minidoka or any Transferred Subsidiary or the partnership agreement of Southern; (ii) subject to obtaining the Consents specifically identified on Schedules 7.3, 7.14(c) and 7.15(b) of Parent's Disclosure Memorandum, conflict with, or result in the breach of, or constitute a default under, or result in the termination, cancellation or acceleration (whether after the filing of notice or the lapse of time or both) of any right or obligation of any Seller or any Transferred Subsidiary under, or give

rise to a loss of any benefit to which such Seller or such Transferred Subsidiary is entitled under, any material Contract or Lease included in the Assets or result in the creation of any Encumbrance upon any of the Assets; or (iii) assuming compliance with the matters set forth in Schedule 7.3 of Parent's Disclosure Memorandum, violate, or result in a breach of or constitute a default under, any law, rule, regulation, judgment, injunction, order, decree or other restriction of any court or governmental authority to which any Seller or any Transferred Subsidiary is subject.

7.5 <u>Binding Effect</u>. This Agreement constitutes, and each of the Ancillary Agreements when executed and delivered by the parties thereto will constitute, a valid and legally binding obligation of each Seller and each Transferred Subsidiary party thereto, enforceable against each of them in accordance with its terms, subject to bankruptcy, insolvency, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles.

# 7.6 <u>Financial Statements; Absence of Certain Changes.</u>

- 7.6.1 Schedule 7.6.1 contains (or, with respect to those financial statements identified on Schedule 7.6.1 as forthcoming (which upon delivery will be reasonably acceptable to Holdings), will within five working days from the date hereof contain) true and correct copies of the Historical Financial Statements and will upon their completion in accordance with Section 10.13 contain true and correct copies of the Subsequent Financial Statements, the Audited Financial Statements, the Stub Period Financial Statements and the Additional Financial Statements. The "Historical Financial Statements" are (x) Parent's internal operating balance sheet, profit and loss statement, and cash flow statement for each of the Operating Units and for Parent's aviation division, transportation operations (including the operations of Parent's Subsidiaries, Minnesota, Dakota and Western Railroad Company and BCT, Inc.) and its cottonwood farm operations, at and for the periods ending on December 31, 2001, December 31, 2002, and December 31, 2003, together with the same statements for the 2004 year to date periods ending May 31, 2004 and June 30, 2004 and comparable statements for the year to date periods ending on May 31, 2003 and June 30, 2004, and (y) a balance sheet and a profit and loss statement for Parent's corporate staff departments at and for each of the periods provided for above.
- 7.6.2 Except as Disclosed on Schedule 7.6.2 of Parent's Disclosure Memorandum, the Historical Financial Statements and the Subsequent Financial Statements fairly present (or, in the case of the Subsequent Financial Statements, will on their delivery to Holdings fairly present), in all material respects, the results of operations for the portions of the Business they report on and the financial condition of the portions of the Business they report on as of the dates, and for the periods, thereof. The Historical Financial Statements and the Subsequent Financial Statements were (or, in the case of the Subsequent Financial Statements, will be) prepared from and are, or in the case of the Subsequent Financial Statements will be, consistent with the financial records of the Business which are maintained by Parent in accordance with the Historical Accounting Procedures. Such financial records are true, accurate and complete in all material respects and are a portion of the books and records from which Parent's audited financial statements are prepared. The Historical Accounting Procedures are consistent with the accounting procedures used by Parent in preparation of Parent's audited financial statements as filed with the Securities and Exchange Commission.

28

- 7.6.3 The Audited Financial Statements, the Stub Period Financial Statements and the Additional Financial Statements will, upon their delivery to Holdings, fairly present in all material respects, the results of operations for the Business (or applicable portion thereof reported on) and the financial condition of the Business (or applicable portion thereof reported on) stated and presented in compliance with GAAP and, except to the extent disclosed therein, on a basis consistent with the Historical Accounting Procedures.
- 7.6.4 The Inventory shown on the Historical Financial Statements and the Subsequent Financial Statements was (or will be in the case of the Subsequent Financial Statements) determined in accordance with Parent's Historical Accounting Procedures, and the Inventory shown on the Audited Financial Statements, Stub Period Financial Statements and the Additional Financial Statements will when delivered be determined in accordance with GAAP, in each case stated at the lower of cost or market.
- 7.6.5 Subject to the next two sentences of this Section 7.6.5, all Accounts Receivable reflected on the Historical Financial Statements were, and all Accounts Receivable to be included as Current Assets in the computation of Closing Net Working Capital are, bona fide receivables, accounted for in accordance with Historical Accounting Procedures (and, in the case of Accounts Receivable included in Closing Net Working Capital, the Accounting Principles), and subject to no offsets or counterclaims (other than ordinary course returns and allowances for which reserves have been, or will be, recorded to the extent required by the Historical Accounting Procedures and GAAP), representing amounts due with respect to actual transactions in the operation of the Business; it being understood that this representation shall not be deemed to constitute a warranty or guaranty that all such accounts receivable shall be collected. Throughout the period covered by the Historical Financial Statements, Parent sold substantially all of such accounts receivable to a wholly owned subsidiary which, in turn, resold a fractional undivided interest in such accounts receivable to third parties in a securitization facility. Such sales are not reflected in the Historical Financial Statements or the Subsequent Financial Statements. Such sales will be discontinued 45 days prior to the date scheduled for Closing.
- 7.6.6 Except as Disclosed on Schedule 7.6.6 of Parent's Disclosure Memorandum, since March 31, 2004, Sellers and their Subsidiaries have conducted the Business in the ordinary and usual course and, other than in the ordinary and usual course, have not, with respect to the Business: (i) sold, assigned, pledged, hypothecated or otherwise transferred any material portion of the Assets (other than sales of Inventory in the ordinary course of business, or sales of Accounts Receivable pursuant to the securitization facility referred to in Section 7.6.5 not less than 45 days prior to Closing); (ii) terminated or materially amended any Contracts or Leases that are individually or in the aggregate material to the Business; (iii) suffered any extraordinary damage, destruction or other casualty loss; (iv) except for normal salary administration, and ordinary course promotions for Employees, increases pursuant to collective bargaining agreements, or other compensation increases (including bonuses), in each case in the ordinary course of business, increased the compensation payable or to become payable by Sellers or the Transferred Subsidiaries to any of the Employees or increased any bonus, insurance, pension or other employee benefit plan, payment or arrangement made by Sellers or the Transferred Subsidiaries, for or with any of the Employees; (v) took any action or failed to any action, which if taken or failed to be taken after the date of this Agreement and prior to the

29

Closing, would require the consent of Holdings pursuant to Section 10.2.1 or Section 10.2.2 or (vi) entered into an agreement to do any of the foregoing.

# 7.7 <u>Litigation and Claims</u>. Except as Disclosed on Schedule 7.7 of the Parent's Disclosure Memorandum:

7.7.1 There is no action (whether civil, criminal or administrative), suit, demand, claim, dispute, hearing, proceeding (including condemnation or other proceeding in eminent domain) or investigation pending, or to the Knowledge of Parent threatened, related to or arising out of the Business or any of the Assets or included in the Assumed Liabilities except for workers' compensation claims, collection matters, and unemployment compensation claims not reasonably expected individually to exceed \$500,000 or \$2,000,000 in the aggregate and for which, in the case of workers' compensation claims, the liability therefor has been included in the Workers Compensation Reserve.

- 7.7.2 None of the Sellers (with respect to the Business), the Transferred Subsidiaries, or the Assets is subject to any order, writ, judgment, award, injunction, or decree or settlement enforceable in any court or governmental or regulatory authority of competent jurisdiction or any arbitrator or arbitrators except for regulatory decrees and orders of general applicability to both the Business and others conducting similar businesses in the affected jurisdiction which do not and could not reasonably be expected to materially impair the operation of the Business or materially detract from the value of the Business.
  - 7.8 Taxes. Except as Disclosed on Schedule 7.8 of Parent's Disclosure Memorandum:
- 7.8.1 Sellers and each of their Affiliates (including each of the Transferred Subsidiaries) have duly and timely filed (or have caused to be duly and timely filed) each Tax Return required to be filed with any Tax Authority which includes or is based upon the Assets, or the operations, ownership or activities of the Business; each of such Tax Returns is complete and accurate in all material respects; and all Taxes due and payable (whether or not shown or required to be shown on such Tax Returns) have been or will be paid by Sellers and their Affiliates (including each of the Transferred Subsidiaries) prior to their due dates.
- 7.8.2 None of the Transferred Subsidiaries is obligated to make any payments (or is a party to any agreement that could obligate it to make any future payments), and none of the Assumed Liabilities is an obligation to make any payments, that will not be deductible under Sections 162(m) or 280G of the Code.
- 7.8.3 None of the Assets or the assets of the Transferred Subsidiaries (i) is subject to any lien arising in connection with any failure or alleged failure to pay any Tax, (ii) secures any debt the interest on which is Tax-exempt under Section 103(a) of the Code, (iii) is required to be or is being depreciated under the alternative depreciation system under Section 168(g)(2) of the Code, (iv) is "limited use property" within the meaning of Revenue Procedure 2001-28, or (v) will be treated as owned by any other person pursuant to the provisions of former Section 168(f)(8) of the Code.

30

- 7.8.4 Sellers (with respect to the Business) and the Transferred Subsidiaries have withheld and paid each Tax required to have been withheld and paid in connection with amounts paid or owing to any Employee, independent contractor, creditor, shareholder, or other party.
- 7.8.5 There are no pending, threatened, or proposed audits, assessments or claims from any Tax Authority for deficiencies, penalties, or interest against Sellers (with respect to the Assets or the Business), any of the Transferred Subsidiaries or any of their assets, operations, or activities.
- 7.8.6 No Transferred Subsidiary owns, directly or indirectly, and none of the Assets consists of any interest in, any entity classified as a partnership for United States federal income tax purposes.
- 7.8.7 No claim has ever been made by a Tax Authority in a jurisdiction where any of the Sellers or the Transferred Subsidiaries do not file Tax Returns that such Seller or Transferred Subsidiary is or may be subject to Taxes assessed by such jurisdiction.
- 7.8.8 None of the Transferred Subsidiaries will be required (A) as a result of a change in method of accounting for a taxable period ending on or prior to the Closing Date, to include any adjustment in taxable income for any taxable period (or any portion thereof) ending after the Closing Date, (B) as a result of any "closing agreement," as described in Section 7121 of the Code (or any similar provision of state, local or foreign income Tax law), to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date, or (C) as a result of any payment received prior to the Closing or any installment sale prior to the Closing to include any item of income in any taxable period (or portion thereof) ending after the Closing Date.
- 7.8.9 None of the Transferred Subsidiaries has distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 or Section 361 of the Code.

# 7.9 <u>Employees and Employee Benefits</u>.

- 7.9.1 All of the Employees based in the United States are employed by Parent except for 42 Employees who are employed by Minnesota, Dakota & Western Railway Company. None of the Transferred Subsidiaries have Employees except for Minnesota, Dakota & Western Railway Company, Boise Cascade do Brasil LTDA, Boise Alljoist Ltd., and Boise Building Products Limited, and none of the Retained Subsidiaries have any Employees. Upon Holdings' request, Parent will promptly provide a list of each Employee who is on disability leave as of any date after the date of this Agreement and prior to the Closing Date and also as of the Closing Date.
- 7.9.2 Schedule 7.9.2(a) of Parent's Disclosure Memorandum lists each Parent Plan, complete and accurate copies of which have been provided (or made available in the dataroom) to Holdings; <u>provided</u>, however, with respect to Parent Plans providing benefits to employees outside the United States, if a complete and accurate copy of such plans has not been

31

provided to Holdings, then summaries of all material terms (including, without limitation, whether any unfunded liabilities with respect to such plan exist) have been provided to Buyer. Except as set forth on Schedule 7.9.2(b) of Parent's Disclosure Memorandum, no Parent Plan provides for post-employment welfare benefits to any Employee other than as required under Section 4980B of the Code. Except as Disclosed on Schedule 7.9.2(c) of Parent's Disclosure Memorandum, each Parent Plan has been and is being maintained in substantial compliance with all applicable laws, including ERISA and the Code, and each plan that is intended to be qualified under Section 401(a) of the Code is so qualified and has received a determination of such qualification from the Internal Revenue Service. Except as Disclosed on Schedule 7.9.2(d) of Parent's Disclosure Memorandum, no Employee is covered by a "multiemployer plan" as defined in Section 4001(a)(3) of ERISA or has been covered by such a plan during the past five years. Except as provided in Section 11.8 with respect to Holdings' General Severance Policy and the 2004 Severance Agreement, none of the Parent Plans will obligate Holdings or any Permitted Affiliate Purchaser or any of their Subsidiaries or any Transferred Subsidiary to pay separation, severance, termination or similar-type benefits in whole or in part as a result of any transaction contemplated by this Agreement or as a result of a "change in control" (as such term is defined in Section 280G of the Code) arising from the execution and performance of this Agreement. No liability to the PBGC (except for routine payment of premiums), Internal Revenue Service, Department of Labor, or otherwise has been or is expected to be incurred with respect to any "employee benefit plan" (as such term is defined under Section 3(3) of ERISA) that is subject to Title IV of ERISA which could result in a liability to Holdings or any Permitted Affiliate Purchaser or any of their Subsidiaries or any Transferred Subsidiary. No "change in

7.10 <u>Compliance with Laws; Governmental Authorizations.</u> Except as Disclosed on Schedule 7.10 of Parent's Disclosure Memorandum, the Business is being conducted in substantial compliance with all laws applicable to the Business, and Sellers and the Transferred Subsidiaries have all material Governmental

Authorizations required for the ongoing conduct of the Business where and as it is currently being conducted. Parent has no Knowledge of any intention or commencement of any process by or before any Governmental Authority to suspend, revoke, materially modify (other than modifications made in connection with a normal renewal of an expiring Governmental Authorization) or cancel any such Governmental Authorization, or any claims by or before any Governmental Authority alleging or investigating any violation of law. Notwithstanding the foregoing, it is understood and agreed by the parties that nothing in this representation is intended to address any compliance issue that is the subject of the representations and warranties set forth in Sections 7.7, 7.8, 7.9, 7.11, 7.12, or 7.13 hereof, and that Sellers make no representations as to the transferability or assignability of any such Governmental Authorization.

- 7.11 Environmental Matters. Except as Disclosed on Schedule 7.11 of Parent's Disclosure Memorandum:
- 7.11.1 Sellers and the Transferred Subsidiaries have complied at all times during the past five years with, and are currently in compliance with, all applicable Environmental Laws

32

applicable to the Business, and there are no liabilities under any Environmental Law with respect to the Business;

- 7.11.2 None of Sellers nor any of the Transferred Subsidiaries have received any written notice of any material violation or alleged material violation of, or any material liability under, any Environmental Law in connection with the Business during the past five years;
- 7.11.3 There are no material writs, injunctions, decrees, orders or judgments outstanding, or any actions, suits, proceedings or investigations pending or, to the Knowledge of Parent, threatened, relating to compliance with or liability under any Environmental Law affecting the Business or the Assets;
- 7.11.4 None of Sellers nor any of the Transferred Subsidiaries nor the Business has any material or potentially material liability with respect to the design, manufacture, sale or distribution of products containing asbestos or any other Hazardous Substance prior to Closing or with respect to the presence of asbestos or silica in any product or item in, on or beneath any property, facility or structure (including, without limitation, the Facilities or the Timberlands) and no damaged and friable asbestos-containing material is currently present at the Facilities or the Timberlands; and
- 7.11.5 Sellers and the Transferred Subsidiaries have made available to Holdings true and correct copies of all environmental audit reports, action plans and updated action plans and Environmental Assessments covering the Facilities and the Timberlands prepared by Sellers in the three year period preceding the date of this Agreement.

#### 7.12 <u>Intellectual Property.</u>

- 7.12.1 Schedule 7.12.1 of Parent's Disclosure Memorandum sets forth a list and description (including the country of registration) of all registered Parent Intellectual Property currently (or, to the Knowledge of Parent, within the last 12 months) used in the Business. Except as Disclosed, no third party has rights in or otherwise has the right to restrict Parent's use of such Parent Intellectual Property, and, to Parent's Knowledge (without any inquiry), no third party has rights in, or otherwise has the right to restrict Holdings' and its Subsidiaries use of such Parent Intellectual Property and the Boise Marks as of and following the Closing.
- 7.12.2 To the Knowledge of Parent, no product (or component thereof) or process currently used, sold, or manufactured by the Business infringes on, misappropriates, or otherwise violates a valid and enforceable intellectual property right of any other Person. Any pending or threatened claim of such infringement Known to Parent is set forth on Schedule 7.12.2 of Parent's Disclosure Memorandum.
- 7.12.3 There are no actions or proceedings pending or, to the Knowledge of Parent, threatened, challenging, and, to the Knowledge of Parent, no person is infringing or otherwise violating, the Parent Intellectual Property, except for challenges, infringements or violations which, individually or in the aggregate, would not have a Material Adverse Effect.

33

- 7.12.4 All of the Parent Intellectual Property used in the Business and all of Parent's and its Subsidiaries' interest in the Boise Marks will be transferred to Holdings at Closing, and such transferred Parent Intellectual Property and the Boise Marks will be available to Holdings after Closing on the same terms and conditions under which it was available to Parent and its Subsidiaries prior to the Closing. The Boise Marks will remain available for use by Parent in its Boise Office Solutions business for the period and to the extent permitted in Section 10.6.
  - 7.13 <u>Labor Matters</u>. Except as Disclosed on Schedule 7.13 of Parent's Disclosure Memorandum:
- 7.13.1 None of Sellers nor any Transferred Subsidiary is a party to any labor or collective bargaining agreement with respect to Employees, no Employees are represented by any labor organization and, to the Knowledge of Parent, there are no organizing or decertification activities (including any demand for recognition or certification proceedings pending or threatened to be brought or filed with the National Labor Relations Board or other labor relations tribunal) involving any of the Employees;
- 7.13.2 There are no strikes, work stoppages, slowdowns, lockouts, or unfair labor practice charges pending or, to the Knowledge of Parent, threatened against or involving the employees of the Business;
- 7.13.3 There are no complaints, charges, claims or grievances against Sellers or any of the Transferred Subsidiaries pending or, to the Knowledge of Parent, threatened to be brought or filed with any governmental authority, arbitrator or court based on or arising out of the employment of any Employee by Sellers or any of the Transferred Subsidiaries, where the aggregate potential liability to Sellers or the Transferred Subsidiaries from such matter if adversely determined is at least \$75,000; and
- 7.13.4 Each Seller and each Transferred Subsidiary are in substantial compliance with all laws relating to the employment of labor, including all such laws relating to wages, hours, collective bargaining, discrimination, civil rights, safety and health, immigration, workers' compensation, layoffs, and the collection and payment of withholding and/or Social Security Taxes and similar Taxes except where such noncompliance would not have a Material Adverse Effect.
- 7.14 <u>Contracts</u>. Schedule 7.14(a) Parent's Disclosure Memorandum contains a list of (a) each Contract which has a remaining unexpired term of at least one year and requires payments by or to the Business of more than \$1,000,000 in any year; (b) all Contracts which require payment (other than for the provision of goods or services in the ordinary course of business) by or to the Business of more than \$1,000,000; (c) any Contract that contains a non-compete covenant or similar provision that could restrict Holdings in its conduct of the Business following Closing, (d) any consulting, retention, severance, bonus, compensation, or similar

agreement with any Employee, other than the Parent Plans, and any employment agreement with any Employee (other than temporary employment contracts that are terminable without penalty at any time at the will of the employer with a term of less than one year and at an annual compensation rate of less than \$75,000 and that provide for no payments of severance), (e) all

collective bargaining agreements affecting any of the Employees, (f) any Contract between any of Sellers or any Retained Subsidiary on one hand and any of Sellers, any Retained Subsidiary, or any Transferred Subsidiary on the other hand, which is related to the Business, other than purchase orders entered into in the ordinary course of business and included in Trade Intercompany Accounts, and (g) any other Contract that is material to the Business or the Assets. Each Contract required to be disclosed on Schedule 7.14(a) of Parent's Disclosure Memorandum is a valid and binding agreement of Sellers or a Transferred Subsidiary and, to the Knowledge of Parent, is in full force and effect. Except as Disclosed on Schedule 7.14(b) of Parent's Disclosure Memorandum, none of the Sellers or any Transferred Subsidiary is in, and, to Parent's Knowledge, no other party thereto is in, default in any material respect under any Contract required to be disclosed in such Schedule. Parent's Disclosures made pursuant to this Section 7.14 include all Contracts that are material to the Business. Solely for purposes of this Section, the term Contract is deemed to also include any such instrument which is held by a Transferred Subsidiary rather than one of Sellers. Schedule 7.14(c) of Parent's Disclosure Memorandum also identifies which of the Contracts listed thereon may not be transferred to Holdings and its Subsidiaries (including indirectly by transfer of the stock of the Transferred Subsidiaries) without the consent of the counterparty thereto or for which rights or obligations of any party thereto will (with or without the passage of time), without the consent of the other party(ies) thereto, terminate, accelerate, be modified or otherwise become due.

Leases. Schedule 7.15(a) of Parent's Disclosure Memorandum sets forth a list of: (i) each Timberlands Lease; (ii) each Equipment Lease with a remaining term of more than one month and rental payments in excess of \$1,000,000 in the aggregate for such lease or which covers equipment with an original purchase cost in excess of \$5,000,000; and (iii) each Facilities Lease, together with, for each such Timberlands Lease, the general location and approximate acreage, and for each such Facilities Lease, the address of each parcel of real property subject to such Lease. Sellers have delivered to Holdings true and correct copies of each such Lease, other than the following Leases, true and correct copies of which will be delivered within ten working days after the date hereof: (i) resource management agreements marked with an asterisk on Schedule 7.15, each of which is substantially similar to the form of resource management agreement previously delivered to Holdings, and which are not (individually or in the aggregate) materially adverse to Holdings and its Subsidiaries, and (ii) timber cutting contracts marked with an asterisk on Schedule 7.15, as to each of which Schedule 7.15 describes the acreage covered thereby and the unexpired term thereof, and which are not (individually or in the aggregate) materially adverse to Holdings and its Subsidiaries. Each of the Leases is a valid and binding agreement of Parent or Southern, as the case may be, and is in full force and effect. There are no defaults under any leases which defaults have not been cured or waived and which would, individually or in the aggregate, have a Material Adverse Effect. None of the Transferred Subsidiaries holds any Timberlands Leases or any Equipment Lease or Facilities Lease which would meet the threshold for disclosure on such Schedule. Parent's Disclosures made pursuant to this Section 7.15 include all Leases that are material to the Business. Solely for purposes of this Section 7.15, the terms Timberlands Lease, Equipment Lease, and Facilities

35

(with or without the passage of time), without the consent of the other party(ies) thereto, terminate, accelerate, be modified or otherwise become due.

# 7.16 <u>Entire Business; Title to Property; Capital Structure of Transferred Subsidiaries.</u>

7.16.1 Except as Disclosed on Schedule 7.16.1 of Parent's Disclosure Memorandum, the Assets (including the Timberlands, Facilities, and Fixtures and Equipment held pursuant to the Leases, and Intellectual Property covered by the IP Licenses) together with the assets held by the Transferred Subsidiaries, and the rights specifically provided or made available to Holdings and its Subsidiaries under the Ancillary Agreements, include all of the buildings, machinery, equipment and other assets (whether tangible or intangible) necessary and adequate for Holdings and its Subsidiaries immediately after Closing to conduct in all material respects the Business as conducted as of the date hereof and since March 31, 2004. None of the Assets is held by, and no aspect of the Business is conducted by, and (except with respect to the balance sheet and profit and loss statements of the corporate staff operations) the Historical Financial Statements do not include any of the financial condition or results of operations of, any of the Retained Subsidiaries or any other Person other than the Sellers and the Transferred Subsidiaries. Schedule 7.16.1 sets forth a description of any and all assets (whether tangible or intangible, real or personal) used in the conduct of the Business, but which do not fall within the definition of "Assets". Schedule D-13 lists the county and approximate acreage of the Timberlands. Except for the Timberlands and the Retained HBU Lands, none of Sellers or any of the Retained Subsidiaries owns or leases any real property for the purpose of growing or harvesting timber. The letters of credit and performance bonds to be assumed by Holdings and its Subsidiaries at the Closing are substantially as set forth on Schedule 7.16.1 of Parent's Disclosure Memorandum, and in any event the aggregate amount thereof shall not exceed \$110 million.

7.16.2 Except as Disclosed on Schedule 7.16.2 of Parent's Disclosure Memorandum, Sellers and the Transferred Subsidiaries have good (and, in the case of the Realty and the Timberlands (as applicable), marketable) title to the Assets, free and clear of all Financial Encumbrances and all other Encumbrances except (i) liens for Taxes (which are not related to income, sales, or withholding Taxes), assessments and other governmental charges not yet due and payable or due but not delinquent as of the Closing Date or being contested in good faith by appropriate proceedings; (ii) mechanic's, workmen's, repairmen's, warehousemen's, carriers, or other like liens arising or incurred in the ordinary course of business for amounts which are not delinquent and which will not individually or in the aggregate materially impair the operation of the Business, original purchase price conditional sales contracts and equipment leases with third parties entered into in the ordinary course of business; (iii) with respect to the Realty and Timberlands, (A) easements, quasi-easements, licenses, covenants, rights-of-way and other similar restrictions, including any other agreements, conditions, restrictions, or other matters which would be shown by a current title report or other similar report or listing, (B) any conditions that may be shown by a current survey (other than, for purposes of Section 15.1.12, Surveys delivered pursuant to Section 9.1.3), title report or physical inspection, and (C) zoning, building and other similar restrictions, none of which Encumbrances in (A) through (C) materially impairs the uses of the Realty or Timberlands as currently used in the Business; (iv) with respect to the Timberlands, such encroachments, boundary line disputes, access restrictions or absence of

36

periods of time thereafter (including for purposes of Section 2.1 and Article IX), "Permitted Encumbrances" shall not include any Financial Encumbrances. Solely for purposes of this Section 7.16.2, each reference to the Assets and each subcategory thereof is intended to include any such asset held by a Transferred Subsidiary.

- 7.16.3 Parent owns, directly or indirectly, all of the shares of issued and outstanding stock or other equity interests of the Transferred Subsidiaries, and no shares of stock or other equity interests of any indirectly owned Transferred Subsidiary are held by any Retained Subsidiary. All shares of stock or other equity interests of the Transferred Subsidiaries are validly issued, fully paid and non-assessable. Except as Disclosed on Schedule 7.16.3(a) of Parent's Disclosure Memorandum, (i) there are no options, warrants, or similar rights to purchase any of the shares of any of the Transferred Subsidiaries, and no obligations binding upon any Transferred Subsidiary to issue, sell, redeem, purchase or exchange any of its capital stock or any right relating thereto, and (ii) there are no shareholders' agreements, voting agreements, voting trusts or other agreements or rights of third parties with respect to or affecting any of the Transferred Subsidiaries or any of their shares of stock.
- 7.16.4 The Assets and the assets of the Transferred Subsidiaries are in operating condition and repair (subject to normal wear and tear) and are in a condition to allow the continued conduct after the Closing by Holdings of the Business as it is currently conducted in all material respects. With respect to the Assets and the assets of the Transferred Subsidiaries, Parent has not failed to make or deferred or postponed any maintenance, repair, capital improvement, or capital expenditure in anticipation of a potential sale of all or part of the Business, including the transactions contemplated by this Agreement and the Ancillary Agreements. Except as set forth in this paragraph and otherwise set forth in this Agreement or any of the Ancillary Agreements, all of the Assets will be sold AS IS and WHERE IS as to the Sellers.
- 7.16.5 Except as set forth on Schedule 7.16.5 of Parent's Disclosure Memorandum, none of the Sellers or their Retained Subsidiaries or any officer or director thereof, or any individual related by blood, marriage or adoption to any such individual or any entity in which any such Person or individual owns any beneficial interest, is a party to any Contract with any of Sellers or any of the Transferred Subsidiaries or has any interest in any property, asset, or right used by Sellers or any Transferred Subsidiary or necessary for the Business. Schedule 7.16.5 of Parent's Disclosure Memorandum describes all services provided to or for the benefit of the Business by any portion of Sellers not included in the Business and any of the Retained Subsidiaries.

37

- 7.16.6 Since March 31, 2004, Sellers and their Subsidiaries have operated the Timberlands in the ordinary course of business consistent with past practice, including (a) conducting their on-going timber harvest operations on the Timberlands substantially in accordance with the species composition, timber products, volume levels, and timing detailed in Parent's Harvest Plan, and (b) performing necessary and prudent maintenance of roads, drainage systems, fire protection and prevention as well as appropriate stand establishment activities.
- 7.17 <u>Finders' Fees</u>. Except for the fees of Goldman Sachs & Co., whose fees will be paid by Parent, there is no investment banker, broker or finder which has been retained by or is authorized to act on behalf of Parent who might be entitled to any fee or commission from Holdings or any of its Affiliates in connection with the transactions contemplated by this Agreement.
- Insurance. Except as Disclosed on Schedule 7.18 of Parent's Disclosure Memorandum, Schedule D-14 hereto identifies each of the insurance policies currently carried by Parent or any of its Subsidiaries which cover risks associated with or arising out of the Business. The column marked "status" indicates for each policy whether it will be retained by Sellers (those policies labeled as "excluded"); fully assigned to Holdings (those policies labeled as "assigned"); or divided between the parties in accordance with Section 10.1.1 (those policies labeled as "divided"). In addition, the status column indicates which of the assigned or divided policies will name Parent and the Retained Subsidiaries as additional insureds. The policies listed in Schedule D-14 as "excluded" will be retained by Sellers and shall constitute Retained Assets. The policies listed in Schedule D-14 as "divided" shall be included as Assets to be transferred hereunder to the extent of the contemplated assignment and as a Retained Asset to the extent of the carved out retained coverages. For each of the Insurance Policies listed thereon, Schedule D-14 sets forth (a) the name of the insurer, the name of the policyholder, and the name of each covered insured, (b) the scope, period and amount of coverage, and (c) a description of any retroactive premium adjustments or other loss sharing arrangements (including any self-insured retention, deductible or similar self-insurance or co-insurance in respect of each such insurance policy). Parent has paid all premiums or other amounts due and owing to such insurers in respect of each policy listed on Schedule D-14 and is not otherwise in default under any such policy.
- 7.19 No Undisclosed Liabilities. No Seller nor any of the Transferred Subsidiaries has any material obligations or liabilities (whether accrued, absolute, contingent, unliquidated or otherwise, whether or not known to Parent, whether due or to become due and regardless of when asserted) arising out of transactions entered into at or prior to the Closing or any action or inaction at or prior to the Closing or any state of facts existing at or prior to the Closing other than: (i) liabilities recorded or disclosed on the December 31, 2003 balance sheet included in the Historical Financial Statements (including any notes thereto, if any); (ii) liabilities and obligations arising from or in connection with matters Disclosed pursuant to Parent's representations and warranties in this Agreement or in the Disclosure Memorandum (none of which, except as Disclosed pursuant to Section 7.7, is a liability resulting from a breach of contract,

38

breach of warranty, tort, infringement claim or lawsuit); (iii) liabilities and obligations which have arisen after December 31, 2003, in the ordinary course of business (none of which, except as Disclosed pursuant to Section 7.7, is a liability resulting from a breach of contract, breach of warranty, tort, infringement claim or lawsuit); (iv) its long term indebtedness and other long term liabilities which are carried on its books at the corporate level and are not Assumed Liabilities; and (v) any Retained Liabilities.

- 7.20 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in this Article 7, neither Parent nor any other Person makes any other express or implied representation or warranty on behalf of any of Sellers. ANY WARRANTY IMPLIED BY LAW INCLUDING WARRANTIES OF MERCHANTABILITY AND OF FITNESS FOR A PARTICULAR PURPOSE ARE HEREBY DISCLAIMED.
- 7.21 <u>Closing Date</u>. The representations and warranties of Sellers contained in this Article 7 and elsewhere in this Agreement and all information contained in any exhibit, schedule or attachment hereto or in any certificate or other writing delivered by, or on behalf of, any of Sellers to Holdings pursuant to this Agreement shall be true and correct in all respects on the Closing Date as though then made, except as affected by the transactions expressly contemplated by this Agreement. Parent shall have the right, at any time prior to the fifth business day prior to the Closing Date to update the portion of the Parent Disclosure Memorandum prepared in respect of the representations and warranties of Sellers in this Article VII to reflect changes in the Business between the date hereof and the Closing Date that are actions taken by Parent permitted under Section 10.2 without Holdings' prior consent or were actions not permitted thereby which were taken with Holdings' consent.

In the event any litigation or claim against any of the Sellers or their Subsidiaries relating to the Business (other than the Transferred Litigation), or any action, complaint, charge, claim, grievance, or other labor matter of the type described in Section 7.13, arises after the date hereof and prior the Closing, then (a)

Parent will promptly deliver written notice to Holdings of the nature of such matter and will provide to Holdings all information concerning such matter as Holdings reasonably requests; (b) Holdings may elect to assume such matter by delivering written notice of such assumption to Parent, in which case a reserve will be established for such matter (which reserve will be determined in accordance with Parent's historical practices and methods for establishing and adjusting book reserves relating to litigation and other claims liabilities and will be reasonably acceptable to Holdings), and such reserve will be added to Schedule D-6 and included in the Litigation and Claims Reserve; and (c) if Holdings elects not to, or does not, assume such matter in writing as described in clause (b) above, then such matter will be treated as Retained Litigation, and all liabilities and obligations in respect of such matter will be Retained Liabilities.

#### Article 8 Holdings' Representations and Warranties

Holdings represents and warrants to Sellers that:

8.1 <u>Organization and Qualification</u>. Each of Holdings and the Permitted Affiliate Purchasers (other than Timber Co., Timber Holdings, and any other Permitted Affiliate Purchaser that is a corporation) is a limited liability company duly organized, validly existing, and in good standing under the laws of the state of Delaware. Timber Co., Timber Holdings and each other Permitted Affiliate Purchaser that is a corporation, is a corporation duly organized, validly existing, and in good standing under the laws of Delaware. Each of Holdings and each of

39

the Permitted Affiliate Purchasers has all requisite power and authority to own and operate its assets and to carry on its business as currently conducted. Each of Holdings and each of the Permitted Affiliate Purchasers is duly qualified to do business and is in good standing as a foreign entity in all jurisdictions in which such qualification is required to permit it to own its assets and to conduct its business (and, as of the Closing Date, will be so qualified to conduct the Business) as it is currently being conducted.

- 8.2 <u>Authorization</u>. Holdings and the Permitted Affiliate Purchasers have all necessary power and authority to execute and deliver this Agreement and each of the Ancillary Agreements, and to perform their respective obligations hereunder and thereunder. The execution, delivery and performance by each of Holdings and each Permitted Affiliate Purchaser of this Agreement and each of the Ancillary Agreements to which such Person is party have been duly and validly authorized and no additional authorization or consent is required in connection with the execution, delivery and performance by them of this Agreement and each of the Ancillary Agreements to which such Person is party.
- 8.3 <u>Consents and Approvals</u>. Except as Disclosed on Schedule 8.3 of the Holdings Disclosure Memorandum, no Consent is required to be obtained by Holdings or any of the Permitted Affiliate Purchasers from, and no notice or filing is required to be given by Holdings or any of the Permitted Affiliate Purchasers to, or made by Holdings or any of the Permitted Affiliate Purchasers with, its members or shareholders or any Governmental Authority in connection with the execution, delivery and performance by such entities of this Agreement and the Ancillary Agreements.
- 8.4 Non-Contravention. Except as Disclosed on Schedule 8.4 of the Holdings Disclosure Memorandum, the execution, delivery and performance by each of Holdings and each Permitted Affiliate Purchaser of this Agreement and each of the Ancillary Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, does not and will not (i) violate any provision of the certificate of incorporation, bylaws, or other organizational documents of Holdings or of any Permitted Affiliate Purchaser; or (ii) assuming compliance with the matters set forth in Section 8.3, violate, or result in a breach of or constitute a default under any law, rule, regulation, judgment, injunction, order, decree or other restriction of any court or Governmental Authority to which Holdings or any of the Permitted Affiliate Purchasers is subject.
- 8.5 <u>Binding Effect</u>. This Agreement constitutes, and each of the Ancillary Agreements when executed and delivered by the parties thereto will constitute, a valid and legally binding obligation of each of Holdings and each of the Permitted Affiliate Purchasers signatory thereto, enforceable against them in accordance with its terms, subject to bankruptcy, insolvency, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles.
- 8.6 <u>Organizational History</u>. Each of Holdings and each of the Permitted Affiliate Purchasers was newly organized for the purpose of undertaking the transactions provided for in this Agreement, and none of them has conducted any operations or incurred any obligations or liabilities except in connection with such transactions.

40

8.7 <u>Capital Structure</u>. As of immediately after Closing, the ownership structure of Holdings and the Permitted Affiliate Purchasers shall, in all material respects, be as depicted in Schedule 8.7 of the Holdings Disclosure Memorandum and neither Holdings nor any of the Permitted Affiliate Purchasers shall have any Subsidiaries other than as depicted on such Schedule 8.7 of the Holdings Disclosure Memorandum, other than wholly-owned Subsidiaries of Operating Holding Co. and Timber Holdings that are formed for the purpose of consummating the transactions contemplated hereby.

# 8.8 Financial Capability.

- 8.8.1 On the Closing Date, Holdings and its Permitted Affiliate Purchasers will, assuming the making by Parent and Parent Sub on the Closing Date of the Reinvestment and receipt of all of the financing contemplated by the Commitment Letters, have sufficient funds to (i) pay the Non-Timber Purchase Price and the cash portion of the Timberlands Purchase Price and (ii) arrange for provision of the standby letter of credit securing the Timber Notes provided for in the arrangements described in the LC Commitment Letters.
- 8.8.2 Schedule 8.8.2 of the Holdings Disclosure Memorandum includes true and correct copies of (a) commitment letters received by Holdings and its Subsidiaries (the "Debt Commitment Letters") to provide the debt financing for the transactions contemplated hereby and (b) a true and correct copy of the commitment letter received by Holdings and its Subsidiaries (the "Equity Commitment Letter" and, together with the Debt Commitment Letters, the "Commitment Letters") from MDP to provide equity financing for the transactions contemplated hereby. As of the date of this Agreement, the Commitment Letters are in full force and effect and have not been amended or, to Holdings' knowledge, rescinded. Holdings or one or more of its Subsidiaries has paid, or at or prior to Closing will pay, all fees and other amounts required by such Commitment Letters to be paid by it at or prior to the Closing.
- 8.9 <u>Representations Relating to the White Paper Business</u>. Holdings and its Affiliates have not at any time during the two months prior to the date of this Agreement received any proposals from, nor engaged in any discussions with, any Person regarding a potential sale of all or substantially all of the assets of the white paper manufacturing, conversion, distribution and sales operations of the Business, and as of the date hereof Holdings and its Affiliates are not aware of any reason to expect that such a sale is likely to occur shortly after the Closing.

8.10 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in this Article 8, neither Holdings nor any other Person makes any other express or implied representation or warranty on behalf of Holdings or any of its Subsidiaries.

# Article 9 Realty

9.1 <u>Title Reports</u>.

9.1.1 Parent has ordered from First American Title Company ("Title Company") preliminary title reports on (i) all of the Timberlands (including the timberlands held pursuant to long term leases and those held in the form of perpetual cutting rights); (ii) all of the Realty (including realty owned by Transferred Subsidiaries (other than Boise Cascade do Brasil LTDA

41

and Compania Industrial Puerto Montt, S.A.); and (iii) the Facility Leases covering the Jackson, Alabama Facility (collectively, the "Titled Property"). Parent will provide such preliminary title reports to Holdings as they are received. Parent shall provide reasonable cooperation with Holdings and Title Company in the review and examination of the title to the Realty and the Timberlands. Such cooperation shall include providing access to Parent's Realty and Timberlands records and including available maps and surveys.

- 9.1.2 Parent and Holdings shall cooperate with the Title Company and use commercially reasonable efforts so that Holdings shall have obtained (a) no later than thirty (30) days prior to the Closing, a commitment from the Title Company for an ALTA Owner's Title Insurance Policy, 1992 Form (or other form of policy acceptable to Purchaser), together with a copy of all documents referenced therein (the "Title Commitments"), and (b) at Closing, title insurance policies from the Title Company (which may be in the form of a mark-up of a pro forma of the Title Commitments) in accordance with the Title Commitments, insuring Purchaser's fee simple title to each owned Titled Property or Holdings' legal, valid, binding and enforceable leasehold interest in each leased Titled Property (as the case may be) (in each case, excluding the Timberlands except to the extent required by Holdings' or its Affiliates' lenders) as of the Closing Date (including all recorded appurtenant easements necessary for access to Titled Property and insured as separate legal parcels) with gap coverage from Parent through the date of recording and extended coverage over the general exceptions, subject only to Permitted Encumbrances, in such amount as Holdings reasonably determines to be the value of the Titled Property insured thereunder (the "Title Policies"). Sellers shall execute any affidavits, indemnities and other agreements or assurances reasonably required by the Title Company to issue the Title Policies. The fees, costs and expenses with respect to the Title Commitments and Title Policies, including ALTA extended coverage, shall be borne equally by Parent and Holdings; provided that 100% of the premium or fee for deletion of the creditors rights exception will be borne by Holdings.
- 9.1.3 Holdings shall have obtained no later than ten (10) days prior to the Closing, a survey for each property set forth on Schedule 9.1.3 dated no earlier than the date of this Agreement, prepared by a licensed surveyor reasonably satisfactory to Holdings, and conforming to 1997 ALTA/ACSM Minimum Detail Requirements for Land Title Surveys, including Table A Items Nos. 1, 2, 3, 4, 6, 7(a), 7(b)(1), 7(c), 8, 9, 10, 11(b), 13, 14 15 and 16, or such lesser standards as the Title Company requires as a condition to the removal of any survey exceptions from the Title Policies in respect of such properties, and certified to Holdings, Holdings' lender and the Title Company, in a form satisfactory to each of such parties (the "Surveys"). The fees, costs and expenses with respect to the Surveys shall be borne equally by Parent and Holdings.
- 9.2 <u>Conveyancing.</u> At Closing, Sellers will convey the Realty to Holdings or its designated Permitted Affiliate Purchasers and the Timberlands to Timber Co. or its designee, in each case by special limited warranty deeds (the "Deeds") in a form customary and suitable for recordation in each applicable jurisdiction and sufficient to convey fee simple title to the Realty and Timberlands, subject only to Permitted Encumbrances, and otherwise in form and substance reasonably satisfactory to Holdings. For the avoidance of doubt, when used in this paragraph the term "special limited warranty deed" shall be interpreted to mean a customary form of deed in which the warranties of Sellers relate only to the period during which the Sellers had title, and in

42

which Sellers represent that they have no knowledge that such warranties are untrue for any period prior thereto.

- 9.3 Environmental Assessments. Parent has delivered or will deliver to Purchasers within 45 days after the date hereof environmental assessments (each an "Environmental Assessment") prepared in accordance with ASTM 1527-00 for each Facility (including Facilities owned by Transferred Subsidiaries), other than (i) Facilities and properties owned by Compania Industrial Puerto Montt, S.A.; (ii) Facilities leased by Boise Building Products Limited; and (iii) Facilities which are leased commercial office space. In addition, Parent will deliver within 60 days after the date hereof Environmental Assessments prepared in accordance with a modified version of ASTM 1528-00 covering all of the Timberlands and the eucalyptus plantations owned by Boise Cascade do Brasil LTDA. If any of such Environmental Assessments report (or disclose any environmental issue that Holdings reasonably determines could constitute) a Recognized Environmental Condition, as defined in the above-referenced ASTM standards (a "REC"), the following procedure shall be utilized by the parties to address the issue presented by such REC:
- 9.3.1 Parent and Holdings shall review each Environmental Assessment as it is made available by Parent. Within 10 days after receipt of each such Environmental Assessment (or within 10 days after the date hereof in the case of Environmental Assessments received prior to execution of this Agreement) either party may request that further investigation of the REC be conducted by a contractor designated by the requesting party or if it is requested by both parties by a contractor agreed upon by the parties, with the expense of the further investigation to be shared equally by the parties. In either case, the results of the additional investigation shall be made available to both parties and the scope of the investigation and technical details shall be consented to by both parties, such consent not to be unreasonably withheld.
- 9.3.2 Following the receipt of such additional investigation or if neither party requests any investigation Holdings may at its option either: (i) delete the affected Facility or Timberlands parcel or the affected portion of the Facility or Timberlands parcel from the transaction (and such deleted parcel or portion thereof shall be a Retained Asset and shall not be included in the definition of Facility or Timberlands); or (ii) take no action in which case the affected Facility or Timberlands parcel shall remain a portion of the Assets. If Holdings takes no action within ten days after expiration of the 10 day period provided for in Section 9.3.1 without a request by either party for additional investigation or within 10 days after receipt of any final report resulting from additional investigation, it shall be deemed to have elected option (ii) under the preceding sentence.
- 9.3.2.1 In the event that Holdings elects to delete the affected Facility or Timberlands parcel (or the affected portion thereof), the Timberlands Purchase Price or the Non-Timber Purchase Price, as appropriate, shall be equitably adjusted to reflect the value of the Asset so deleted; and
- 9.3.2.2 In the event that Holdings elects not to delete the affected Facility or Timberlands Parcel or affected portion thereof, the transaction shall be completed without adjustment of either the Non-Timber Purchase Price or the Timberlands Purchase Price and any liability incurred by Holdings

shall be a Retained Environmental Liability (<u>provided</u> that Holdings shall use commercially reasonable efforts to remedy as promptly as practicable after Closing such condition with the proceeds of indemnification therefor).

9.3.3 Holdings shall have no obligation to attend and carry out the actions required of it at the Closing, as set forth in Section 15.1.8, if the matters identified in the Environmental Assessments could reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

#### Article 10 Covenants of the Parties

- Access. Prior to the Closing, Sellers and the Transferred Subsidiaries will permit Holdings and the Permitted Affiliate Purchasers and their respective financing sources and representatives to have full access, during regular business hours and upon reasonable advance notice, to the Assets and the Business (including personnel and outside accountants and, with Parent's consent (not to be unreasonably withheld), other advisors), subject to reasonable rules and regulations of Parent concerning visitors to its Facilities, and shall furnish, or cause to be furnished, to Holdings any financial and operating data and other information that is available with respect to the Business, the Assets, or the Assumed Liabilities as Holdings shall from time to time reasonably request. Holdings shall abide by the terms of the Confidentiality Agreement with respect to such access and any information furnished to it or its representatives pursuant to this Section 10.1. Notwithstanding anything herein to the contrary, Holdings shall not be permitted to perform any invasive testing of any of the Realty without specific additional authorization from Parent (which consent shall not be unreasonably withheld), or to perform any testing which would cause a breach of any Lease to which Parent is a party, provided that, upon request from Holdings, Parent shall use reasonable efforts to obtain any necessary consent to permit such testing.
- Conduct of Business. During the period from the date hereof to the Closing, except as otherwise contemplated by this Agreement or as Holdings shall otherwise agree in writing in advance, Parent shall, and shall cause its Subsidiaries to, conduct the Business in the ordinary and usual course. Without limiting the foregoing, during such period Parent shall, and shall cause its Subsidiaries to, observe the negative and positive restrictions set forth in Sections 10.2.1 and 10.2.2 below.
- 10.2.1 During the period from the date hereof to the Closing, except as otherwise expressly provided for in this Agreement or as Holdings shall otherwise consent in writing (which consent shall not be unreasonably withheld), with respect to the Business, the Assets or the Assumed Liabilities, other than as set forth in Schedule 10.2 of Parent's Disclosure Schedule, Parent shall not:
- 10.2.1.1 enter into commitments for new capital expenditures in excess of \$25,000,000 in the aggregate to the extent not otherwise contemplated in the 2004 Capital Spending Authorizations;
- 10.2.1.2 except for pending transactions which are specifically Disclosed in Schedule 10.2.1.2 of Parent's Disclosure Memorandum, dispose of or otherwise

44

transfer, or incur, create or assume any Encumbrance (other than Permitted Encumbrances) on any asset of the Business, other than sales of Inventory or dispositions of worn out or obsolete assets, in each case in the ordinary course of business consistent with past practice. Notwithstanding the foregoing, Parent may sell the Retained HBU Lands (and any proceeds therefrom will be retained by Parent) and may sell not more than 500 additional acres of Timberlands in the aggregate transferred in the ordinary course. The Timberlands Purchase Price shall be reduced dollar-for-dollar by the amount of the proceeds (net of transaction costs) from all such ordinary course sales (other than sales of the Retained HBU Lands) pursuant to the immediately preceding sentence, and such proceeds shall not be included as a Current Asset in the determination of Closing Net Working Capital nor as cash that increases the purchase price pursuant to Section 3.2.4. Notwithstanding and in addition to the foregoing, Parent may pledge Accounts Receivable generated after it commences the wind-down of its receivables securitization program required by Section 7.6.5 to the extent necessary to fund the partial liquidation of such program, so long as such pledge is released at Closing;

10.2.1.3 institute any material change in the methods of purchase, sale, lease or other accounting or engage in any activity which would accelerate the collection of Accounts Receivable, accelerate or delay the payment of the accounts payable, or increase, reduce or otherwise affect the amount of Inventory (including raw material, packaging, work-in-process, or finished goods) on hand; or reduce or otherwise change any book reserves relating to workers' compensation, litigation, and other claims liabilities other than as set forth in the definitions of Workers Compensation Reserve and Litigation and Claims Reserve;

10.2.1.4 on or after the 45th day prior to the date set for Closing hereunder, sell any Accounts Receivable to Parent's receivables securitization facility;

10.2.1.5 acquire (by merger, exchange, consolidation, acquisition of stock or assets or otherwise) any Person or division or material assets thereof, whether or not engaged in a business similar to the Business; provided that nothing in this Section 10.2.1.5 shall preclude any acquisition by Boise Office Solutions of a Person, or the assets of a Person, engaged in the office products distribution business, so long as such transaction would have no effect on the Business, the Assets, the Transferred Subsidiaries, or the transactions contemplated by this Agreement;

10.2.1.6 sell, transfer, or grant licenses of Intellectual Property to any Person or allow registered Intellectual Property to lapse, expire, or become abandoned;

10.2.1.7 amend any Transferred Subsidiary's certificate of incorporation, bylaws, or other charter documents;

10.2.1.8 grant (or agree to grant) to any Employee any salary or wage increases except for regularly scheduled salary reviews and promotional increases consistent with past practices in the ordinary course of business or change or amend any Parent Plan affecting Employees, other than, in either case, pursuant to renegotiation of any collective bargaining agreements in the ordinary course or as required under law; provided that Parent may amend the Pension Plans and/or Defined Contribution Plans as it reasonably deems necessary or advisable solely for the purpose of preparing for and accomplishing the transfers contemplated

- under Sections 11.4 and 11.5; and provided further that nothing in the foregoing shall preclude Boise from entering into the 2004 Severance Agreements with its executive officers as set forth on Schedule D-19, or from adopting the 2004 Retention Bonus Plan for key employees as set forth on Schedule D-18;
- 10.2.1.9 enter into any transactions affecting the Business with Affiliates not in the ordinary course of business consistent with past practice other than the elimination of inter-company obligations involving Transferred Subsidiaries provided for in Section 10.4;
- 10.2.1.10 enter into, amend, or terminate any Contracts or Lease required to be disclosed hereunder, including the BOS Paper Supply Agreement (except as set forth in Section 18.2);
- 10.2.1.11 (a) with respect to the Assets, the Business or any Transferred Subsidiary, make or change any Tax election, change an annual Tax accounting period, adopt or change any Tax accounting method, or (b) with respect to any Transferred Subsidiary, file any amended Tax Return, enter into any Tax closing agreement, settle any Tax claim or assessment, surrender any right to claim a Tax refund, or consent to the extension or waiver of any limitation period applicable to any Tax, claim or assessment;
- 10.2.1.12 Sellers shall not with respect to the Transferred Subsidiaries, and Sellers shall cause the Transferred Subsidiaries not to, use any net operating loss carryforward, alternative minimum Tax net operating loss carryforward or any alternative minimum Tax credit carryforward, in each case to the extent permitted by applicable law; or
  - 10.2.1.13 agree, in writing or otherwise, to do any of the foregoing.
  - 10.2.2 During the period between the date hereof and Closing, Parent shall and shall cause its Subsidiaries to:
- 10.2.2.1 use commercially reasonable efforts to (i) preserve intact the organization and goodwill of the Business, (ii) keep available the services of its officers and employees as a group (provided that such efforts shall not require Parent to pay any bonuses or other amounts beyond normal compensation to such persons, except that Parent will adopt the 2004 Retention Bonus Plan for key employees as set forth on Schedule D-18), (iii) maintain satisfactory relationships with its material suppliers and customers and other Persons having business relationships with it, and (iv) maintain all Governmental Authorizations required for the conduct of the Business;
- 10.2.2.2 use reasonable commercial efforts to maintain its Facilities and Assets in accordance with its normal and customary maintenance practices and carry out its ongoing plan of capital expenditures in the ordinary course of business in an aggregate amount consistent with the 2004 Capital Spending Authorizations for the Business; provided that, so long as Parent and its Subsidiaries have exercised reasonable commercial efforts as required by the preceding phrase, the foregoing shall not constitute an absolute guaranty that all planned capital expenditures will be completed as scheduled in such authorizations;

46

- 10.2.2.3 promptly, but in any event within two (2) business days after becoming aware thereof, notify Holdings of any representation or warranty made hereunder ceasing to be true, or any breach of any covenant or agreement made hereunder;
- 10.2.2.4 notify Holdings of any emergency or other change in the normal course of the Business or in the condition of the Assets or the Assumed Liabilities or the operation of the Business and of any governmental or third party complaint, investigation or hearing (or communication indicating that such a complaint, investigation or hearing is or may be contemplated), if such emergency, change, complaint, investigation or hearing would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect;
- 10.2.2.5 prepare and timely file all applicable Tax Returns with respect to the Assets or the Business or any Transferred Subsidiary in a manner consistent with past practice, and pay all applicable Taxes when due and payable; and
- 10.2.2.6 operate the Timberlands in the ordinary course of business consistent with past practice, including (a) conducting their ongoing timber harvest operations on the Timberlands substantially in accordance with the species composition, timber products, volume levels, and timing detailed in Parent's Harvest Plan, and (b) performing necessary and prudent maintenance of roads, drainage systems, fire protection and prevention as well as appropriate stand establishment activities.
- 10.3 <u>Efforts to Achieve Closing</u>. Each of Parent and Holdings will use commercially reasonable efforts to fulfill the conditions precedent to the other's respective obligations hereunder and to secure all Consents required to be obtained by it in connection with the transactions contemplated hereby, and Parent and Holdings will cooperate in all reasonable requests to fulfill the conditions precedent to their and the other party's obligations described in this Section 10.3, in each case as soon as practicable but in any event not later than the Outside Date. Without limiting the generality of the foregoing,
- 10.3.1 Parent and Holdings will file within 10 Business Days of the date hereof the required notice (including all documentary materials) under the HSR Act and will as soon as practicable make required filings under any similar regulatory programs applicable to the transaction in Canada, Brazil and the United Kingdom. Each party shall promptly file any additional information requested in respect of such filings by any Governmental Authority as soon as practicable after receipt of request therefor. The filing fees for such filings shall be shared equally by Parent and Holdings.
- 10.3.2 Holdings shall, and shall cause each of its respective Subsidiaries to, use reasonable efforts to satisfy all conditions precedent to be satisfied by Holdings or its Affiliates in the Commitment Letters and to otherwise obtain, on behalf of Holdings and its Subsidiaries, the proceeds of the debt financing and equity financing contemplated by the Commitment Letters; provided that nothing in this Section 10.3.2 or elsewhere in this Agreement shall require (i) the payment by Holdings or any of its Subsidiaries of any material amount of cash other than as required by the Debt Commitment Letters or any fee letters executed in connection with the Debt Commitment Letters, (ii) the agreement by Holdings or any of its Subsidiaries to any material economic terms other than those contemplated by either the Debt Commitment Letters

47

or the Equity Commitment Letter or to other terms that are not commercially reasonable, or (iii) Holdings or any of its Subsidiaries to seek to obtain any alternative financing in the event that any portion of the financing contemplated by the Debt Commitment Letters or the Equity Commitment Letter is unavailable.

10.3.3 Parent shall use its reasonable efforts to cooperate with Holdings in connection with the financing contemplated by the Commitment Letters, including using (and causing its Subsidiaries to use) reasonable efforts to satisfy all conditions precedent to be satisfied by Parent and its Subsidiaries in the Commitment Letters, providing information to and permitting the financing sources and their representatives access to the Assets and the Business, as provided in Section 10.1 hereof, participating in meetings with prospective investors and participating (and permitting members of its senior management to participate) in "road

shows" in connection with the financing, participating in meetings with rating agencies, participating in drafting sessions related to the offering materials for the debt financing of Holdings and its Subsidiaries contemplated by the Debt Commitment Letters, causing the present and former independent accountants for Parent to participate in drafting sessions related to the offering materials for the debt financing contemplated by the Debt Commitment Letters and making work papers available to Holdings and its Subsidiaries, the underwriters for the debt financing and their respective representatives; provided that Holdings shall endeavor to minimize the interference with the ability of such senior management members to carry out their normal responsibilities for ongoing management of the Business resulting from such participation in drafting sessions and the road show. Out of pocket costs incurred by Parent for participation by its external accountants, outside counsel, and other outside advisors and consultants in the foregoing shall be paid by Holdings; provided that the fees of KPMG for preparation of the Regulation S-X Financial Statements shall be shared equally by Parent and Holdings.

- 10.4 <u>Transferred Subsidiaries</u>. Prior to Closing, Parent shall and shall cause its Subsidiaries to:
- 10.4.1 cause all inter-company indebtedness owed by any Transferred Subsidiary to Parent or a Retained Subsidiary or by Parent or a Retained Subsidiary to a Transferred Subsidiary, to be eliminated so that at Closing there will be no inter-company indebtedness between Parent (or its Retained Subsidiaries) and any Transferred Subsidiaries, except for Trade Intercompany Accounts; and
- 10.4.2 cause each Transferred Subsidiary which holds any assets which are not related to the Business to transfer such assets to Parent or a Retained Subsidiary; and cause each Transferred Subsidiary which has any obligations constituting Retained Liabilities to assign such obligations to Parent, and Parent will assume such Retained Liabilities.
- 10.4.3 Parent shall indemnify Holdings and each Transferred Subsidiary for any loss, liability or Tax incurred by Holdings or such Transferred Subsidiary in connection with the actions referred to in Section 10.4.1 and Section 10.4.2.
- 10.5 <u>Further Assurances</u>. At any time after the Closing Date, Holdings, and Sellers shall promptly execute, acknowledge, and deliver any other assurances or documents reasonably requested by Holdings or Seller, as the case may be, and necessary for them or it to satisfy their

48

or its respective obligations hereunder or obtain the benefits contemplated hereby and to confirm to third parties the transfer of Assets and the Assumed Liabilities provided for herein. Without limiting the generality of the foregoing, Holdings agrees that if any of the Transferred Subsidiaries are found to own assets that are not related to the Business, or if any Retained Assets are inadvertently transferred to Holdings, Holdings, or its applicable Permitted Affiliate Purchaser, shall transfer or cause the Transferred Subsidiary to transfer such assets to Parent, or as Parent shall direct, at Parent's expense (including any Tax incurred by Holdings or its Subsidiaries) but without consideration. Similarly, if, after the Closing, Parent or Holdings identifies any assets that should have been transferred to Holdings as part of the Assets, but were not, Parent shall transfer such assets to Holdings, or a designated Permitted Affiliate Purchaser at Parent's expense and without further consideration. A similar process shall be followed with Assumed Liabilities and Retained Liabilities which are not properly transferred or retained.

- 10.6 <u>Use of Boise Marks</u>. As soon as commercially practicable after Closing, Parent and its Retained Subsidiaries shall discontinue the use of the Boise Marks. Notwithstanding the foregoing, the following specific limitations shall be observed by Parent:
- 10.6.1 Parent and its Retained Subsidiary, Boise Cascade Office Products Corporation, shall change their names to a variant of OfficeMax or other name not including the words "Boise" nor "Cascade" within five business days after Closing and the legal names of Parent's remaining Retained Subsidiaries shall be changed to eliminate the words "Boise" and "Cascade" within three months after Closing.
  - 10.6.2 Within 60 days after Closing, Parent shall change its New York Stock Exchange Trading Symbol from BCC to OMX.
- 10.6.3 Existing stocks of printed stationary, including letterhead, invoices, acknowledgements and checkstock (for bank accounts not included in the Assets) may continue to be used for up to three months. Where possible without unreasonable effort or expense, such stationary shall be stamped or overprinted before use to disclose the correct identity of the user.
  - 10.6.4 Signage on facilities and rolling stock shall be changed over within three years and five years, respectively; and
  - 10.6.5 Parent's office products catalogs and other marketing materials shall not utilize the Boise Marks in the 2006 catalog cycle or thereafter.

Nothing in the foregoing shall preclude Parent from using appropriate Boise Marks associated with paper products supplied to it by Holdings after Closing in the remarketing of any such products through its office products distribution business so long as such use is limited to, and clearly identified with, such Holdings products and is otherwise consistent with any terms that Holdings or any of its Subsidiaries imposes on the use of the Boise Marks with its customers. In no event shall Parent or any of the Retained Subsidiaries use the Boise Marks in any manner not used by them in the 12 month period prior to the date hereof or in connection with goods or services of a different character or quality than those offered by them prior to the date hereof.

49

Each party shall use commercially reasonable efforts to clarify any confusion concerning their respective identities which arises after Closing. Parent shall indemnify Holdings against third party claims asserted against Holdings arising from any use by Parent or its Retained Subsidiaries of the Boise Marks after the Closing.

# 10.7 <u>Certain Matters Related To Retained and Assumed Liabilities.</u>

- 10.7.1 With respect to all Retained Liabilities, Holdings Indemnified Parties shall reasonably cooperate with Parent, provide Parent as promptly as possible with notices and other information received by such parties as well as all relevant materials, information and data requested by Parent and shall grant Parent, without charge, such reasonable access to employees of the Business and to the Assets as may be reasonably necessary for Parent to defend against, settle, and discharge the Retained Liabilities.
- 10.7.2 With respect to all Assumed Liabilities, Parent Indemnified Parties shall reasonably cooperate with Holdings, provide Holdings as promptly as possible with notices and other information received by such parties as well as all relevant materials, information and data requested by Holdings and shall grant Holdings, without charge, such reasonable access to employees of Parent as may be reasonably necessary for Holdings to defend against, settle, and discharge the Assumed Liabilities.

# 10.8 <u>Inter-company Agreements</u>.

- 10.8.1 Parent (on the one hand) and Boise Office Solutions (on the other hand) shall, during the period between the date hereof and Closing, perform in the ordinary course and refrain from amending the BOS Paper Sales Agreement (except as set forth in Section 18.2) and shall not grant to either party any material waiver of the provisions of such contract which will have any effect after Closing. Parent shall take such actions as may be necessary to cause Boise Office Solutions to comply with the foregoing covenant.
- 10.8.2 All other inter-company agreements between Parent and any of its Affiliates which relate to the conduct of the Business or between any Transferred Subsidiary and Parent or a Retained Subsidiary shall be terminated prior to Closing and all liabilities and obligations arising thereunder and any other intercompany accounts, indebtedness, liabilities, or obligations shall be fully discharged prior to Closing, except for any Trade Intercompany Accounts.
- 10.8.3 Parent shall indemnify Holdings and each Transferred Subsidiary for any loss, liability or Tax incurred by Holdings or such Transferred Subsidiary in connection with the actions referred to in Section 10.8.1 and Section 10.8.2.
- 10.9 <u>Records and Retention and Access</u>. Holdings shall keep and preserve in an organized and retrievable manner the Retained Records in its possession at Closing and the Transferred Records for the lesser of (i) seven years from the Closing Date or (ii) the retention period for such records under the longer of Holdings' or Parent's records retention schedules applicable to such records. Upon expiration of such period for any specific Transferred Record or Retained Record, Holdings shall notify Parent that it intends to destroy such records unless

50

Parent takes possession of them within 60 days following such notice and upon expiration of such period may promptly destroy any such record not taken by Parent. Notwithstanding the foregoing, Parent may take possession of the Retained Records at any time on 30 days notice to Holdings. While such Retained Records and Transferred Records remain in existence, each party shall allow the other party, its representatives, attorneys and accountants, at the requesting party's expense, access to the Books and Records upon reasonable request and advance notice and during normal business hours, subject to the confidentiality provisions set forth in Section 19.18. Holdings' obligations with respect to such records shall include maintaining, for at least the retention period specified above, computer systems permitting access to any of such records which are stored in electronic form in a fashion which is not less efficient than current access methods.

#### 10.10 Insurance.

10.10.1 Sellers shall use commercially reasonable efforts to arrange for assignment of each policy which under the terms of Schedule D-14 is indicated to be an "assigned" policy. For those policies which are required by Schedule D-14 to be divided, Sellers shall, in consultation with Holdings, use commercially reasonable efforts to either arrange a carve out of an appropriate portion of the coverage for the benefit of Holdings (or Sellers if the policy is to be transferred to Holdings) or arrange for the cancellation and reissue of the divided policy as two policies of like coverage, limits, deductible, and other material terms for the respective benefit of Sellers and Holdings. The assigned policies and the portions of the partially assigned policies to be assigned and any replacement policies issued to Holdings as a result of a policy which is divided shall be the "Transferred Insurance Policies". Parent and its Subsidiaries shall be additional insureds on the Transferred Insurance Policies to the extent indicated in Schedule D-14; provided that notwithstanding anything herein to the contrary, Holdings and its Subsidiaries shall not have any obligations to process claims on behalf of Parent and its Subsidiaries except to the extent provided under the Mutual Administrative Services Agreement.

10.10.2 Sellers covenant and agree that (i) each Seller shall, and shall cause each of its Affiliates to, continue to maintain, without amendment or modification, the Transferred Insurance Policies (or if such policies are terminated, substantially equivalent policies) for the benefit of the Business, with respect to occurrences prior to the Closing Date, regardless of whether claims are brought prior to or after the Closing Date; (ii) prior to Closing, Sellers shall give notice to the insurers under the Transferred Insurance Policies and other policies of each claim then pending or threatened against Parent or any of its Subsidiaries with respect to any of the Assets or the Business which they reasonably expect to exceed 50% of the applicable deductible or other loss retention feature of the applicable Transferred Insurance Policies; (iii) Holdings and its Subsidiaries will, with respect to the Assets and the Business, be provided full coverage under the Transferred Insurance Policies after the Closing (subject only to deductibles, self-insured retentions and policy limits set forth on Schedule D-14), with respect to occurrences prior to and/or after the Closing, regardless of whether claims are brought prior to or after the Closing; (iv) after Closing, Holdings and its Subsidiaries will have the right to bring claims (whenever brought) as a direct insured party under the Transferred Insurance Policies for all occurrences that relate to actions, events, conditions and/or operations of the Assets and the Business prior to and/or after the Closing Date within the scope of the Transferred Insurance Policies; and (v) Sellers shall cause Holdings and its Subsidiaries to be "additional insureds"

51

under each insurance policy that affected any portion of the Assets or the Business which is not a Transferred Insurance Policy and shall ensure that Holdings and its Subsidiaries will, in such capacity, have the same coverage under such insurance after the Closing as existed for the Business and the Assets prior to Closing. Holdings and its Subsidiaries will be responsible for satisfying any deductibles, self-insured portions, retentions and other retained amounts on insurance coverage with respect to losses arising under claims made under the Transferred Policies to the extent relating to Assumed Liabilities and Sellers shall be responsible for any such deductibles, self-insured portions, retentions and other retained amounts on insurance coverage with respect to any such losses arising under the Transferred Policies to the extent relating to Retained Liabilities and under all policies that are Retained Assets.

10.11 Exclusivity. Neither Parent nor any of its Affiliates will, nor will they permit or cause their respective officers, directors, agents or employees to, solicit, initiate, encourage, discuss, negotiate or enter into any arrangements relating to, any possible sale or other disposition of all or any material part of the Assets or of the Business other than the sale of Inventory in the ordinary course of business or other sales or dispositions permitted by this Agreement (whether by merger, reorganization, recapitalization or otherwise) with any party other than Holdings and its Affiliates (a "Company Takeover Proposal"), or provide any information to any other party regarding the Business other than information which is traditionally provided in the regular course of its business operations to third parties where Parent and its officers, directors, employees, and agents have no reason to believe that such information may be utilized to evaluate a Company Takeover Proposal. Parent and its officers, directors, employees, agents and Affiliates (i) do not have any agreement, arrangement, or understanding with any third party with respect to any Company Takeover Proposal (other than confidentiality agreements), (ii) will cease and cause to be terminated any and all discussions with third parties regarding any Company Takeover Proposal, and (iii) will promptly notify Holdings if any Company Takeover Proposal, or any inquiry or contact with or by any Person or entity with respect thereto, is made (including the identity of the proposing Person and the terms of such proposal).

If within 30 days after the date hereof Parent receives, without any solicitation thereof which is made after the date hereof, a Company Takeover Proposal that constitutes a Company Superior Proposal (as defined below), then in such event Parent shall notify Holdings that it has received a Company Superior Proposal that Parent's Board of Directors is prepared to accept and shall provide Holdings with a description of all material terms of the Company Takeover Proposal and the identity of the party making such offer. Holdings may, at any time during the five business days after delivery of such notice, at its option, submit a counter-

proposal providing for the delivery of aggregate consideration for the Business (valuing Timber Notes for such purpose at their face value) that is at least equal to the aggregate consideration to be paid for the Business under the Company Superior Proposal (as reduced to take into account the effect of the break-up fee that would be payable under Section 14.3.1 hereof) (a "Matching Offer"). If Parent delivers such a notice of a Company Superior Proposal and after five business days from delivery of such notice Holdings has not submitted a Matching Offer to Parent, then either Parent or Holdings may terminate this Agreement pursuant to Section 14.2.5 hereof with the effect specified therein and in Section 14.3.1 hereof. A "Company Superior Proposal" means a Company Takeover Proposal that provides for cash consideration to be paid at closing for the Business that is at least \$250,000,000 higher than the aggregate consideration (without reduction

52

for the amount of the Reinvestment or the break-up fee payable under Section 14.3.1 hereof, and valuing Timber Notes for such purpose at their face value) to be paid for the Business under this Agreement, and contains terms and conditions that Parent's Board of Directors determines in good faith, after consultation with counsel and a financial advisor of nationally recognized reputation and taking into account all relevant material terms of such Company Takeover Proposal and this Agreement, are more favorable to the shareholders of Parent than the transactions provided for herein, and as to which Parent's Board of Directors reasonably determines that rejection of such Company Takeover Proposal would constitute a violation by the Board of Directors of its fiduciary duties as directors of Parent under applicable laws.

#### 10.12 <u>Noncompetition</u> From and after the Closing:

10.12.1 Noncompetition. In consideration of the mutual covenants provided for herein, during the period beginning on the Closing Date and ending on the fifth anniversary of the Closing Date (the "Noncompete Period"), Parent shall not, and shall cause its Affiliates not to, directly or indirectly (i) engage in any business that competes with all or any portion of the Business (as conducted as of Closing), anywhere within the United States, or (ii) induce or attempt to induce any customer or other business relation of Holdings or any of its Subsidiaries to terminate or reduce such relationship with Holdings or any of its Subsidiaries; provided that Parent and its Affiliates shall not be deemed to be competing in violation of this Section 10.12 by virtue of its or their (w) equity interest relating to the Reinvestment, (x) ownership of less than 5% of the outstanding stock of any publicly-traded corporation, (y) acquisition of any Person (whether by asset purchase, stock purchase, merger or otherwise) engaged in a business that competes with the Business if such businesses are not such target Person's primary business and, on a pro forma basis after giving effect to such acquisition, the aggregate sales attributable to the business(es) competitive with the Business represent less than 3% of aggregate consolidated pro forma revenues of Parent, its Subsidiaries and such Person, or (z) ordinary course business operation of Boise Office Solutions substantially in the manner operated as of the date of this Agreement. This Section 10.12 shall not apply to any entity which might acquire Parent or any assets or division of Parent, subject to the terms of the Shareholder Agreements.

10.12.2 Severability. The parties hereto agree that the covenant set forth in Section 10.12.1 is reasonable with respect to its duration, geographical area and scope. If the final judgment of a court of competent jurisdiction declares that any term or provision of Section 10.12.1 is invalid or unenforceable, the parties agree that the court making the determination of invalidity or unenforceability shall have the power to reduce the scope, duration or area of the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified after the expiration of the time within which the judgment may be appealed.

#### 10.13 <u>Delivery of Additional Financial Statements</u>.

10.13.1 <u>Subsequent Statements</u>. As soon as practicable, but in any event no later than 30 days after the end of each calendar month following June 30, 2004, Parent will

53

deliver to Holdings a copy of Parent's internal operating balance sheet, profit and loss statement, and cash flow statement for each the Operating Units and for Parent's aviation division, transportation operations (including the operations of Parent's Subsidiaries, Minnesota, Dakota and Western Railroad Company and BCT, Inc.) and its cottonwood farm operations, at and for the year to date periods ending on the last day of each such month, together with a balance sheet and a profit and loss statement for Parent's corporate staff departments for each of the periods provided for above (collectively the "Subsequent Financial Statements"). At the time of delivery, the Subsequent Financial Statements shall be attached as part of Schedule 7.6.1.

10.13.2 Audited Year-End Statements. At least 30 days prior to the Closing, Parent shall furnish or shall cause Parent's independent accountants to furnish the following financial statements for the Business, each in a form meeting the requirements of Regulation S-X of the Securities Act of 1933, as amended ("Regulation S-X"): (i) the audited consolidated balance sheet of the Business (after elimination, to the extent permitted by Regulation S-X, of any Retained Assets or any Retained Liabilities and any operations not included in the Business) as at December 31, 2003 and December 31, 2002, together with the related audited consolidated statement of income and consolidated balance sheet of the Timberlands Business (after elimination of any Retained Assets or Retained Liabilities and any operations not included in the Business) as at December 31, 2003 and December 31, 2002, together with the related audited consolidated statement of income and consolidated statement of cash flows for the twelve-month periods ended December 31, 2003, December 31, 2002, and December 31, 2001, and (iii) the audited consolidated balance sheet of the Non-Timberlands Business (after elimination of any Retained Assets or any Retained Liabilities and any operations not included in the Business) as at December 31, 2003 and December 31, 2002, together with the related audited consolidated statement of income and consolidated statement of cash flows for the twelve-month periods ended December 31, 2002, together with the related audited consolidated statement of income and consolidated statement of cash flows for the twelve-month periods ended December 31, 2003, December 31, 2002, and December 31, 2001 (the financial statements referred to in clauses (i), (ii) and (iii) foregoing, collectively referred to herein as the "Audited Financial Statements"). At the time of such delivery, the Audited Financial Statements shall be attached as part of Schedule 7.6.1.

10.13.3 <u>Unaudited Stub Period Statements</u>. Furthermore, as soon as practicable, but in any event not less than 30 days prior to Closing, Parent shall furnish the following financial statements for the Business, each in a form meeting the requirements of Regulation S-X: (i) the unaudited combined balance sheet of the Business (after elimination, to the extent permitted by Regulation S-X, of any Retained Assets or any Retained Liabilities and any operations not included in the Business) as at June 30, 2004 (and as at June 30, 2003), together with the related unaudited consolidated statement of income and consolidated statement of cash flows for the period from January 1, 2004 through June 30, 2004 (and for the period from January 1, 2003 through June 30, 2003), (ii) the unaudited balance sheet of the Timberlands Business (after elimination of any Retained Assets or Retained Liabilities and any operations not included in the Business) as at June 30, 2004 (and as at June 30, 2003), together with the related unaudited statement of income and statement of cash flows for the period from January 1, 2004 through June 30, 2004 (and for the period from January 1, 2003 through June 30, 2003), and (iii) the unaudited consolidated balance sheet of the Non-Timberland Business (after elimination of any Retained Liabilities and any operations not included in the Business)

as at June 30, 2004 (and as at June 30, 2003), together with the related unaudited consolidated statement of income and consolidated statement of cash flows for the period from January 1, 2004 through June 30, 2004 (and for the period from January 1, 2003 through June 30, 2003) (the financial statements referred to in clauses (i), (ii) and (iii) foregoing, collectively referred to herein as the "Stub Period Financial Statements"). At the time of such delivery, the Stub Period Financial Statements shall be attached as part of Schedule 7.6.1.

- 10.13.4 <u>Additional Financial Statements</u>. In addition, promptly after a request therefor, Parent shall provide to Holdings and its Subsidiaries, each other financial statement and all other financial data required by Regulation S-X in connection with the debt financing contemplated by the Debt Commitment Letters, including any offering memorandum, prospectus, or registration statement delivered or filed in connection therewith (collectively, the "Additional Financial Statements"). At the time of such delivery, the Additional Financial Statements shall be attached as part of Schedule 7.6.1.
- 10.13.5 Additional Covenants. The financial statements and financial data in this Section 10.13 (other than the Subsequent Financial Statements) are collectively known as the "Regulation S-X Financial Statements." Parent will prepare, or cause the Regulation S-X Financial Statements to be prepared, in accordance with GAAP and in a form meeting the requirements of Regulation S-X for Holdings' or its Subsidiaries' offering memorandum, prospectus, registration statement and any amendments thereto in connection with the debt financing contemplated by the Debt Commitment Letters and any updated filings. Parent shall furnish to Holdings and its Subsidiaries any information or documents requested by Holdings or any of its Subsidiaries, which is in Parent's possession or to which Parent has access, constituting, or necessary or desirable for the completion of, the Regulation S-X Financial Statements or any updated filings or amendments thereto, and Parent agrees to execute any reasonably necessary management representation letters to permit Holdings or its Subsidiaries' independent accountants to issue unqualified reports with respect to the Regulation S-X Financial Statements to be included in the offering memorandum, prospectus, registration statement and any updated filings or amendments thereto.
- 10.13.6 <u>Consent of Parent Auditors</u>. Parent shall use its reasonable best efforts to obtain prior to the Closing the written agreement (the "Auditor Consent Letter") of KPMG LLP ("KPMG") to permit the use of the Regulation S-X Financial Statements (a) in connection with offerings of securities by Holdings and/or one or more of its Subsidiaries as contemplated by the Commitment Letters, and (b) subject to KPMG's normal procedures, in other private or public offerings of securities as may be reasonably requested by Holdings and/or one or more of its Subsidiaries. In addition, Parent will use commercially reasonable efforts to cause KPMG to provide a comfort letter in accordance with SAS 72 for any such offering.
- 10.14 <u>Restrictions on Sale of White Paper Business</u>. Holdings agrees that it will not, and will cause its Subsidiaries not to, transfer all or substantially all of the assets constituting the white paper manufacturing, conversion, distribution and sales operations of the Business (including by transfer of a majority of the equity interest in any entity containing such operations) to any Person at any time during the 6-month period commencing on the Closing Date.

55

- 10.15 <u>Assignment of Timberlands Purchase Rights</u>. Timber Co. has advised Parent and Southern that it intends to assign its rights and obligations in respect of its purchase of the Timberlands to Timber LLC prior to Closing and that Timber LLC will on the terms and conditions set forth below direct Parent and Southern to convey the Timberlands to Timber Co. Sellers hereby consent to such transfer so long as it conforms to the following terms and conditions (and, in such case, Parent and Southern will absolutely, unconditionally, and irrevocably release Timber Co. from the Purchaser Payment Performance Obligations):
- 10.15.1 Such assignment shall be in writing and a copy thereof shall have been provided to Parent prior to Closing. It shall provide that Timber LLC shall assume and agree to the due and punctual performance of the terms and conditions, undertakings, covenants and obligations of Timber Co. hereunder in respect of the purchase of the Timberlands;
- 10.15.2 Subsequent to the assignment provided for in Section 10.15.1 and at or before Closing: (i) Timber LLC will direct Parent and Southern to convey the Timberlands to Timber Co. upon Closing; and (ii) Timber Co. will absolutely, unconditionally, and irrevocably assume the due and punctual performance of the terms, conditions, undertakings, covenants, and obligations assumed and to be performed or observed by Timber LLC under this Agreement and under applicable law in respect of the sale of the Timberlands (and will assume all rights and benefits relating to the Timberlands under this Agreement and any other agreements and under applicable laws), excepting only Timber LLC's obligations under the Timber Notes and the other documents provided for in the LC Commitment Letters, or any other liability with respect to the payment of the Timberlands Purchase Price (such excepted obligations of performance by Timber LLC hereinafter referred to as "Purchaser Payment Performance Obligations"). At Closing, Parent and Southern will convey the Timberlands to Timber Co. in accordance with Timber LLC's direction, and Parent and Southern will absolutely, unconditionally, and irrevocably release Timber LLC from all of the terms, conditions, understandings, comments and obligations under this Agreement and under applicable law other than the Purchaser Payment Performance Obligations.
- 10.15.3 Timber Co. and Timber LLC shall covenant in the assignment document and for the express benefit of Sellers that during the term of the Timber Notes:
- 10.15.3.1 without the prior written consent of Parent (which consent, in the case of amendments that would not have an adverse effect on Sellers will not be unreasonably withheld, and otherwise may be withheld by Seller in its sole discretion), neither Timber Co. nor the manager of the Timber LLC shall make any modification or amendment to, or grant any waiver under, Timber LLC's operating agreement or cause or permit any interest in Timber LLC to be issued to any person other than Timber Co.;
- 10.15.3.2 without the prior written consent of Parent (which consent shall not be unreasonably withheld, conditioned, or delayed), Timber Co. shall not sell, assign or transfer all or any portion of its interest in Timber LLC to any Person nor shall the manager of the Timber LLC resign;
- 10.15.3.3 without the prior written consent of Parent (which consent may be withheld by Parent in its sole discretion), neither Timber Co. nor the manager of Timber LLC

56

shall make any election that would cause Timber LLC to be treated as an association taxable as a corporation for federal income tax purposes; and

10.15.3.4 without the prior written consent of Parent (which consent may be withheld by Parent in its sole discretion) neither Timber Co. nor the manager of the Timber LLC shall cause or permit Timber LLC to engage in any activities prohibited by the terms of Timber LLC's operating agreement, or cause or permit Timber LLC to engage in any activities inconsistent with the purposes of Timber LLC as provided in Timber LLC's operating agreement;

provided, however, that the foregoing shall only be applicable to Timber Co. prior to the transfer of Timber Co.'s interest in Timber LLC and to the manager of the Timber LLC prior to resignation of the manager of Timber LLC, in each case in compliance with the provisions of this Subsection 10.15.3. Timber Co.'s right to transfer its membership interest in Timber LLC and the manager's right to resign (if the member to which Timber Co. transfers its interest determines to replace the manager) shall be subject to the additional condition that Timber Co. has obtained agreement of its transferee and of the replacement manager (if there shall be one) (in each case, expressly stated to be for the benefit of and enforceable by Parent) to be bound by the provisions of the assignment of Timber Co.'s Interest as if such transferee were Timber Co. and the replacement manager were the manager named therein (including the specific restrictions contained in Sections 10.15.3.1 through 10.15.3.4 above).

10.16 <u>Cooperation Concerning Timber Notes</u>. Holdings and Timber Co. acknowledge that Parent and Southern desire to obtain installment sale treatment for federal and state income Tax purposes on the sale of the Timberlands (to the extent of the Timber Notes) and obtain the credit enhancement of the Timber Notes provided for in the arrangements called for in the LC Commitment Letters. The parties agree to work cooperatively to complete the arrangements described in the LC Commitment Letters on a basis which is reasonably acceptable to both parties in an effort to satisfy the objectives of Parent and Southern described in the preceding sentence.

# Article 11 Employees

- 11.1 <u>Termination of Employees</u>. As of 11:59 p.m., Mountain standard time, on the day immediately preceding Closing, Parent shall terminate its employment of all of the Employees except for Employees employed by the Transferred Subsidiaries. With respect to each Employee terminated by Parent, Parent shall be responsible for payment when due of all salaries, wages, commissions and bonuses and other incentive compensation (including without limitation, a pro-rata payment under each incentive plan (determined in good faith) with respect to periods prior to the Closing Date) and any amounts due or to become due under the 2004 Retention Bonus Plan. Parent shall reimburse the Employees for all travel expenses and other business-related expenses arising prior to Closing. Parent will indemnify Holdings for any severance and retention obligations to current or former Employees of the Business as a result of any termination prior to the Closing
- 11.2 <u>Obligation to Hire Employees</u>. Effective 12:00 a.m., Mountain standard time, on the day of Closing, Holdings shall offer employment to each of the Employees, including

57

Employees then on leave but excluding any Employees receiving long-term disability benefits, and Employees on leave shall be deemed to have accepted employment as of the Closing Date unless such Employee declines such offer. Such offer shall be at a wage rate and on other terms and conditions which (i) in respect of Employees whose employment is governed by a collective bargaining agreement, is in accordance with the requirements of such agreement, and (ii) in respect of all other Employees, provides for base salaries, base wages and commission, and cash bonus opportunity, which in the aggregate are substantially comparable to those in effect for each such Employee under its employment with Parent immediately prior to the Closing Date. Holdings will provide incentive compensation for management via equity-based awards and/or equity investment opportunities. Holdings shall not be obligated to pay severance to any Employee who rejects Holding's offer of employment. All Employees who accept employment with Holdings on the Closing Date shall commence such employment as of immediately after their termination by Parent so that there is no gap in their employment or in their health and welfare benefit plan coverage. All Employees who accept employment with Holdings pursuant to this Section 11.2 shall be referred to herein as "Transferred Employees." The parties acknowledge that Holdings may cause one or more of its Subsidiaries to satisfy its obligations under this Article 11, including the obligation to offer employment to and hire the Transferred Employees.

Notwithstanding anything to the contrary herein, in connection with any offer of employment to an Employee who is party to a 2004 Severance Agreement, Holdings may include as part of such offer that such Employee will enter into a severance agreement or arrangement with terms and conditions (other than with respect to the aggregate amount of severance benefits) different than the 2004 Severance Agreement, so long as Holdings offers such Employee an alternative offer (which may contain lesser compensation and benefits) that includes the 2004 Severance Agreement. With respect to any Transferred Employee who is party to a 2004 Severance Agreement, Parent agrees at Holdings' request to take all such actions and enter into such agreements as may be necessary for Parent to assign to Holdings all of the rights and benefits applicable to Parent (including covenants from such Transferred Employee) under such 2004 Severance Agreement.

#### 11.3 Health and Welfare Benefit Plans.

- 11.3.1 Parent shall remain solely responsible in accordance with its employee health and welfare benefit plans, programs, and policies for the satisfaction of all claims regardless of when filed (subject to claim filing requirements of such plans) for (i) medical, dental and vision benefits claims which relate to services provided to Employees (and their covered dependents) prior to Closing, and (ii) life insurance, accident or disability benefits (other than workers' compensation benefits) claims which relate to events or injuries incurred prior to Closing.
- 11.3.2 Except as otherwise provided under the Mutual Administrative Services Agreement, Holdings shall establish, effective as of the Closing Date, health and welfare benefit plans for the Transferred Employees which (i) in respect of unionized Transferred Employees, meet the requirements of applicable collective bargaining agreements, and (ii) in respect of all other Transferred Employees, provide health and welfare benefits which in the aggregate are substantially comparable (taking into account premium participation, co-payment, deductibles,

58

amount of coverage and other material terms) to the benefits currently provided by Parent which are listed on Schedule 7.9.2(a) of Parent's Disclosure Memorandum. Transferred Employees shall be enrolled in such plans as of the beginning of their employment by Holdings without waiting period or other initial limitation of coverage (except as applied under Parent Plans). Holdings shall not terminate or make materially adverse reductions in coverage or Employee increases in cost of participation for a period of two years after Closing. Holdings' new health and welfare benefits shall include transition provisions coordinating the transfer of Transferred Employees from Parent's benefits to Holdings' benefits without duplication of annual deductibles and other yearly calculations. Employees (and covered dependents) who are hospitalized as of the Closing will remain on Parent's medical coverage until discharged from the hospital, at which time they will be eligible to enroll under Holdings' medical plan. Employees who are on long-term disability as of the Closing, and Employees who first become eligible for long-term disability benefits after the Closing because they remain disabled from the Closing Date through the end of an applicable elimination or waiting period, will remain on Parent's long-term disability coverages unless they return to active employment with Holdings. Parent shall transfer to Holdings its contract with the service provider for its flexible spending accounts for Transferred Employees (or provide a replacement contract on the same terms limited in scope to the Transferred Employees) and assets equal to Transferred Employees' account balances as of the date of transfer, and Holdings shall maintain substantially comparable flexible spending plans for at least two years after the date of Closing. Holdings' flexible spending account plan shall reimburse Transferred Employees for eligible medical and dependent care expenses incurred by such Transferred Employees (and their covered dependents) in 2004

11.3.3 Parent will provide eligibility for coverage and will be solely responsible for complying with all employer responsibilities under provisions of the Consolidated Omnibus Budget Reconciliation Act ("COBRA") to Employees (and their covered dependents) in respect of its termination of their employment with Parent. Holdings will be solely responsible for complying with all employer responsibilities under COBRA on and after the Closing Date with respect to all Transferred Employees (and their covered dependents).

# 11.4 Pension Plans.

- 11.4.1 On or after July 31, 2004, but no later than the Closing Date, Parent shall spin off (within the meaning of Section 1.414(l)-1(n) of the Treasury Regulations) from each of the Pension Plans that portion of each plan attributable to Employees who shall be identified as of the date of the spinoff (collectively, the "Spunoff Pension Plans"). Each of the Spunoff Pension Plans will be a mirror plan of each respective Pension Plan in all material respects and Parent will not amend the Spunoff Pension Plans except as required under applicable law or as necessary to accomplish the spinoff and transfer under this Section 11.4. Assets assigned to the Spunoff Pension Plans shall be in the form of cash or marketable securities that are acceptable to Holdings, provided that such acceptance shall not be unreasonably withheld.
- 11.4.2 Parent agrees to transfer to Holding on the Closing Date Spunoff Pension Plans that are each fully funded on an Accumulated Benefit Obligation basis as described below.

59

11.4.2.1 On the Closing Date, Parent shall make a contribution to each Spunoff Pension Plan. The amount of the contribution shall be determined as follows: Parent will provide data to Parent's actuary that is complete and accurate upon which Parent's actuary's calculations will be based. Parent shall cause Parent's actuary to calculate the Accumulated Benefit Obligation for each Spunoff Pension Plan (determined as of the Closing Date in accordance with Statement of Financial Accounting Standards No. 87) using a discount rate equal to 6.25%, and all other actuarial assumptions, procedures, processes and methods (disregarding changes in procedures and processes under Parent's actuary's valuation software that also apply on a consistent basis to Parent's actuary's other clients) used in the preparation of the January 1, 2003 Milliman USA Actuarial Valuation Report for each respective Pension Plan and participant data as of the date of the spinoff (recognizing service, age, salary and benefit multipliers for benefit accrual purposes through the Closing Date (but not taking into account new hires, deaths, disabilities, terminations or other such changes in participant data that occur after the date of the spinoff and on or prior to the Closing Date) if the spinoff happens prior to Closing) (the "ABO"). The market value of plan assets shall be determined as of the day prior to the Closing Date for each respective Spunoff Pension Plan and subtracted from the ABO for each respective Spunoff Pension Plan, and Parent shall contribute the resulting amount to each respective Spunoff Pension Plan. The contribution shall be made in the form of cash or marketable securities that are acceptable to Holdings, provided that such acceptance shall not be unreasonably withheld.

11.4.2.2 Immediately after the contributions determined under Section 11.4.2.1 are made, Parent shall assign to Holdings, and Holdings shall assume from Parent, the sponsorship of the Spunoff Pension Plans. Parent shall take all actions necessary to satisfy requests of the Pension Benefit Guaranty Corporation with respect to such spinoff and transfer, including without limitation, the posting of bonds or collateral.

11.4.2.3 Parent shall cause Parent's actuary to determine and certify in writing to Holdings in form and substance satisfactory to Parent and Holdings, on or prior to the Closing Date, that (i) the spinoff under Section 11.4.2.1 complies with Code Section 414(l), (ii) that the ABO that was calculated for each respective Spunoff Pension Plan under Section 11.4.2.1 was calculated in accordance with Section 11.4.2.1 and (iii) each participant's accrued benefit used in the determination of the ABO was calculated in accordance with the terms of each respective Spunoff Pension Plan based on the participant data provided to such actuary by Parent. Not later than 30 days prior to Closing, Parent (or Parent's actuary) shall provide to Holdings' actuary all information necessary to verify such calculations. Not later than 15 days prior to Closing, Parent (or Parent's actuary) shall provide to Holdings' actuary all information necessary to verify such calculations, including without limitation, the accrued benefit and ABO for each Employee included in the calculation and all demographic information pertaining to the calculation.

11.4.3 At Closing Parent shall fully vest and freeze the benefits accrued to date by participating Transferred Employees under the Boise Cascade Corporation Supplemental Pension Plan and it shall pay such frozen benefits to affected Transferred Employees when and as they may become due under such plan. Prior to the Closing, Parent shall amend the Boise Cascade Corporation Supplemental Early Retirement Plan for Executive Officers to provide (i)

60

for the recognition of service for Transferred Employees with Holdings and its Affiliates after the Closing Date for the purpose of meeting vesting and eligibility requirements and (ii) that to the extent that any Transferred Employee participating under such plan as of the Closing Date retires from Holdings and its Affiliates after reaching age 55 with 10 or more years of service and 5 or more years of service as an executive officer (taking into account service with Parent and its Affiliates prior to the Closing Date and service with Holdings and its Affiliates on and after the Closing Date), such plan will provide a benefit to such Transferred Employee as if he or she had met all applicable age and service requirements as of the Closing. Holdings shall create plans that in the aggregate are substantially comparable to the Unfunded Pension Plans for the affected Employees, which shall be maintained without material change at least two years after Closing. Holdings shall provide participants in the Unfunded Pension Plans with a past service credit reflecting their past service with Parent and it shall be entitled to an offset for benefits earned under such plans equal to the participating Transferred Employee's frozen benefit under Parent's plan.

11.4.4 Holdings will maintain the Spunoff Pension Plans and Holdings' replacement unfunded pension plans without any material and adverse (to the Transferred Employees) change for two years after Closing.

# 11.5 <u>Defined Contribution Retirement Plans</u>.

11.5.1 Effective as of the Closing Date or as soon as reasonably practicable thereafter, Holdings shall establish one or more defined contribution retirement plans (collectively, the "Holdings DC Plans") which will provide each of the Transferred Employees for a period of at least two years matching contribution percentages that in the aggregate are substantially comparable to that available to such Transferred Employees immediately prior to the Closing Date under the Defined Contribution Plans, provided that Holdings shall not be required to establish an Employee Stock Ownership Plan (an "ESOP") and provided further that Holdings shall not be required to maintain a Parent common stock account. Except as not permitted by a trustee or recordkeeper of any Holdings will provide Transferred Employees with an opportunity to transfer any Parent common stock they hold in the Defined Contribution Plans (not relating to the ESOP accounts) into a qualified plan account that accepts such Parent common stock.

11.5.2 Each of the Transferred Employees shall receive credit for eligibility and vesting purposes under the Holdings DC Plans for all service credited under the Defined Contribution Plans.

11.5.3 As soon as practicable following the Closing Date, but in no event later than 60 days after the Closing Date, all assets and liabilities of the Defined Contribution Plans attributable to the Transferred Employees' vested and non-vested account balances shall be transferred (within the meaning of Section 1.414(l)-1(o) of the Treasury Regulations) in the form of cash or other marketable securities reasonably acceptable to Holdings, Parent common stock not relating to the ESOP accounts (except as not permitted under Section 11.5.1) and notes associated with plan loans to the Holdings DC Plans. Parent agrees not to place any Transferred Employee's plan loan into default on or after the Closing Date and prior to the transfer of account balances under this paragraph.

61

- 11.6 <u>Incentive Compensation and Deferred Compensation Plans</u>. Holdings will not be obligated to assume Parent's deferred compensation and incentive compensation plans. Parent will discharge its obligations to Employees under such plans in accordance with their terms and Section 11.1 hereof.
- 11.7 <u>Vacation</u>. Holdings shall establish a vacation policy or policies that in the aggregate are substantially comparable to Parent's current vacation policies in effect immediately prior to the date hereof for Transferred Employees and are in compliance with applicable collective bargaining agreements. Such policies shall recognize service credit for all past service recognized for Transferred Employees under Parent's policies in effect on the date hereof. Except insofar as Parent is required by applicable state law to compensate Transferred Employees in cash upon termination of employment for earned and unused vacation, Holdings shall assume the liability for such accrued and unused vacation to the extent accrued as a Current Liability in the computation of Closing Net Working Capital and discharge such liability in accordance with the terms of Holdings' established policies, provided that such policies shall not unreasonably restrict the use of such accrued and unused vacation. To the extent such a liability transfer is not permitted by applicable state law, no liability assumption shall occur and Parent shall pay affected Transferred Employees in cash for earned and unused vacation when and as required by applicable state law.
- Severance and Retention Obligations. Parent will retain all severance, change of control, retention and other similar obligations that arise prior to the Closing or with respect to terminations of employment that occur prior to the Closing Date or with respect to resignations by Transferred Employees (whether prior to or after the Closing Date) for "good reason" (within the meaning of the 2004 Severance Agreements) where events forming the basis for the claim of "good reason" occurred prior to the Closing Date (unless the event forming the basis for such claim of "good reason" consists of the terms of employment offered by Holdings). Parent also will retain (i) all employment, change of control, retention and similar agreements (and obligations thereunder) other than temporary employment contracts that are terminable without penalty at any time at the will of the employer with a term of less than one year and at an annual compensation rate of less than \$75,000 and that provide for no payments of severance, (ii) all payments, bonuses or other obligations to provide compensation or benefits as a result of the occurrence of the transaction and (iii) all obligations to provide retention payments to Employees, including officer and non-officer Transferred Employees, pursuant to retention plans, policies or agreements in existence as of the Closing Date (including, without limitation, the 2004 Retention Bonus Plan) regardless of whether such obligations are or become payable prior to, on or after the Closing Date. After Closing, Holdings will adopt a general severance policy that will be in the aggregate substantially comparable to Section VII.A. of the Parent's Termination of Employment Corporate Policy Number 10.2, dated July 26, 2004 (which is attached hereto as Schedule 11.8 in its currently applicable form and the form applicable on and after such date) ("Holdings General Severance Policy"). Holdings will be responsible for all severance costs under the Holdings General Severance Policy with respect to Transferred Employees who are terminated by Holdings after Closing and under the 2004 Severance Agreements `for any Transferred Employees who are officers immediately prior to Closing and who are terminated by Holdings after Closing. During the two year period after Closing, for all terminations of Transferred Employees who are not officers as of the Closing Date, Holdings shall provide severance benefits at a rate no lower than the maximum severance rate (within the

62

applicable severance range) which would have been applicable to such Employee at the time of Closing under Section VII.A. of the Parent's Termination of Employment Corporate Policy Number 10.2, dated July 26, 2004.

- 11.9 Workers' Compensation. Holdings shall assume all liabilities arising from workers compensation claims, whether characterized as medical or indemnity, or arising under any other similar government-mandated programs which are based on injuries incurred by Employees or former employees in connection with the Business, without regard to the date of occurrence of the injury or injuries giving rise to the workers' compensation claim. Holdings shall have the right to submit claims under relevant workers' compensation insurance policies and related agreements maintained by Parent and its affiliates prior to the Closing Date (each, a "Pre-Closing Workers' Compensation Policy") for claims brought by any current or former employee of the Business arising out of an event occurring prior to the Closing Date to the full limits of such Pre-Closing Workers' Compensation Policy to the extent covered by such Pre-Closing Workers' Compensation Policy. With respect to each Pre-Closing Workers' Compensation Policy other than those which are assigned to Holdings, Parent and its Affiliates shall (i) ensure such policies are in full force and effect as of the Closing Date; (ii) not take any action or inaction to amend or terminate such policies after the Closing Date with respect to coverage for Holdings; and (iii) maintain any letter-of-credit or security bonds needed to maintain any such policy. Holdings and Parent shall cooperate fully with the relevant insurer in the defense of any claim brought under any insurance policies pursuant to this Section 11.9.
- Retiree Benefits. Parent shall be liable for all liabilities and obligations to provide coverage in connection with claims for benefits brought by or in respect of former and retired employees of the Business (and their covered dependents) who had terminated employment or retired prior to Closing (including Employees who are eligible to retire as of the day prior to Closing whose employment is terminated pursuant to Section 11.1) under any of Parent's retirement welfare benefit plans, including, but not limited to, such liabilities with respect to medical, dental, life insurance, health, accident or disability benefits, but excluding any claims for or relating to workers' compensation benefits for which Holdings is responsible under Section 11.9 hereof. Holdings will assume the obligation to provide retiree welfare benefits to (i) active union Transferred Employees after their post-Closing retirements (in accordance with the applicable collective bargaining agreements), (ii) active non-union hourly Transferred Employees in Parent's Boise Building Solutions, Manufacturing Division after their post-Closing retirements (with a 40% employer subsidy that may be reduced or eliminated in the future)and (iii) active salaried Transferred Employees after their post-Closing retirements (with the entire cost paid by such employees); provided, however, that Holdings shall have the right (subject to applicable collective bargaining agreements) to modify or terminate such benefits in the future.
- 11.11 <u>No Third-Party Beneficiary.</u> This Agreement is being entered into solely for the benefit of the parties hereto, and the parties do not intend that any Employee or any other Person shall be a third-party beneficiary of the covenants by either Parent or Holdings contained in this Agreement.
- 11.12 <u>Workers Adjustment and Retraining Act ("WARN")</u>. Holdings shall indemnify and hold Parent harmless from and against any and all liabilities, claims and losses (including attorney's fees, court costs, and other litigation expenses) arising under the Worker Adjustment

63

and losses (including attorney's fees, court costs, and other litigation expenses) arising under the WARN Act by reason of the shutdown of any Facility or layoff of Employees which occurs or has occurred prior to the Closing.

- 11.13 <u>Non-solicitation of Employees</u>. For a period of two years following the Closing, Parent and its Affiliates shall not directly or indirectly solicit any Transferred Employee for employment without the prior consent of Holdings. The foregoing shall not prohibit Parent from (i) soliciting any Transferred Employee who has terminated his or her employment with Holdings prior to such solicitation, or (ii) offering employment to any Employee who approaches Parent without prior solicitation by Parent, whether in response to a general advertisement by Parent, or otherwise.
- 11.14 <u>Modifications to Benefits</u>. Subject to the two-year restrictions on modification of compensation and benefits set forth in this Article 11 and any other applicable restrictions under collective bargaining agreements, Holdings may modify or terminate its compensation and benefit plans at any time in its discretion; provided, however, that notwithstanding the various two-year restrictions on modification of compensation and benefits set forth in this Article 11, Holdings may initially adopt compensation and benefit plans different from those provided by Sellers and may modify its compensation and benefit plans during such two-year period following the Closing, in either case so long as its compensation and benefit plans (including equity arrangements), in the aggregate, are substantially comparable to that provided to Transferred Employees immediately prior to the Closing Date or as otherwise consistent with changes under collective bargaining agreements.

#### Article 12 Tax Matters

#### 12.1 <u>Section 338(h)(10) Election</u>.

- 12.1.1 (i) At Holdings' option, Parent and Holdings (and/or its designated Permitted Affiliate Purchaser(s)) shall jointly make Elections (and any comparable election under state or local tax law) with regard to each of the Transferred Subsidiaries for which such an election may be made;
- (ii) Parent and Holdings shall, as promptly as practicable following the Closing Date, cooperate with each other to take all actions necessary and appropriate (including filing such forms, returns, elections, schedules and other documents as may be required) to effect and preserve timely Elections in accordance with the provisions of Treasury regulations promulgated under Section 338 of the Code (or any comparable provisions of state or local tax law) or any successor provisions:
- (iii) Parent and Holdings shall report the purchase by Holdings of the Transferred Subsidiaries pursuant to this Agreement consistent with the Elections (and any comparable elections under state or local tax laws) and shall take no position inconsistent therewith in any Tax Return, any proceeding before any taxing authority or otherwise; and

64

- (iv) Nothing in this Agreement shall prevent Holdings or its Affiliates from making an election under Section 338(g) of the Code with respect to the acquisition (or deemed acquisition) of the equity of each Transferred Subsidiary organized under the laws of a foreign jurisdiction.
- 12.1.2 In connection with each Election, Holdings shall prepare, and Parent shall agree to (which agreement shall not be unreasonably withheld), the determination of the "Aggregate Deemed Sales Price" (as defined under Treasury regulation Section 1.338-4) and the allocation of each such "Aggregate Deemed Sales Price" among the assets of each respective Transferred Subsidiary, which shall be consistent with the Allocation contemplated by Section 3.4 hereof. The determination of the amount of each such "Aggregate Deemed Sales Price" and the allocation thereof shall be made in accordance with Section 338(b) of the Code and applicable Treasury regulations. Parent and Holdings shall (i) be bound by such determination and such allocation for purposes of determining any Taxes, (ii) prepare and file, and cause their affiliates to prepare and file, Tax Returns on a basis consistent with each such determination of the "Aggregate Deemed Sales Price" and each such allocation, and (iii) take no position, and cause their affiliates to take no position, inconsistent with any such determination or allocation on any applicable Tax Return in any proceeding before any taxing authority or otherwise. In the event that any such allocation and/or determination is disputed by any Tax Authority, the party receiving notice of the dispute shall promptly notify the other party hereto concerning resolution of the dispute.

#### 12.2 Tax Return Filing and Payment of Taxes Responsibility.

- 12.2.1 Parent shall (i) prepare and file, or cause to be prepared and filed, on a timely basis (in each case, at its sole cost and expense) all Tax Returns with respect to the Transferred Subsidiaries or otherwise relating to the Assets or the Business for all Tax Periods ending on or before the Closing Date (the "Pre-Closing Returns"), and (ii) pay, or cause to be paid, all Taxes shown to be due on such Pre-Closing Returns.
- 12.2. Holdings shall (i) prepare and file, or cause to be prepared and filed, on a timely basis (in each case, at its sole cost and expense) all other Tax Returns for all taxable periods ending after the Closing Date with respect to the Transferred Subsidiaries or otherwise relating to the Assets or to the Business, including Straddle Period Returns, and (ii) pay, or cause to be paid, all Taxes shown to be due on such Tax Returns.
- 12.2.3 In connection with any Straddle Period Return, Parent shall be liable for and shall pay, or cause to be paid, any Taxes for which Parent has agreed to indemnify Holdings pursuant to Section 12.4 hereof. Holdings shall provide Parent with copies of any Straddle Period Returns at least twenty (20) days prior to the due date thereof (giving effect to any extensions thereto), accompanied by a statement calculating in reasonable detail the Parent indemnification obligation pursuant to Section 12.4 hereof (the "Tax Indemnification Obligation"). Parent shall pay, or cause to be paid, to Holdings the amount of the Tax Indemnification Obligation within ten (10) days of receiving copies of the Straddle Period Returns (the "Parent Preliminary Tax Payment"). If Parent objects to the determination of the Tax Indemnification Obligation, Holdings and Parent shall use reasonable efforts to resolve such dispute, but if a final resolution is not obtained within fifteen (15) days, the dispute shall be

65

submitted to the Independent Accountants with instructions that such Independent Accountants render a determination with respect to the dispute within thirty (30) days. The determination of the Independent Accountants shall be conclusive and binding upon the parties in the manner and to the same effect as a binding arbitration award. All costs and expenses incurred relating to the engagement of the Independent Accountants shall be shared equally by Parent and Holdings. If the Tax Indemnification Obligation, as finally determined, exceeds the amount of the Parent Preliminary Tax Payment, Parent shall promptly pay, or cause to be paid, the amount of such excess to Holdings. If the Tax Indemnification Obligation, as finally determined, is less than the amount of the Parent Preliminary Tax Payment, Holdings shall promptly refund the amount of such deficiency to Parent.

12.2.4 Parent and Holdings shall reasonably cooperate, and shall cause their respective affiliates, officers, employees, agents, auditors and representatives to reasonably cooperate, in preparing and filing all Tax Returns (including amended returns and claims for refund), including the issuance of a power

of attorney and maintaining and making available to each other all records necessary in connection with Taxes and in resolving all disputes and audits with respect to all taxable periods relating to Taxes.

12.3 <u>Transfer and Similar Taxes</u>. Notwithstanding any other provision of this Agreement to the contrary, all fees, sales (including, without limitation, bulk sales), use, transfer, gains, stamp, duties, value-added, recording and similar Taxes ("Transfer Taxes") incurred in connection with the transactions contemplated by this Agreement shall be borne equally by Holdings and/or its Subsidiaries, on the one hand, and Parent and its Retained Subsidiaries, on the other hand. The party charged by law with the duty of making the necessary filings and returns with respect to such Transfer Taxes shall do so in accordance with applicable law, and shall be entitled to reimbursement from the other party to the extent that such party, in connection with such filings, pays an amount in excess of what it is required to pay hereunder.

# 12.4 <u>Tax Indemnification</u>.

12.4.1 Parent shall indemnify Holdings and its Affiliates and hold them harmless from and against (i) any Taxes of Sellers and its Affiliates (other than Taxes described in Section 12.4.2), (ii) any liability for Taxes of the Transferred Subsidiaries or otherwise relating to the Assets or Business for all taxable periods ending on or before the Closing Date and for the Pre-Closing Period, (iii) any liability for Taxes which result from (A) the deemed sale of assets pursuant to the Elections or any election under Section 338(g) of the Code, and (B) the deemed sale of assets pursuant to any comparable elections under state or local tax laws, (iv) any liability for Taxes imposed upon any of the Transferred Subsidiaries pursuant to Treasury Regulation Section 1.1502-6 (or similar provision under any state, local or foreign Tax law) as a result of being a member of an affiliated group, within the meaning of Section 1504 of the Code (or similar provision under any state, local or foreign Tax law), prior to the Closing, and any liability for Taxes imposed on any of the Transferred Subsidiaries as a transferee or successor (where such Transferred Subsidiary became a transferee or successor before the Closing) or by contract (where such contract was entered into prior to the Closing), and (v) any Loss attributable to any breach of the representations and warranties contained in Section 7.8.

66

- 12.4.2 Holdings shall indemnify Parent and its affiliates and hold them harmless from and against (i) any liability for Taxes of the Transferred Subsidiaries or otherwise relating to the Assets or Business for any taxable period ending after the Closing Date (except to the extent such taxable period began before the Closing Date, in which case Holdings' indemnity will cover only that portion of any such Taxes that is not attributable to the Pre-Closing Period), and (ii) any liability for sales and use Taxes, value added Taxes, and real and personal property Taxes attributable to Pre-Closing Periods to the extent that each is accrued on the Closing Balance Sheet and included as a Current Liability in the computation of Closing Net Working Capital.
- 12.4.3 In the case of any Straddle Period, the liability for Taxes for the Pre-Closing Period shall be determined on an interim-closing-of-the-books basis as if such taxable period ended on and included the Closing Date, except that (i) all standard deductions, exemptions, allowances and other similar items shall be apportioned to the pre-Closing period on a per diem basis and (ii) real and personal property Taxes shall be apportioned between Parent and Holdings in accordance with the principles under Section 164(d) of the Code.

# 12.5 <u>Procedures Relating to Indemnification of Tax Claims</u>.

- 12.5.1 If a claim for Taxes is made or a notice of an audit is issued by any Tax Authority in writing, which, if successful, might result in an indemnity payment pursuant to Section 12.4, the party seeking indemnification (the "Tax Indemnified Party") shall promptly notify the other party (the "Tax Indemnifying Party") in writing of such claim (a "Tax Claim") within a reasonably sufficient period of time to allow the Tax Indemnifying Party effectively to contest such Tax Claim, and in reasonable detail to apprise the Tax Indemnifying Party of the nature of the Tax Claim, and provide copies of all correspondence and documents received by it from the relevant Tax Authority. Failure to give prompt notice of a Tax Claim hereunder shall not affect the Tax Indemnifying Party's obligation under Section 12.4, except to the extent that the Tax Indemnifying Party is materially prejudiced by such failure to give prompt notice.
- 12.5.2 With respect to any Tax Claim which might result in an indemnity payment to Holdings pursuant to Section 12.4 (including, without limitation, Taxes relating to a Pre-Closing Period), Parent shall control all proceedings taken in connection with such Tax Claim to the extent relating to periods ending prior to the Closing or any Pre-Closing Period and, without limiting the foregoing, may in its sole discretion and at its sole expense pursue or forego any and all administrative appeals, proceedings, hearings and conferences with any Tax Authority with respect thereto, and may, in its sole discretion, either pay the Tax claimed and sue for a refund where applicable law permits such refund suits or contest such Tax Claim. Parent shall not under any circumstances settle or otherwise compromise any Tax Claim referred to in the preceding sentence without Holdings' prior written consent, not to be unreasonably withheld. In connection with any proceeding taken in connection with such Tax Claim, (i) Parent shall keep Holdings informed of all material developments and events relating to such Tax Claim if involving a material liability for Taxes, and (ii) Holdings shall have the right to participate, at its sole expense, in (but not control) any such proceedings. Holdings shall cooperate with Parent in contesting such Tax Claim, which cooperation shall include, without limitation, the issuance of a power of attorney, the provision to Parent of records and information which are reasonably relevant to such Tax Claim, and making employees available to provide additional information

67

or explanation of any material provided hereunder or to testify at proceedings relating to such Tax Claim. In connection with such cooperation, Parent shall reimburse Holdings for any out-of-pocket expenses incurred with respect thereto.

- 12.5.3 With respect to any Tax Claim not described in the preceding paragraph (including that which might result in an indemnity payment to Parent pursuant to Section 12.4), Holdings shall control all proceedings in accordance with provisions that are parallel to those in Section 12.5.2.
- Refunds and Credits. Any refunds and credits of Taxes of any of the Transferred Subsidiaries or otherwise relating to the Assets or the Business realized after the Closing Date with respect to (i) any taxable period ending on or before the Closing Date shall be for the account of Parent, and if received or utilized by Holdings or any of its Affiliates, shall be paid to Parent within five (5) business days after Holdings or any of its Affiliates receives such refund or utilizes such credit, (ii) any taxable period beginning after the Closing Date shall be for the account of Holdings, and if received or utilized by Parent, or any of its Affiliates, shall be paid by Parent to Holdings within five (5) business days after Parent, or any of its Affiliates, receives such refund or utilizes such credit, (iii) any Straddle Period shall be apportioned between Parent and Holdings in the same manner as such Taxes originally had been allocated pursuant to Sections 12.2.3 and 12.4.3 hereof, and (iv) any Election (and any comparable elections under state or local tax laws) shall be for the account of Parent (but only to the extent Sellers paid the Taxes being refunded or credited), and if received or utilized by Holdings, any of its Affiliates or either of the Transferred Subsidiaries shall be paid to Parent within five (5) business days after Holdings, any of its Affiliates or either of the Transferred Subsidiaries such credit. Notwithstanding the foregoing, Parent shall not be entitled to the benefit of any refund or credit to the extent such refund or credit (i) arises from a step-up in the Tax basis of Assets as a result of the Elections, (ii) is attributable to the carryback of a loss arising after the Closing Date, or (iii) was reflected as an asset on the Closing Balance Sheet.

- 12.7 <u>Termination of Tax Sharing Agreements</u>. On or prior to the Closing Date, all Tax sharing agreements and similar arrangements to which either of the Transferred Subsidiaries is a party or otherwise relating to the Assets or the Business shall be terminated, all obligations thereunder shall be settled, and no additional payments shall be made under any thereof after the Closing Date.
- 12.8 <u>Employee Payroll Information</u>. Parent shall transfer to Holdings any records relating to withholding and payment of income and unemployment Taxes (federal, state and local) and FICA Taxes with respect to wages paid to Transferred Employees by Parent during the calendar year in which the Closing occurs (including, without limitation, Forms W-4, Employee's Withholding Allowance Certificate). Holdings shall provide the Transferred Employees with Forms W-2, Wage and Tax Statement, for such calendar year setting forth the wages paid and Taxes withheld with respect to the Transferred Employees for the calendar year in which the Closing occurs by Parent and Holdings as predecessor and successor employers, respectively, as provided by Revenue Procedure 96-60.

68

12.9 <u>Survival of Tax Provisions</u>. Notwithstanding anything to the contrary in Section 17.1, any claim to be made pursuant to this Section 12 must be made before the 60<sup>th</sup> day after the expiration (with extensions) of the applicable statute of limitations relating to the Taxes at issue.

#### Article 13 Risk of Loss

If any loss or damage is suffered prior to Closing to any portion of the Non-Timber Assets which is not a Material Adverse Change, Parent shall at its sole cost and expense repair or replace such damaged or lost assets as soon as reasonably practicable and the Closing shall proceed without delay or adjustment to the Non-Timber Purchase Price on account of such loss or damage. Holdings shall not, in such event, be entitled to or have any claim against the proceeds of insurance held by Parent to the extent covering such loss, unless Parent shall fail to repair or replace such damaged or lost assets prior to Closing, in which event Holdings shall be entitled to have a claim against the proceeds of such insurance and Parent shall be responsible for the costs thereof in excess of the amount of such proceeds. If any of the Timberlands are destroyed prior to Closing the Timberlands Purchase Price shall be adjusted by an amount equal to the fair market value of the standing timber destroyed. If any of the Assets are condemned prior to Closing and the magnitude of such condemnation is not sufficient to constitute a Material Adverse Change, Closing shall proceed without delay or adjustment of the Non-Timber Purchase Price or the Timberlands Purchase Price and Parent shall deliver the condemnation proceeds to Holdings, or if such condemnation proceedings have not been concluded prior to Closing, Parent shall at Closing assign its rights in such proceeding to Holdings. The provisions of this Article 13, including the obligations of Parent hereunder, shall survive the Closing and the delivery of the Deeds.

#### Article 14 Closing and Termination

- 14.1 <u>Closing</u>. Subject to satisfaction or waiver (in writing by the party entitled to waive satisfaction of such condition) of the conditions specified herein, the Closing shall occur at 9:00 a.m., local time, on September 30, 2004 at the offices of Parent, 1111 West Jefferson Street, Boise, Idaho 83728-0001, or at such other time and place as Purchaser and Parent mutually agree upon in writing. If Closing does not occur on such date, or any other date agreed upon by the parties, because one or more conditions precedent are not satisfied or waived in writing by the party entitled to waive satisfaction of such condition, then in such event the Closing shall be delayed to a later date agreed upon by the parties, and if the parties are unable to agree upon such a date, it shall be the first business day which is at least five (5) days after the day all of the conditions precedent are satisfied or waived. The date and time at which the Closing occurs shall be the "Closing Date."
  - 14.2 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing as follows, and in no other manner:
    - 14.2.1 By mutual consent of Holdings and Parent.
- 14.2.2 By either Parent or Holdings, if, through no fault of the party seeking termination, the Closing shall not have occurred on or prior to the Outside Date.

69

- 14.2.3 By either Parent or Holdings if any court or governmental body having competent jurisdiction shall have issued a preliminary injunction or other similar or final order enjoining the transactions contemplated hereby.
- 14.2.4 By Holdings in the event any representation or warranty made herein for the benefit of Holdings, or in any certificate, schedule or documents furnished to Holdings, pursuant to this Agreement is untrue in any material respect, or Parent shall have defaulted in any material respect in the performance of any material obligation under this Agreement, which breach or default has not been cured within 10 days (or such longer period as is reasonably necessary to cure such breach or default) after notice of such breach or default.
- 14.2.5 By either Parent or Holdings if Parent has delivered notice to Holdings pursuant to Section 10.11 that it intends to accept a Company Superior Offer, and after five business days from delivery of such notice Holdings has not submitted a Matching Offer to Parent as described in Section 10.11.
- 14.2.6 By Parent at any time prior to Closing if any representation or warranty made herein for the benefit of Parent, or in any certificate, schedule, or documents furnished to Parent, pursuant to this Agreement is untrue in any material respect, or Holdings shall have defaulted in any material respect in the performance of any material obligation under this Agreement, which breach or default has not been cured within 10 days (or such longer period as is reasonably necessary to cure such breach or default) after notice of such breach or default.
- 14.3 <u>Effect of Termination</u>. In the event of the termination of this Agreement as provided in Section 14.2, written notice thereof shall forthwith be given by the terminating party or parties to the other party or parties specifying the provision hereof pursuant to which such termination is made, and this Agreement shall forthwith become null and void, and there shall be no liability on the part of any of the parties hereto except that the obligations of the parties hereto contained in this Section 14.3 and in Article 19 (other than Section 19.16) shall survive any termination hereof, and except that nothing herein will relieve any party from liability for any breach of this Agreement prior to such termination.
- 14.3.1 If this Agreement is terminated pursuant to Section 14.2.5, Parent shall pay Holdings or its designee the sum of \$185,000,000 in cash on the date of such termination, and payment of such amount shall be Holdings' sole and exclusive remedy with respect to the termination of this Agreement pursuant to Section 14.2.5.

# Article 15 Holdings Conditions to Closing; Parent Closing Deliveries

15.1 <u>Holdings</u>. Neither Holdings nor any of its Subsidiaries shall have any obligation to attend and carry out the actions required of it at the Closing unless all of the following conditions precedent shall have been satisfied or waived in a writing executed by Holdings:

15.1.1	Continued Truth of Representations and Warranties. Without giving effect to any disclosure pursuant to Section 10.2.2.3, all of the
representations and warra	anties of Parent contained in this Agreement not qualified by materiality or Material Adverse Effect shall continue to be true and correct at th
Closing in all material re	spects as if made on and as of the

70

Closing Date, and all of the representations and warranties of Parent contained in this Agreement qualified by materiality or Material Adverse Effect shall continue to be true and correct at the Closing in all respects as if made on and as of the Closing Date.

- 15.1.2 <u>Performance of Obligations</u>. Parent and its Subsidiaries shall have performed or tendered performance of, in all material respects, each and every one of their respective obligations hereunder which by its terms is to be performed at or prior to Closing.
- 15.1.3 Third-Party Consents. Parent shall have obtained and delivered to Holdings all Consents to the assignment of those Contracts and Facility Leases listed on Schedule 15.1.3 (the "Required Consents"), in form and substance reasonably satisfactory to Holdings, or the parties shall have reached agreement on a mutually satisfactory work around arrangement pursuant to Section 5.4.
- 15.1.4 <u>Legal Proceedings</u>. No order shall have been entered and not vacated by or before any court, administrative agency or other governmental authority to restrain, prohibit or invalidate any of the transactions contemplated by this Agreement. No action, suit, proceeding or investigation by any administrative agency, governmental authority, or other Person regarding the transactions contemplated by this Agreement shall have been instituted and be continuing. In the event that such shall occur, Closing shall be delayed, but not beyond the Outside Date, while the parties take commercially reasonable steps to resolve such litigation (which shall not include the payment of any sum of money by any party except in its sole discretion).
  - 15.1.5 Material Adverse Changes or Effects. Since May 31, 2004, there shall have been no Material Adverse Change or Material Adverse Effect.
- 15.1.6 <u>Permits and Governmental Consents</u>. All Governmental Authorizations required for the operation of the Business shall have been obtained by Holdings or transferred to or reissued in Holdings' name and all material Consents and approvals of any federal, state, county, local, or other governmental or regulatory body required for any of the execution, delivery, or performance of this Agreement shall have been obtained, except for any such Governmental Authorization or any Consent which is routinely and customarily given or granted, issued, or transferred in connection with similar transactions by the governmental entity empowered to do so after the transaction has closed, and the absence of which will not have a Material Adverse Effect on the Business or the Assets.
- 15.1.7 <u>HSR Filing</u>. The applicable waiting period under the HSR Act with respect to the actions contemplated by this Agreement shall have expired or been earlier terminated and any similar waiting period shall have expired and the applicable governmental consent shall have been obtained from the reviewing authorities in Brazil, Canada, and the United Kingdom.
- 15.1.8 <u>Environmental Assessment</u>. Holdings shall have completed its review of the Environmental Assessments to be provided by Parent and its other related environmental due diligence activities and shall have reasonably determined that there does not exist any environmental compliance or liability issue with respect to a Facility (other than a Retained Environmental Liability) which, individually or in the aggregate, would reasonably be expected

71

to have a Material Adverse Effect. In the event that Holdings makes such determination, Holdings shall have no obligation to attend and carry out the actions required of it at the Closing unless and until Parent either (a) conducts and presents the results of additional environmental assessment work with respect to the subject Facility which demonstrates, to the reasonable satisfaction of Holdings, that no reasonable expectation of a Material Adverse Effect exists; or (b) agrees to treat the subject matter of such determination as a Retained Environmental Liability or agrees to an alternative indemnification or risk allocation arrangement with respect thereto which is reasonably satisfactory to Holdings.

- 15.1.9 <u>Financial Statements</u>. Not less than 30 days prior to Closing, Parent shall have delivered to Holdings and its Subsidiaries the Audited Financial Statements, the Stub Period Financial Statements and the Subsequent Financial Statements prepared and presented in accordance with <u>Sections 7.6</u> and <u>10.13</u>. None of the financial statements included in the Audited Financial Statements or the Stub Period Financial Statements, as so delivered, will when compared to the corresponding unaudited financial statement for the same period included in the Historical Financial Statements or the Subsequent Financial Statements, reflect any material diminution in the value of the Business as a whole, or any material diminution in the net income, cash flows, or net worth of the Business.
- 15.1.10 <u>Amended Paper Sales Agreement</u>. Parent and Boise Office Solutions shall have entered into the Amended Paper Sales Agreement as required by Section 18.2, the Amended Paper Sales Agreement shall be assigned to Holdings in the Closing, and such Amended Paper Sales Agreement shall be in full force and effect.
- 15.1.11 <u>Financing Arrangements; No Liens or Indebtedness</u>. Holdings and its Subsidiaries shall have received all of the proceeds of the debt financing contemplated by the Debt Commitment Letters, the equity financing contemplated by the Equity Commitment Letter and the Reinvestment contemplated to be received by Holdings and its Subsidiaries at the Closing, in each case on terms as described in the Commitment Letters (or in the case of the Reinvestment, the terms hereof and the Ancillary Agreements) or otherwise on a basis satisfactory to Holdings and Parent. Each of the Assets shall be free and clear of all Encumbrances (other than Permitted Encumbrances), and the Business shall not have or be liable for any Indebtedness or obligation giving rise to any lien on any of the Assets. Timber LLC shall have obtained the standby letter(s) of credit from the Credit Enhancement Banks as necessary to satisfy its obligations under Section 4.1.2.
- 15.1.12 <u>Title; Surveys</u>. Each of the Title Policies shall have the creditor's right exception deleted, and shall include an extended coverage endorsement (insuring over the general or standard exceptions) and all other endorsements reasonably requested by Holdings, in form and substance reasonably satisfactory to Holdings, and the Surveys shall not disclose any material encroachment from or onto any of the Titled Property or any portion thereof or any other material survey defect which has not been cured or, provided the Title Company will issue a further assurance endorsement with respect to such defect, insured over to Holdings' reasonable satisfaction prior to the Closing.
- 15.1.13 <u>Landlord Deliveries</u>. Without limiting any other covenant or agreement of Parent, Parent shall have obtained and delivered an estoppel certificate, a waiver of

landlord liens, collateral assignment of lease and/or leasehold mortgage from each landlord or other party whose consent thereto is required under any particular Facility Lease or Timberlands Lease with respect to those sites listed on Schedule 15.1.13 or for which such deliveries shall have been requested by the lenders in connection with the financing contemplated by the Debt Commitment Letters.

- 15.1.14 <u>Auditor Consent Letter</u>. Parent shall have obtained and delivered to Holdings and its Subsidiaries a copy of the Auditor Consent Letter and related "cold comfort" letter for the financing arrangements contemplated by the Debt Commitment Letter, each of which shall be in form and substance reasonably satisfactory to Holdings, its Subsidiaries and the underwriters for the debt financing contemplated by the Debt Commitment Letters.
- 15.1.15 <u>Spunoff Pension Plans</u>. Parent shall have made a contribution to each Spunoff Pension Plan, and each such Spunoff Pension Plan shall be fully funded on an Accumulated Benefit Obligation basis as described in Section 11.4.2.1.
  - 15.1.16 <u>Delivery of Closing Documents</u>. Parent shall have tendered delivery to Holdings of all of the following:
    - 15.1.16.1 <u>Certificates of Title</u>. A certificate of title for each registered motor vehicle held by Sellers or any of their Subsidiaries.
- 15.1.16.2 <u>Opinion of Counsel</u>. The opinion of J. W. Holleran, Senior Vice President and General Counsel of Parent, in the form attached as Schedule 15.1.16.2.
- 15.1.16.3 <u>Certified Resolutions.</u> A copy of the resolutions of the boards of directors or other governing bodies of Sellers authorizing the execution and performance of this Agreement and the Ancillary Agreements, certified by the Secretary or an Assistant Secretary of each of the Sellers;
- 15.1.16.4 <u>Closing Certificate</u>. A certificate signed by an officer of Parent to the effect that the conditions specified in Section 15.1.1, Section 15.1.2 and Section 15.1.5 have been satisfied in full;
- 15.1.16.5 <u>Assignments</u>. Such assignments, bills of sale, stock transfer powers, and other instruments of conveyance as may be required to convey the Assets from Sellers to Holdings, in form and substance reasonably satisfactory to Holdings and its Subsidiaries, together with assignments of Timberlands Leases shall be in recordable form and otherwise in form and substance reasonably satisfactory to Holdings and its Subsidiaries;
- 15.1.16.6 <u>Intellectual Property Transfers.</u> Documents acceptable for recordation in the United States Patent and Trademark Office, the United States Copyright Office and any other similar domestic or foreign office, department or agency for the conveyance of registered Intellectual Property otherwise in form and substance reasonably satisfactory to Holdings and its Subsidiaries.
- 15.1.16.7 <u>Deeds.</u> Deeds to the Realty and Timberlands satisfying the requirements of Section 9.2. Deeds to the Timberlands shall convey the Timber

73

- Co. Deeds to the Realty shall convey the Realty to Holdings or its nominated Permitted Affiliate Purchaser;
  - 15.1.16.8 <u>Minute Books</u>. The original minute books and stock transfer records of each Transferred Subsidiary.
- 15.1.16.9 <u>FIRPTA Certificate</u>. A duly executed affidavit of non-foreign status by each of Sellers described in Section 1445 of the Code and the regulations thereunder;
- 15.1.16.10 <u>Confidentiality Agreement Assignment</u>. An executed assignment by Parent of its rights as they relate to the Business under each of the confidentiality agreements executed by or on behalf of Parent and/or any of its Subsidiaries in connection with the proposed sale of the any of the Business, which assignment shall be in form and substance reasonably satisfactory to Holdings;
  - 15.1.16.11 <u>Ancillary Agreements</u>. Executed copies of each of the Ancillary Agreements to which Parent or its Subsidiaries is a party;
  - 15.1.16.12 Additional Items. Such additional documents, instruments, and other items as counsel for Holdings may reasonably request.
- 15.1.17 <u>Complete Schedules</u>. The parties accept, acknowledge, and agree that they are executing this agreement with the Schedules and Disclosure Memoranda hereto in draft form. The parties agree to work together in good faith and use their reasonable best efforts to agree upon and finalize mutually acceptable Schedules and Disclosure Memoranda within 72 hours after execution of this Agreement. The parties' reaching mutual agreement on the final form of the Schedules and Disclosure Memoranda is a condition to Closing hereunder.

# Article 16 Parent Conditions to Closing; Holdings Closing Deliveries

and

- 16.1 <u>Parent</u>. Parent shall have no obligation to attend and carry out the actions required of it at the Closing unless all of the following conditions precedent shall have been satisfied:
- 16.1.1 <u>Continued Truth of Representations and Warranties</u>. All of the representations and warranties of Holdings contained in this Agreement not qualified by materiality or Material Adverse Effect shall continue to be true and correct at the Closing in all material respects as if made on and as of the Closing Date and all of the representations and warranties of Holdings contained in this Agreement qualified by materiality or Material Adverse Effect shall continue to be true and correct at the Closing in all respects as if made on and as of the Closing Date.
- 16.1.2 <u>Performance of Obligations</u>. Holdings shall have substantially performed or tendered substantial performance of each and every one of its obligations hereunder which by its terms is to be performed at or prior to the Closing.

74

- 16.1.4 <u>Governmental Filings</u>. The applicable waiting period under the HSR Act with respect to the actions contemplated by this Agreement shall have expired or been earlier terminated and any similar waiting period shall have expired or the applicable governmental consent shall have been obtained from the reviewing authorities in Brazil, Canada, and the United Kingdom.
- 16.1.5 <u>Timber Note Credit Enhancement</u>. Timber LLC shall have obtained one or more standby letters of credit from the Credit Enhancement Banks and entered into the related agreements substantially consistent with the terms of the LC Commitment Letters and shall have tendered delivery of the Timber Notes and the standby letter(s) of credit to Parent and Southern.
  - 16.1.6 <u>Delivery of Closing Documents</u>. Holdings shall have tendered delivery to Parent of all of the following:
- 16.1.6.1 <u>Resolutions</u>. A copy of the resolutions of the boards of directors or other governing bodies of Holdings and the Permitted Affiliate Purchasers authorizing the execution and performance of this Agreement and each of the Ancillary Agreements certified by the secretary or an assistant secretary of each;
- 16.1.6.2 <u>Closing Certificate</u>. A certificate signed by an officer of Holdings on behalf thereof to the effect that each of the conditions to Closing specified in Section 16.1.1 and Section 16.1.2 have been satisfied in full;
- 16.1.6.3 <u>Opinion</u>. An opinion of Kirkland & Ellis LLP, counsel to Holdings, substantially in the form of Schedule 16.1.6.3 attached hereto;
- 16.1.6.4 <u>Purchase Price</u>. Receipt by Parent of cash in an amount equal to the sum of the cash portion of the Timberlands Purchase Price provided for in Section 4.1.1 and the Estimated Non-Timber Purchase Price determined pursuant to Section 4.2;
  - 16.1.6.5 <u>Timber Notes</u>. The Timber Notes and the standby letter(s) of credit called for in Section 4.1.2 hereof;
- 16.1.6.6 <u>Ancillary Agreements</u>. Executed copies of each of the Ancillary Agreements to which Holdings or any of its Subsidiaries is a party; and
  - 16.1.6.7 <u>Additional Items</u>. Such additional documents, instruments, and other items as counsel for Sellers may reasonably request.
- 16.1.7 <u>Complete Schedules</u>. The parties accept, acknowledge, and agree that they are executing this agreement with the Schedules and Disclosure Memoranda hereto in draft form. The parties agree to work together in good faith and use their reasonable best efforts to agree upon and finalize mutually acceptable Schedules and Disclosure Memoranda within 72

75

hours after execution of this Agreement. The parties' reaching mutual agreement on the final form of the Schedules and Disclosure Memoranda is a condition to Closing hereunder.

# Article 17 Survival of Representations and Warranties; Indemnifications

- 17.1 Survival. The representations and warranties of the parties shall survive the Closing for 18 months; provided that (a) the representations and warranties of Sellers contained in Section 7.9 and Section 7.11 shall survive the Closing until the fifth anniversary of the Closing Date, (b) the representations and warranties of Sellers contained in Sections 7.1 through 7.5 hereof, Section 7.16.2, Section 7.16.3, and Section 7.17 shall survive the Closing forever, (c) and the representations and warranties of Holdings contained in Sections 8.1 through 8.5 hereof shall survive the Closing forever. All claims, causes of action and rights of the parties based on the representations and warranties of the other party shall expire upon expiration of the applicable limitation period set forth in the immediately preceding sentence and no claim based on a breach of any such representation and warranty may be asserted unless written notice reasonably identifying the breach claimed shall have been given to the party allegedly in breach prior to expiration of such applicable limitation period. In addition, as between Sellers and Holdings and its Permitted Affiliate Purchasers, all representations and warranties contained in any Deed conveying Timberlands or Realty shall likewise survive indefinitely. No claim shall be made for breach of any representation or warranty contained in such Deeds except in accordance with the procedures provided for indemnification claims in this Article 17.
- 17.2 <u>Indemnification by Sellers</u>. Sellers shall, jointly and severally, indemnify, defend and hold harmless Holdings, its Subsidiaries and each of the Permitted Affiliate Purchasers and, if applicable, their respective directors, officers, shareholders, partners, members, agents and employees and their heirs, successors and assigns (the "Holdings Indemnified Parties") from, against and in respect and to the extent of any damages, claims, losses, charges, actions, suits, proceedings, deficiencies, interest, penalties, and reasonable costs and expenses of defense (including reasonable attorneys' fees, removal costs, remediation costs, closure costs, fines, penalties and expenses of investigation and ongoing monitoring) (collectively, the "Losses") imposed on, sustained, incurred or suffered by or asserted against any of the Holdings Indemnified Parties, directly or indirectly, relating to or arising out of:
- 17.2.1 any breach of any representation or warranty made by Parent or Sellers contained in this Agreement, or in any certificate, opinion, or other document delivered by or on behalf of Parent or Sellers pursuant to the requirements of this Agreement;
  - 17.2.2 the Retained Liabilities;
  - 17.2.3 the breach of any covenant or agreement of Parent or Sellers contained in this Agreement;
  - 17.2.4 provided such claim is made during the 5-year period commencing on the Closing Date, any Indemnified Environmental Liabilities; and
- 17.2.5 provided such claim is made during the 18-month period commencing on the Closing Date, any Losses arising from, related to or incurred in connection with any state of

76

## 17.3 Sellers Liability Limits.

- 17.3.1 Sellers shall not be liable to the Holdings Indemnified Parties pursuant to Section 17.2.1 for any Losses arising as a result of any breach of any representation or warranty except to the extent (and then only to the extent) the aggregate of such Losses (when added to the aggregate Losses under Section 17.2.4, whether or not in excess of the Environmental Deductible) exceeds the Deductible, and then only for all such Losses in excess thereof up to an aggregate amount (when added to the aggregate Losses under Section 17.2.4, whether or not in excess of the Environmental Deductible) equal to the Cap; provided that neither the Cap nor the Deductible shall apply to any breach of any representation or warranty of Sellers made in any of Sections 7.1 through 7.5 hereof, Section 7.16.1, Section 7.16.2, Section 7.16.3, Section 7.16.5, Section 7.16.6, or Section 7.17.
- 17.3.2 Sellers shall not be liable to the Holdings Indemnified Parties pursuant to Section 17.2.4 for any Losses relating to Indemnified Environmental Liabilities except to the extent (and then only to the extent) the aggregate of such Losses exceeds the Environmental Deductible, and then only for all such Losses in excess thereof up to an aggregate amount (when added to the aggregate Losses under Section 17.2.1) equal to the Cap.
- 17.3.3 The Deductible and the Cap shall not apply to Losses suffered by the Holdings Indemnified Parties from a breach by Sellers or any of their Subsidiaries of any of its covenants hereunder, its obligations hereunder in respect of the Retained Liabilities, or its obligations under Section 17.2.5. Claims made by any Holdings Indemnified Party against any of the Sellers for breach of any representation or warranty contained in any Deed given pursuant to this Agreement shall likewise not be subject to the operation of the Deductible and the Cap.
- 17.4 <u>Holdings Indemnity</u>. Holdings shall indemnify, defend and hold harmless Parent and each of its Retained Subsidiaries and their respective directors, officers, shareholders, partners, members, agents and employees and their heirs, successors and assigns (the "Parent Indemnified Parties") from, against and in respect and to the extent of any Losses directly or indirectly caused by or relating to (i) any breach by Holdings of any of its representations and warranties made in this Agreement, (ii) any breach of its covenants made in this Agreement, or (iii) its failure to discharge the Assumed Liabilities when and as they become due.
- 17.5 <u>Holdings Liability Limits</u>. Holdings shall not be liable to the Parent Indemnified Parties for any Losses arising as a result of any breach of any representation or warranty made by Holdings, except to the extent (and then only to the extent) the aggregate of such Losses exceeds the Deductible, and then only for all such Losses in excess thereof up to an aggregate amount equal to the Cap. The deductible and the Cap shall not apply to Losses suffered by the Parent Indemnified Parties from a breach by Holdings of any of its covenants hereunder or of its obligations in respect of the Assumed Liabilities.

77

## 17.6 <u>Indemnification Procedures</u>.

- 17.6.1 Any Person making a claim for indemnification pursuant to Section 17.2 or 17.4 above (an "Indemnified Party") must give the party from whom indemnification is sought (an "Indemnifying Party") written notice of such claim describing such claim with reasonable particularity and the nature and amount of such Loss to the extent that the nature and amount of such Loss is known at such time) (an "Indemnification Claim Notice") promptly after the Indemnified Party receives any written notice of any action, lawsuit, proceeding, investigation or other claim (a "Proceeding") against or involving the Indemnified Party by a Governmental Authority or other third party or otherwise discovers the liability, obligations or facts giving rise to such claim for indemnification; provided that, the failure to notify or delay in notifying an Indemnifying Party will not relieve the Indemnifying Party of its obligations pursuant to Section 17.2 or 17.4, as applicable, except to the extent that (and only to the extent that) such failure shall have caused the damages for which the Indemnifying Party is obligated to be greater than such damages would have been had the Indemnified Party given the Indemnifying Party prompt notice hereunder.
- 17.6.2 The Indemnifying Party shall have 30 days from the personal delivery or mailing of the Indemnification Claim Notice (the "Notice Period") to notify the Indemnified Party (i) whether or not the Indemnifying Party disputes the liability of the Indemnifying Party to the Indemnified Party hereunder with respect to such claim or demand, and (ii) whether or not it desires to defend the Indemnified Party against such claim or demand.
- 17.6.3 If (i) the Indemnifying Party agrees in writing to be responsible for the full amount of such Loss, and (ii) the claim for indemnification does not relate to a matter (A) that, if determined adversely, could reasonably be expected to expose the Indemnified Party to criminal prosecution or penalties, (B) that, if determined adversely, could reasonably be expected to result in the imposition of a consent order, injunction or decree which would restrict the activity or conduct of the Indemnified Party or any Affiliate thereof, or (C) for which the Indemnified Party shall have reasonably concluded, in good faith, after consultation with the Indemnifying Party, that such representation is likely to result in a conflict of interest or materially jeopardize the viability of such defense, then the Indemnifying Party shall have the right to defend the Indemnified Party by appropriate proceedings and shall have the sole power to direct and control such defense. If any Indemnified Party desires to participate in any such defense, it may do so at its sole cost and expense.

The Indemnifying Party in no event shall have any right to control (as opposed to participate in pursuant to Section 17.6.4 hereof) the defense of any claim and shall pay the expenses of the Indemnified Party's defense of such claim if:

- 17.6.3.1 the Indemnifying Party does not agree in writing to be responsible for the full amount of any claim;
- 17.6.3.2 the claim for indemnification relates to a matter (A) that, if determined adversely, could reasonably be expected to expose the Indemnified Party to criminal prosecution or penalties, (B) that, if determined adversely, could reasonably be expected to result in the imposition of a consent order, injunction or decree which would restrict the activity or conduct of the Indemnified Party or any Affiliate thereof, or (C) for which the Indemnified party shall have reasonably concluded, in good faith, after consultation with the Indemnifying Party,

78

that such representation is likely to result in a conflict of interest or materially jeopardize the viability of such defense; or

17.6.3.3 a court determines that the Indemnifying Party is not vigorously defending the claim.

17.6.4 If the claim relates to a matter for which both the Indemnifying Party and any Indemnified Party could be liable or responsible hereunder, such as a Loss for which both parties could be partially liable due to the Cap and Deductible, the Indemnifying Party and the Indemnified Parties shall cooperate in good faith in the defense of such action. No party shall settle any claim without the prior consent of the other party (which consent shall not be unreasonably withheld); provided, however, that an Indemnified Party shall not be required to consent to any settlement if the proposed settlement (i) does not provide for a full release of all claims against such Indemnified Party, (ii) is on a basis which would result in the imposition of a consent order, injunction or decree or any other restriction on the activity or conduct of such Indemnified Party, or (iii) is on a basis which could, in such Indemnified Party's judgment, expose such Indemnified

Party to criminal liability or require an admission of wrongdoing by such Indemnified Party; provided further that, the foregoing notwithstanding, an Indemnified Party may settle or compromise any claim without the prior consent of the Indemnifying Party if under Section 17.6.3 the Indemnifying Party had no right to control the defense of such claim. If an Indemnified Party does not consent to a definitive settlement proposed by the Indemnifying Party (with respect to which a settlement agreement has been agreed to by all parties other than the Indemnified Party) which settlement satisfies the foregoing clauses (i) through (iii) or if the Indemnifying Party does not consent to a settlement proposed by an Indemnified Party, then the party declining such settlement shall thereafter have full control of the defense of such claim, and the maximum liability of the party that proposed such settlement shall be determined as though such matter had settled on the terms so proposed, and, if applicable, the amount of the proposed settlement, together with all legal costs and expenses incurred in connection with such matter through and including the proposed settlement date, shall be deemed the amount of the Loss of the Indemnified Party for purposes of determining whether the Cap and Deductible have been met. If both parties agree to the settlement, the relative liabilities of the parties for such Losses shall be determined as provided in the other provisions of this Article 17.

- 17.6.5 All costs and expenses incurred by the Indemnifying Party in defending a claim or demand under Section 17.6.3, and all costs and expenses incurred by the Indemnified Party in defending a claim or demand which the Indemnifying Party has elected not to defend (including by virtue of its failure to give timely notice to the Indemnified Party) or is not permitted to defend under Section 17.6.3 shall be a liability of, and shall be paid by, the Indemnifying Party.
- 17.6.6 To the extent the Indemnifying Party shall direct, control or participate in the defense or settlement of any third-party claim or demand, the Indemnified Party will give the Indemnifying Party and its counsel access to, during normal business hours, the relevant business records and other documents, and shall permit them to consult with the employees and counsel of the Indemnified Party. The Indemnifying Party and Indemnified Parties shall use their best efforts in the defense of all such claims.

79

- 17.7 <u>Acknowledgement Regarding Liabilities</u>. Holdings and Sellers acknowledge the allocation of relative responsibility for liabilities under this Agreement is a material term of this Agreement, and that they have taken such matters into consideration in determining the financial and other terms of this transaction. Accordingly, the parties agree that a party shall be entitled to the indemnification or other remedies provided in this Agreement by reason of any breach of any such representation, warranty, covenant or agreement by another party notwithstanding whether such party or any employee, representative or agent of such party seeking to enforce a remedy knew or had reason to know of such breach. Furthermore, for all purposes of indemnification under this Article 17, determinations of whether a representation or warranty has been breached and of the amount of Losses arising from any such breach shall be made without giving effect to any materiality, Material Adverse Effect or Knowledge qualifications contained in such representation or warranty.
- 17.8 Computation of Losses Subject to Indemnification. The amount of any Loss for which indemnification is provided under this Article 17 shall be computed net of any insurance proceeds actually received by the Indemnified Party in connection with such Loss. Indemnification for any Loss shall be determined and paid without reduction for any Tax Benefits not yet realized by the Indemnified Party. The Indemnified Party will pay to the Indemnifying Party the amount of any Tax Benefits attributable to the Loss actually realized by the Indemnified Party promptly after such Tax Benefits are realized; provided, however, that in the event such Tax Benefits are realized prior to the indemnification payment hereunder, such indemnification payment shall be reduced by Tax Benefits previously realized in lieu of a separate payment to the Indemnifying Party. The amount of any Tax Benefit shall be determined (i) by comparing the liability of the Indemnified Party for Taxes, determined without the Loss and any corresponding indemnification payments, to the liability of the Indemnified Party for Taxes, taking into account the Loss and any corresponding indemnification payments, and (ii) by treating any items attributable to the Loss as the last items claimed by the Indemnified Party in any given Tax Period. The amount of any Loss for which indemnification is provided under this Article 17 shall exclude consequential damages (it being understood that a reduction in the value of the Business or the Assets shall not be considered consequential damages) and punitive damages by an Indemnified Party; provided that, any consequential damages and punitive damages of a third party for which an Indemnified Party is liable shall be included in computing such Indemnified Party's Loss.
- 17.9 Exclusive Remedy. Each party acknowledges that, except as set forth in Article 12 and Section 19.16, from and after the Closing this Article 17 constitutes the sole remedy of any party with respect to any matter arising hereunder (other than with respect to claims based on fraud), including any Losses or liability under any Environmental Law or with respect to any Hazardous Substances (including claims under the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA")) and expressly waives any other rights or causes of action, including under any Environmental Law (including CERCLA) or with respect to any claim involving the presence of or exposure to any Hazardous Substances.
- 17.10 <u>Tax Indemnities Separate</u>. The Tax indemnifications extended by the parties pursuant to Article 12 hereof shall be independent of the indemnifications provided for in this Article 17, and no claim for Loss shall be asserted under this Article 17 which is a matter which is governed by the Tax indemnifications provided for in Article 12. Payments to be made

80

pursuant to Article 12 shall not be subject to the Deductible and Cap limitations contained in this Article 17 and shall not affect the calculation of the satisfaction of such limits.

17.11 <u>Adjustments to Purchase Price</u>. Indemnity payments made pursuant to Article 17 or Article 12 shall be treated by the parties as an adjustment to the Non-Timber Purchase Price or the Timberlands Purchase Price, as applicable.

## Article 18 Transition Services; Amended Paper Sales Agreement

- 18.1 At Closing, Parent and Holdings shall enter into a Mutual Administrative Services Agreement in form and substance substantially similar to the form attached as Schedule 18.1 (the "Mutual Administrative Services Agreement").
- 18.2 At Closing Parent shall, and shall cause Boise Office Solutions to, enter into an Amended Paper Sales Agreement in form and substance substantially similar to the form attached as Schedule 18.2 (the "Amended Paper Sales Agreement"), which agreement will amend and supersede the BOS Paper Sales Agreement, and which agreement will be assigned to Holdings at the Closing.

## Article 19 Notices

19.1 <u>Notices</u>. All notices, demands and other communications given or delivered under this Agreement shall be in writing and shall be deemed to have been given (i) when delivered personally to the recipient, (ii) one business day after being sent to the recipient by reputable overnight courier service (charges prepaid), (iii) upon machine-generated acknowledgment of receipt after transmittal by facsimile if so acknowledged to have been received before 5:00 p.m. on a business day at the location of receipt and otherwise on the next following business day, provided that such notice, demand or other communication is also deposited within 24 hours thereafter with a reputable overnight courier service (charges prepaid) for delivery to the same Person, or (iv) five days after being mailed to the

recipient by certified or registered mail, return receipt requested and postage prepaid. Notices, demands and other communications to any party shall, unless another address is specified in writing, be sent to the address or facsimile number indicated below:

To Holdings: Forest Products Holdings, LLC

c/o Madison Dearborn Partners, LLC

Three First National Plaza

Suite 3800 Chicago, IL 60602

Telephone: (312) 895-1000 Fax: (312) 895-1056 Attn: Samuel M. Mencoff Thomas S. Souleles

With a copy to: Kirkland & Ellis LLP

200 East Randolph Drive Chicago, IL 60601

81

Telephone: (312) 861-2000 Fax: (312) 861-2200

Attn: William S. Kirsch, PC Jeffrey W. Richards Richard J. Campbell

To Parent: Boise Cascade Corporation

1111 West Jefferson Street Boise, Idaho 83728 Telephone (208) 384-7557 Fax: (208) 384-4912

Attn: Chairman and Chief Executive Officer

With a copy to: Boise Cascade Corporation

1111 West Jefferson Street Boise, Idaho 83728 Telephone (208) 384-7704 Fax: (208) 384-4912

Attn: Senior Vice President and General Counsel

- 19.2 <u>Amendment; Waiver</u>. Any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed, in the case of an amendment, by Parent, on behalf of all Sellers, and Holdings, or in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power, or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power, or privilege. The rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by law.
- Assignment. Except as otherwise provided herein, no party to this Agreement may assign any of its rights or obligations under this Agreement without the prior written consent of the other party hereto (not to be unreasonably withheld), except that a party may, without the consent of any other party hereto, (a) collaterally assign its rights and obligations under this Agreement to a lender as security for a loan to Holdings or any of its Subsidiaries, (b) Holdings and each of its Subsidiaries may assign its rights, but not its obligations, hereunder to one or more of its Affiliates, and (c) Holdings and its Subsidiaries may assign their rights under this Agreement to one (but not more than one) purchaser in connection with any sale of all or substantially all of the Business.
- 19.4 <u>Entire Agreement</u>. This Agreement together with all Schedules hereto, and the parties' respective Disclosure Memoranda, and the Ancillary Agreements contain the entire agreement between the parties hereto with respect to the subject matter hereof and supersede all prior agreements and understandings, oral or written, with respect to such matters, except for the obligations of the parties under the Confidentiality Agreement and the obligations of Madison Dearborn Partners, Inc., set forth in that certain letter dated January 21, 2004, to Goldman, Sachs & Co. on behalf of Parent, which obligations will remain in full force and effect; <u>provided</u> that

82

the obligations of Holdings and each of its Affiliates under the Confidentiality Agreement and such letter shall terminate as of the Closing.

- 19.5 <u>Fulfillment of Obligations</u>. Any obligation of any party to any other party under this Agreement or any of the Ancillary Agreements, which obligation is performed satisfied or fulfilled by an Affiliate of such party, shall be deemed to have been performed, satisfied, or fulfilled by such party. Without limiting the generality of the foregoing, each party agrees that Holdings may cause performance of any of its obligations hereunder by any one or more of its Subsidiaries and, to the extent that performance by such Subsidiary(ies) satisfies the obligations of Holdings hereunder, Holdings shall be deemed to have fully performed such obligation.
- 19.6 <u>Parties in Interest.</u> This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to confer upon any Person other than Sellers, Holdings, the Permitted Affiliate Purchasers, the Indemnified Parties and their respective successors or permitted assigns, any rights or remedies under or by reason of this Agreement.
- 19.7 <u>Public Disclosure</u>. Notwithstanding anything herein to the contrary, except as may be required to comply with the requirement of any applicable laws and the rules and regulations of each stock exchange upon which the securities of one of the parties (or its Affiliate) is listed, no press release or similar public announcement or communication shall, prior to the Closing, be made or caused to be made concerning the execution or performance of this Agreement unless specifically approved in advance by all parties hereto. Holdings understands and acknowledges that Parent has concluded that upon signing of this Agreement it will

be legally obligated to make a public disclosure of such fact and of the material terms of the transaction and that it has reviewed and approved Parent's proposed external and internal information releases.

- 19.8 Expenses Except as otherwise expressly provided in this Agreement, whether or not the transactions contemplated by this Agreement are consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be borne by the party incurring such expenses; provided that, notwithstanding the foregoing, each of Holdings or one or more of its Subsidiaries, on the one hand, and Parent or one or more of its Subsidiaries, on the other hand, shall each pay 50% of the fees for filings required to be made with Governmental Authorities, as described in Section 10.3.1.
- 19.9 <u>Schedules</u>. The disclosure of any matter in any Schedule shall not be deemed to constitute an admission by Parent or Holdings or otherwise imply that any such matter is material for the purposes of this Agreement.
- 19.10 <u>Bulk Transfer Laws</u>. Holdings acknowledges that Parent has not taken, and does not intend to take, any action required to comply with any applicable bulk sale or bulk transfer laws or similar laws; provided that, Parent shall indemnify Holdings for any Losses arising from such non-compliance. Such indemnification shall not be subject to the Deductible and Cap provisions of Article 17.

83

- 19.11 <u>Governing Law; Submission to Jurisdiction; Selection of Forum</u>. This Agreement shall be governed by, and construed in accordance with, the laws of the state of Delaware. Each party hereto agrees that it shall bring any action or proceeding in respect of any claim arising out of or related to this Agreement or the transaction contained in or contemplated by this Agreement, whether in tort or contract or at law or in equity, exclusively in the United States District Court for the District of Delaware or any state court located in Delaware (the "Chosen Courts") and (i) irrevocably submits to the exclusive jurisdiction of the Chosen Courts, (ii) waives any objection to laying venue in any such action or proceeding in the Chosen Courts, (iii) waives any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any party hereto, and (iv) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 19.1 of this Agreement.
- 19.12 <u>Counterparts</u>. This agreement may be executed in one or more counterparts (including by facsimile), each of which shall be deemed an original, and all of which shall constitute one and the same Agreement.
- 19.13 <u>Headings</u>. The heading references herein are for convenience purposes only, do not constitute a part of this Agreement and shall not be deemed to limit or affect any of the provisions hereof.
- 19.14 <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (i) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision; and (ii) the remainder of this Agreement and the application of such provision to other Person or circumstances shall not be affected by such invalidity unenforceability.
  - 19.15 <u>Business Day.</u> Any reference herein to a business day shall be any day other than a Saturday, Sunday or any United States federal holiday.
- 19.16 <u>Specific Performance</u>. The Sellers acknowledge that the Business is unique and recognize and affirm that in the event of a breach of this Agreement by any of the Sellers, money damages may be inadequate and Holdings and the Permitted Affiliate Purchasers may have no adequate remedy at law. Accordingly, the Sellers agree that Holdings shall have the right, in addition to any other rights and remedies existing in its favor, to enforce its rights and the obligations of the Sellers hereunder not only by an action or actions for damages but also by an action or actions for specific performance, injunctive and/or other equitable relief. If any such action is brought by Holdings or any of its Affiliates to enforce this Agreement, the Sellers hereby waive the defense that there is an adequate remedy at law.
- 19.17 <u>Independent Significance</u>. The parties intend that each representation, warranty, and covenant contained herein shall have independent significance. If any party has breached any representation, warranty, or covenant contained herein in any respect, the fact that there exists another representation, warranty, or covenant relating to the same subject matter

84

(regardless of the relative levels of specificity) which the party has not breached shall not detract from or mitigate the fact that the party is in breach of the first representation, warranty, or covenant.

19.18 <u>Confidential Information</u>. Parent shall, and shall cause each of its Affiliates to, treat and hold as confidential any information concerning the Business that is not already generally available to the public (the "Confidential Information"), refrain from using any of the Confidential Information except in connection with this Agreement and the Ancillary Agreements or disclosing such Confidential Information to any Person (other than Holdings and its Affiliates), and (except as otherwise provided herein) deliver promptly to Holdings, at the request and option of Holdings, all tangible embodiments (and all copies) of the Confidential Information which are in its or their possession or under its or their control. In the event that Parent or any of its Affiliates is requested or required (by oral question or request for information or documents in any legal proceeding, interrogatory, subpoena, civil investigative demand, or similar process) to disclose any Confidential Information, Parent shall notify Holdings promptly of the request or requirement so that Holdings may seek an appropriate protective order or waive compliance with the provisions of this Section 19.18. If, in the absence of a protective order or the receipt of a waiver hereunder, Parent or one of its Affiliates is, on the advice of counsel, compelled to disclose any Confidential Information to any tribunal or else stand liable for contempt, Parent or such Affiliate may disclose to the tribunal only so much of the Confidential Information as it is on the advice of counsel compelled to disclose; provided that such disclosing Person shall use its best efforts to obtain an order or other assurance that confidential treatment shall be accorded to such portion of the Confidential Information required to be disclosed as Holdings shall designate.

Holdings shall, and shall cause each of its Affiliates to, treat and hold as confidential any information concerning Boise Office Solutions that is not already generally available to the public (the "Parent Confidential Information"), refrain from using any of the Parent Confidential Information except in connection with this Agreement and the Ancillary Agreements or disclosing such Parent Confidential Information to any Person (other than Parent and its Affiliates), and (except as otherwise provided herein) deliver promptly to Parent, at the request and option of Parent, all tangible embodiments (and all copies) of the Confidential Information which are in its or their possession or under its or their control. In the event that Holdings or any of its Affiliates is requested or required (by oral question or request for information or documents in any legal proceeding, interrogatory, subpoena, civil investigative demand, or similar process) to disclose any Confidential Information, Holdings shall notify Parent promptly of the request or requirement so that Parent may seek an appropriate protective order or waive compliance with the provisions of this Section 19.18. If, in the absence of a protective order or the receipt of a waiver hereunder, Holdings or one of its Affiliates is, on the advice of counsel, compelled to disclose any Confidential Information to any tribunal or else stand liable for contempt, Holdings or such Affiliate may disclose to the tribunal

85

IN WITNESS WHEREOF, the parties have caused this agreement to be signed as of the date first set forth above.

# BOISE CASCADE CORPORATION

By /s/ George J. Harad

Title Chairman and Chief Executive Officer

# BOISE SOUTHERN COMPANY

By Its General Partner Boise Cascade Corporation

By /s/ George J. Harad

Title Chairman and Chief Executive Officer

# MINIDOKA PAPER COMPANY

By /s/ J. W. Holleran

Title President

# FOREST PRODUCTS HOLDINGS, L.L.C.

By: Madison Dearborn Capital Partners IV, L.P.

Its: Managing Member

By: Madison Dearborn Partners IV, L.P.

Its: General Partner

By: Madison Dearborn Partners, L.L.C.

Its: General Partner

By: /s/ Samuel M. Mencoff

Its:

# Thomson StreetEvents™

# **Conference Call Transcript**

**BCC - Boise Cascade Conference Call** 

Event Date/Time: Jul. 26. 2004 / 10:00AM ET

Event Duration: 1 hr 12 min

{W:\WDOX\SEC\175777:0003:00117634:}

Thomson StreetEvents

streetevents@thomson.com

617.603.7900

www.streetevents.com

© 2004 Thomson Financial. Republished with permission. No part of this publication may be reproduced or transmitted in any form or by any means without the prior written consent of Thomson Financial.

1

## CORPORATE PARTICIPANTS

#### **George Harad**

Boise Cascade Corporation - Chairman & CEO

#### Chris Milliken

Boise Cascade Corporation - President & CEO, Boise Office Solutions

## **Ted Crumley**

Boise Cascade Corporation - SVP & CFO

# CONFERENCE CALL PARTICIPANTS

# **Kevin Cohen**

CSFB - Analyst

# Joe Stivaletti

Goldman Sachs - Analyst

## **David Cohen**

Farallon Capital - Analyst

# **Gal Glyderman**

UBS - Analyst

# **Mark Connelly**

Credit Suisse First Boston - Analyst

## **Steve Chercover**

D.A. Davidson - Analyst

# Mark Weintraub

Buckingham Research Group - Analyst

# **Donald Lipkin**

Bear Stearns - Analyst

## Mark Hebbel

Barclays - Analyst

# **Bill Hoffman**

UBS - Analyst

# Mark Wilde

Deutsche Bank - Analyst

## Jeremy Lester

SAB Capital - Analyst

**Craig Brandon** 

Eden Bank - Analyst

#### **Brad Carlson**

Idaho Business Review - Analyst

#### **Bryan Gallot**

Gunnison - Analyst

#### Tom Yanko

Morgan Stanley - Analyst

#### Al Marino

Citigroup - Analyst

#### Peter Ruschmeier

Lehaman Brothers - Analyst

#### **Alan Hunt**

Provident Investment - Analyst

#### Noah

#### Rod Levy

Leeborg Limited Partners - Analyst

#### Era Sock

Severen Revolt Capital - Analyst

## Michael Baker

Deutcshe Bank - Analyst

## **Henry Levan**

John A. Levan and Company - Analyst

## **Nathaneal August**

Kay Capital - Analyst

#### **Noah Eclaise**

Globus capital - Analyst

# **Aaron Cocke**

Analyst

## **Theodore Crumley**

© 2004 2002 Thomson Financial. Republished with permission. No part of this publication may be reproduced or transmitted in any form or by any means without the prior written consent of Thomson Financial.

4

## PRESENTATION

# Operator

Good morning. My name is Carmen and I will be your conference facilitator today. All lines have been placed on mute to prevent any background noise. After the speakers' remarks, there will be a question and answer period. If you would like to ask a question during that time, simply press star and then the number one on your telephone keypad. Questions will be taken in the order that they are received, If you would like to withdraw your question, press the pound key. Before we begin, I note that this call will contain forward-looking statements about the planned transactions between Boise and Madison Dearborn Partners and the futures of both companies going forward. Although these statements reflect management's expectations today, they are subject to a number of risks and uncertainties. The actual outcome of the transaction may differ materially from the outcome expressed or implied in this call. For a discussion of the factors that may cause actual results to differ from the results we anticipate today, please refer to Boise's recent filings with the SEC including the current report on Form 8-K, which we filed earlier this morning. It is now my pleasure to introduce to George Harad, Chairman and CEO, Boise Cascade Corporation. Mr. Harad, you may begin your conference.

# George Harad - Boise Cascade Corporation - Chairman & CEO

Thank you and good morning everyone. We appreciate your attendance on this call today. Joining me on the call are Chris Milliken, Division President and CEO of Boise Office Solutions; and Ted Crumley, Boise's Chief Financial Officer. As those who have followed our Company know the transformation of Boise has been under way for many years and today we'd like to describe the final steps in that process. Before we start however, I'd refer you to Slide-2 on the webcast screen, which is our forward-looking statement advisory. Today's press release and copies of this presentation are posted on the Investor Relations page of our Web site. If we turn to Slide-3, as you know we announced this morning the sale of Boise's paper, forest products, and timberland assets. This transaction will complete Boise's transformation from a predominantly manufacturing-based company, which was our profile in the mid 1990s to a world-scale distribution company. By separating the company into two ongoing entities, we will successfully establish Boise Office Solutions, soon to be OfficeMax, as a strong independent company in the office products distribution business. And we'll place our paper and forest products assets in the hands of a management team focused on those businesses, and finally importantly, in the process, we believe we will realize significant value for our shareholders.

Turing to Slide four, the assets which we are selling, are being sold to affiliates of Boise Cascade LLC, a new company being formed by Madison Dearborn Partners or MDP, which is a privately held equity investment firm located in Chicago. Included in the sale are substantially all of the assets of the Boise Building Solutions,

Boise Paper Solutions, and our 2.3m acres of timberland as well as the Boise Cascade Corporation headquarters building here in Boise, Idaho. The purchase price for these assets will be approximately \$3.7b and the transaction is expected to close by mid-November of this year. Turing to slide five. Boise Cascade L.L.C or the new Boise will manage Boise's legacy paper and forest product businesses under the Boise brand, and will operate from the same corporate headquarters in Boise, Idaho. It's new Chief Executive Officer will be Tom Stephens, who is formerly the President and Chief Executive Officer of MacMillan Bloedel, and previously was the former Chairman, Chief Executive Officer and President of Manville Corporation. Boise Cascade L.L.C will hire virtually all of the former management and employees of the businesses, which we are selling, as well as corporate staff supporting these forest product businesses.

Looking on slide six, on the Boise Cascade corporation side, Boise Cascade Corporation will change its Company and trade name to OfficeMax shortly after the close. It will continue to operate its office product distribution business as its principle business. This business had annualized sales in the first half of 2004 of about \$8.6b. OfficeMax, which consists of the contract business of Boise Office Solution in retail operation of OfficeMax will serve all customer segments through all channels in the US, Canada, Mexico, Australia and New Zealand. When the transaction is complete, OfficeMax will trade on the New York Stock Exchange under the ticker symbol OMX, and its corporate headquarters will be in Itasca, Illinois. At the close of the transaction, I will become Executive Chairman of the Board of OfficeMax. Chris Milliken, who is currently the Division President and CEO of Boise Office Solutions, will be elected President and Chief Executive Officer of OfficeMax. Turning to slide seven, we'll go into some of the financial overview of the approximately \$3.7b in gross proceeds, Boise expects to realize \$3.1b to \$3.2b in the cash proceeds. That would be after our reinvestment of approximately \$175m in Boise Cascade L.L.C. Our retained interest in Timberland installment note and certain other transaction related settlement. On slide eight, we show you the use of proceeds. We currently expect to use approximately \$2.2b to \$2.3b of the net proceeds to pay down debt and meet other obligation. Those payments will include funding pension requirements of about \$200m. A reduction in accounts receivable financing of about another \$200m, and debt retirements and repurchases between \$1.8b and \$1.9b. We expect to return the remaining net proceeds of approximately \$800m to \$1b to our shareholders through common and/or preferred stock buybacks, cash dividends or combinations. These plan debt retirements and equity repurchase or dividends will begin soon after the transaction closes. However, we expect the entire process is l

5

Turning to slide nine, when OfficeMax's debt reaches our near-term targets following the debt retirement it's capital structure we believe will allow to be competitive and to grow. To illustrate what we're showing you hear is Boise's June 30 2004 actual capital structure and a pro forma capital structure at the same date including the completed transaction. The pro forma structure consists of approximately \$2.2b to \$2.4b of equity, between \$200m and \$300m of book balance sheet debt and a book debt to capitalization ratio of between 8% and 11%. Non-balance sheet obligations will include about \$100m of accounts receivable securitization financing and annual lease obligations of roughly \$350m. Turning to slide ten, I would like to touch on just three other material items of the transaction. As we mentioned in the slide, which illustrates the use of proceeds, OfficeMax will invest \$175m in Boise Cascade, L.L.C. In that investment, OfficeMax will be treated in the same manner as Madison Dearborn Partners and we will have one Board seat and voting rights proportional to our interest. To maximize the cash we received from the Timberland is included in this transaction, we will utilize an installment note structure. We intent to monetize about 90% of the note, subsequent to the closing and retain a residual interest of about 10%. And finally, the paper supply agreement that has existed between Boise Paper Solutions and Boise Office Solutions will continue under the new ownership. Boise Cascade, L.L.C. will be the primary supplier of cut-size paper to OfficeMax and other details of the paper supply agreement will continue, but they are of course confidential. Under this agreement, we expect Boise Cascade, L.L.C. to sell approximately 700,000 tons of cut-size paper to OfficeMax in 2004. Now, I would like to ask Chris Milliken to touch briefly on the guidance that we have given for OfficeMax for 2004. Chris?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Good morning. I'm on slide number 11. This morning, I would like reiterate the 2004 guidance for OfficeMax that we provided on the Boise July 20 earnings conference call. We expect same-location sales growth to range between 4% and 6%. Operating income should be in the \$210m to \$240m range. And the operating margin should be between 2.4% and 2.6%. In all three of the ranges that I just mentioned, we are more likely to be at the lower end than at the higher end of the range. We continue to believe that we will achieve \$80m of synergies with perhaps some modest upside potential this year. We continue to expect overall synergies of \$130m by 2006 for OfficeMax. We anticipate integration and step-up costs to total approximately \$50m this year. And capital expenditures are likely to come in between \$170m and \$175m. We expect EBITDA to range between \$370m and \$400m. Depreciation and amortization is likely to be about \$160m. Needless to say, all of the 45,000 associates of OfficeMax are extremely excited about this transaction and we think it provides a great opportunity to focus on the office products business and provides us a foundation to go forward. And we look forward to sharing information with you on an ongoing basis. And now, back to you, George.

# George Harad - Boise Cascade Corporation - Chairman & CEO

I'll move on slide 12, and just to summarize today's announced transaction will complete Boise's transformation from predominantly a manufacturing base company to wholesale distribution company. By separating the Company into two ongoing entities, we will successfully establish Boise Office Solutions again soon to be OfficeMax, as a strong independent Company in the office products distribution center, and we placed our Paper & Forest Product assets in the hands of the management team focused on those businesses. In the process, we were realized some very significant value for our shareholders'. With that overview, we would now be pleased to take your questions.

6

# QUESTION AND ANSWER

# Operator

At this time, I would like to remind everyone in order to ask a question, please press star one on your telephone keypad. We'll pause for just a moment to compile the Q&A roster. Your first question comes Kevin Cowen (ph) with CSFB.

# Kevin Cohen - CSFB - Analyst

Good morning. Thanks and congratulations on completing this strategy review. If you can just expand upon how the company will take out the noncallable debt and the timing that would be great? Thanks.

## George Harad - Boise Cascade Corporation - Chairman & CEO

Ted, you want to comment on the process?

## **Theodore Crumley**

Yes, we haven't, obviously, fully laid out the process shift, but we have in our \$2.3b worth of debt, we have a significant portion to that debt that we can easily get to, if you look at our financial statements we'll be filing before the end of June second quarter; we have \$700m to \$800m within the short-term debt, our revolver — our revolving credit agreement — other agreements that we can pay down immediately. We also have some term debt that is coming due in 2005 and the short-term that obviously we will be able to get to quite easily. The remainder of the debt that we wish to pay down, will be looking at obviously, market conditions, the particulars of the debt. We'll probably going to the market tender for that and there obviously would be some cost to bring that debt indicated us down to the capitalization levels that we've shown here in the presentation.

## Kevin Cohen - CSFB - Analyst

Great. Thanks a lot. I appreciate it.

## Operator

Your next question comes from the line of Mark Connelly with CSFB.

## Mark Connelly - Credit Suisse First Boston - Analyst

Thank you. Just a couple of quick things. First, will there be any compensation program changes that OfficeMax now that this thing is all worked out or is everything that's have been put in to place staying? Second, the equity investment, can you give us some flavor as to why you are maintaining that equity stake, I know we've seen Madison Dearborn do that before? And the last question is with respect to the underfunded pension, can you tell us what's happening with pension plan? I see the \$200m, I'm wondering about the rest of it?

## **Theodore Crumley**

Sure. Let me take it in reverse order, if I could. With respect to the pension plan, Madison Dearborn or Boise Cascade, L.L.C., new to for me to say that. We'll take virtually all of the current and active employees in the businesses they are buying, and for the salaried employees, we will transfer both the pension liabilities for those active employees, and assets roughly equivalent to that liabilities. So that the pension plan for the active employees will be fully funded on the closing. On the OfficeMax side, we will have an underfunded plan on the order of about \$150m to \$160m, but remember that OfficeMax froze its defined benefit plan at the end of 2004. And so it is no longer accruing new liabilities under that plan. What was the second part of the three parts, Mark?

## Mark Connelly - Credit Suisse First Boston - Analyst

The equity investments?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Yes. Well, we are making an equity investment first of all because we think it will be a good investment. We think that as many of you do that the paper cycle is just beginning to turn up and we are looking forward to seeing some return on what will obviously be a leveraged investment. What I would also tell you that we have an interest in a smooth transition here because Boise Cascade, L.L.C. will be a major supplier to OfficeMax, and that relationship is very important to both the companies. Finally, with respect to the compensation benefits programs on OfficeMax. Chris?

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

We don't see any change in the compensation at this time.

## Mark Connelly - Credit Suisse First Boston - Analyst

Perfect. Thank you very much.

7

## Operator

Your next question comes from Joe Stivaletti with Goldman Sachs.

# Joe Stivaletti - Goldman Sachs - - Analyst

Yes. Hi. I just wanted to follow-up one question on the debt side. You show that you are going to leave \$200m to \$300m of debt outstanding after the fact. Is there other specific features of your debt structure that you are looking to leave outstanding that would make up that amount?

## **Theodore Crumley**

Yes. This is Ted again. Yes, I think as we look through our entire portfolio, there are some pieces of debt that obviously either because of their nature and also because of the coupon on them. It is unlikely that we would bring those in and as you know from past experience that in a tender we may not be able to attract all of any particular issue back in. So, that's an estimate not of so much of specific debt as where we would like to see this capital structure end up at the end of the day, and it will be a mix of pieces of the debt portfolio.

# George Harad - Boise Cascade Corporation - Chairman & CEO

Yes, I would add without making a forecast that if you look at the guidance we have given, we would expect OfficeMax in 2004 to be cash flow positive. If we hit our synergy targets and integration costs keep continue to come down it will become even more cash flow positive next year, and so we don't think that that debt level of debt would impede in any way OfficeMax's ability to manage its affairs and to reinvest for growth.

# Joe Stivaletti - Goldman Sachs - - Analyst

So, the — there is no specific line item, no specific bonds, or revenue bond, or particular coupon and particular issue that you are planning to leave outstanding, you know that you just say we don't bother to take that out. You think it is a case that margin ends will remain outstanding and that will make up the \$200m to \$300m.

Yes. It is potentially both, I wouldn't identify at this time any particular issue that where we would leave the outstanding.

## Joe Stivaletti - Goldman Sachs - - Analyst

I guess, just finally on that. I mean the bonds you issued back in October, is it your interpretation under the covenant that you would have to look to take all those bonds out as a part of this transaction?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Yes. They would — obviously the sheer size of lows and those bonds were issued with the understanding among ourselves that we had this objective to deliver fairly quickly. So, obviously those bonds would be ones that would be likely in the portfolio.

#### Joe Stivaletti - Goldman Sachs - - Analyst

You mean likely it will be taken out?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Should be, — should be taken out. Yes.

## Joe Stivaletti - Goldman Sachs - - Analyst

Okay. Thanks a lot.

## Operator

Your next question comes from the line of Gal Glyderman (ph) with UBS.

## Gal Glyderman - UBS - Analyst

Hi. Thank you. Just a couple of quick questions. George, could you please explain exactly what your role will be as the Executive Chairman?

#### George Harad - Boise Cascade Corporation - Chairman & CEO

Yes. Well, first of all I'll obviously Chair Board meetings and oversee the governance process of the Company. Chris will be the CEO, and we'll run the Company, and I will assist in the transition, as we deal with the residual issues from the sale of the assets and put in place the full corporate organization that we are going to need to have, which — most of which exists in Office Max today, but some parts of which will have to be supplemented.

8

## Gal Glyderman - UBS - Analyst

Okay. Is there a set time frame for that or?

# George Harad - Boise Cascade Corporation - Chairman & CEO

That's not been established yet. But it will be a transitional role.

## Gal Glyderman - UBS - Analyst

Okay. Are there any requirements in terms of regulatory approval etcetera, that could push back the closure date in November or what type of approvals do you need?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, without tying to give a legal opinion here, we obviously would need to have antitrust approvals both here and in a couple of overseas countries like Brazil. But, we don't see any issues with respect to that, and there are normal closing conditions. But again we think the risk of any of those intruding is relatively small.

# Gal Glyderman - UBS - Analyst

Okay. Thank you very much.

# Operator

Your next question comes from the line of Mark Weintraub with Buckingham Research.

## Mark Weintraub - Buckingham Research Group - Analyst

Thank you. Could you just go through any tax implications, on the way that you have structured the deal?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Ted?

# **Theodore Crumley**

Yes. It's obviously the — this is all preliminary estimates. But if we look through the assets that we are selling, obviously, the basis in our timberlands, from a tax standpoint, actually on a book standpoint is quite low relative to today's market values. So, that's one thing that we have concentrated on. As George touched on, we would plan to use the installment note structure, with those timberlands in order to defer taxes. On the rest of the assets, there are obviously some assets that there will be tax gains on. At this point in time, we still have an annual carry forward that we have to utilize. So our objective at the end of the day is, that the actual cash taxes from the transaction would be relatively modest, on the front end of the transaction, and it will give — If you look at our balance sheet, and see the deferred tax benefits and liabilities on the balance sheet. And as I said, we also have well over \$100m worth of NOLs depending on the exact timing of the closure.

## Mark Weintraub - Buckingham Research Group - Analyst

So, can you give us any initial estimate on how much a deferred tax increase there, might be related to the transaction?

## **Theodore Crumley**

I haven't worked through all of the entries, because obviously, it's not just the transaction, but it's — it will impact the historical carrying assets and liabilities, as far as taxes are concerned on our assets, and so we will have to — we will be working through those estimates in the coming weeks.

## Mark Weintraub - Buckingham Research Group - Analyst

Okay. And just separately, I believe, you had about \$180m of cash at the end of the last quarter?

#### Theodore Crumley

That's correct.

# Mark Weintraub - Buckingham Research Group - Analyst

Is that still a good cash number to be using or is some — is that going to change in the midst of these transactions?

# **Theodore Crumley**

I believe the on going cash requirements of the OfficeMax business will be lower than that. That was the case we were carrying at the end of June that was necessary to operate the entire company including the forest products side. So, I would expect that to come down. I think, in January — we talked about the OfficeMax, Office Solutions business and you know a rough estimate would be that — \$125m or so would probably be a good operating cash number.

9

# Mark Connelly - Credit Suisse First Boston - Analyst

Okay. And then lastly, Chris you reaffirmed the OfficeMax guidance for '04. Does that include corporate? Or is — are there going to be corporate allocations that aren't captured on that slide?

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

There will be some additional corporate expense that will be put into OfficeMax or let's put it this way, reside in OfficeMax once the Forest Products assets are sold off. Our best guess at this point is it's about \$10m.

# Mark Connelly - Credit Suisse First Boston - Analyst

Okay.

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Annually. On an annual basis.

## Mark Connelly - Credit Suisse First Boston - Analyst

Great. Thanks very much.

# Operator

Your next question comes from the line of Steve Chercover with D.A. Davidson.

# Steve Chercover - D.A. Davidson - Analyst

Good morning. Yes, my questions are also with respect to tax, and I guess they have been answered. But, were there any write-downs or would they've already been disclosed in this?

## **Theodore Crumley**

We will take care of any issues; we'll obviously be going through all the assets making allocations, which would be necessary both from a book and a tax standpoint. And as there are obviously pluses and minus in our entire portfolio. So, that will all be taken care of in the transactions if there are any write-downs that are required.

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

And I think, it's fair to say at this point and again we are estimating some things depending on when the business closes and so on — when the transaction closes. But at this point, on the initial transaction, we would expect to be around breakeven to small profit and then we would expect to incur some expenses as we brought some of the debt back. And — so, overall we are thinking of this as a kind of a breakeven transaction from a book basis.

#### Steve Chercover - D.A. Davidson - Analyst

Pretty fair to say to say — I recognize that it's not carved in stone, but the 3.1 to 3.2 there would be no share or anything left over?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

I don't know what you mean by —

## Steve Chercover - D.A. Davidson - Analyst

Sorry. That wasn't a very articulate way of putting it, but it's going to be quite clean. One other question. Is it possible to give some sort of provisional ex-dividend date if you chose to, in fact, distribute the cash in that fashion?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

No. I think it's premature to do that because again, we are not even sure at this point what the exact closing date will be. Obviously, we would like to move this along as quickly as possible, but there are some long lead-time items. We are selling 2.3m acres of timberland, and we need to get deeds and title work done on all of that property.

#### Steve Chercover - D.A. Davidson - Analyst

Okay, final question. It sounds like the ongoing Boise LLC will be intact, but if Madison Dearborn were to decide at some stage to start to monetize timberland or sell off various assets, is that their prerogative? And do you some how benefit through your equity ownership?

# **Theodore Crumley**

Well. We would have the same results that MDP would have. Although obviously, our share of the company would be smaller. And — you know in terms of what their plan is — their initial plan is to operate the business, whether they make other decisions in the future, you kind of, have to talk to them, once they got — once the

10

transaction is closed and they've got their management structure in place.

#### Steve Chercover - D.A. Davidson - Analyst

Okay thanks. Congratulations.

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Thank you.

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Your next question comes from the line of David Cohen with Farallon Capital.

# David Cohen - Farallon Capital - Analyst

Good morning. First of all, George, really I just wanted to thank you and congratulate you. It was an amazing transformation of the company, very bold decision, I think obviously, very right and hats off to you for taking and making these tough decisions.

## George Harad - Boise Cascade Corporation - Chairman & CEO

Thank you, David.

## David Cohen - Farallon Capital - Analyst

And second question was, I assume you won't comment on what percent of the equity of LLC will own or any other capital structure that I'll see?

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Well, actually our \$175m will be divided into two pieces, and the first piece would be about a \$110m of common equity, which would represent just a little under 20% of the Boise Cascade, L.L.C., and we will then have about \$65m in a Class A stock which would have a pegged dividend but certain rights, priority rights and distributions.

# David Cohen - Farallon Capital - Analyst

Right. And then, secondly, if you could talk a little about building products and what happens with the building products division, it's not really mentioned?

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

No, the building solutions business also is being acquired by BC, L.L.C. So, it's all of the paper and forest products side of the business.

# David Cohen - Farallon Capital - Analyst

And that includes the distribution side of building as well?

# Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Yes, it does.

## David Cohen - Farallon Capital - Analyst

And, I mean lastly I wondered if you could just comment on — you made a comment that its,`we are selling the business, but yet it's quite an attractive piece to earn equity.` This is a sort of contradictory but not necessarily, if you could just reconcile it, how you are thinking about it?

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Well, first of all, we think that the strategic clarity that this brings to the company is, it creates some value and the direction we want to take the ongoing company. But I think that, given the leverage on this and the timing of it in the upcycle, if there is some return to be gained, we want that to be available to our shareholders. So, we thought carrying a small residual investment in the company would be a good way to assure that we can capture some of that upside. And as I said the relationships between these two companies will still be quite significant until this provides us with an opportunity to have some ongoing dialogue with a major supplier.

## David Cohen - Farallon Capital - Analyst

Excellent. Thank you and really congratulations to you and the Board and everyone in building, incredibly bold decision, I think the right one.

11

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

Thank you.

## Operator

Your next question comes from the line of Donald Lipkin with Bear Stearns.

## Donald Lipkin - Bear Stearns - Analyst

Good morning. I'm concerned with the tax exempt debt, that the company has outstanding there, maybe 13 or more series of tax exempted that you either guarantee or issued for you and I was wondering, are you retiring all that debt?

## **Theodore Crumley**

Well, there is a — if you are referring to — we have some industrial revenue bond with debt with certain of the facilities. Again, we'll just look at that as part of the portfolio, some of that debt, we will be able to retire, I think very effectively and some of that we will look at and it depends on the restrictions around the individual debt pieces and the cost of bringing that in. So, as I said before I haven't identified specifically the pieces of debt that were likely to be left in OfficeMax's structure at the end of the day. But we will try to work through the repayment of that debt as efficiently as possible.

# Donald Lipkin - Bear Stearns - Analyst

But it wasn't either retired or deceased, it would stay with OfficeMax it wouldn't go with the leverage density?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Yes, that is correct.

# Donald Lipkin - Bear Stearns - Analyst

Okay, I just wanted to make sure of that. Great, thank you very much.

## Operator

Your next question comes from the line of Mark Hebbel (ph) with Barclays.

# Mark Hebbel - Barclays - Analyst

Yes, very good morning, and again congratulations also. I was wondering if you had a chance to speak with the agencies about this event today and what your long-term rating goals are?

## **Theodore Crumley**

We are — this is Ted again. We obviously when we began this process and with the completion of the OfficeMax acquisition a year ago or less than a year ago, we spent time with the agency explaining the long-term objectives of the Company, explaining the long-term capitalization that we would like to see it have. So, they are well aware of where we are headed and what our goals are. Obviously, with the announcement of the transaction this morning, our Treasurer is speaking with them to reinforce where we are headed and some more specifics on how we will get there. But you can see from the capitalization that we put in the slide, that what we are trying is to make certain that OfficeMax, if you will comes out of the box with strong capital structure they can compete with. We understand they have a significant amount of their facilities that are leased and we've also obviously shared that information with the agency. We would not expect — although they would always be hopeful we do not expect day 1for it to be an investment grade entity, but I think in very near term given the growth in the entity and the strong cash flows and the very modest term debt that we plan to have in its structure, so that's definitely a goal for us.

# Mark Hebbel - Barclays - Analyst

Okay. Thank you.

## Operator

Your next question comes from the line of Bryan Gallot with Gunnison. (ph)

#### Bryan Gallot - Gunnison - Analyst

Great. I would also like to extend my congratulations like the earlier caller. I just have a couple of quick questions for you. One is the timing and method of cash distribution I know you obviously are not going to outline the details specifically, but is it your intention to kind of do this over time or within several months of closing, regardless of whether you choose it to do a tender or dividend method?

# George Harad - Boise Cascade Corporation - Chairman & CEO

This is George. Obviously we aren't going to be able to do much of anything expect do some pre-positioning perhaps before the close. After the close, I guess the best way to say it is, we would

12

like to move as expeditiously as possible, but we certainly don't want that to cause unnecessary expense. So, if you thought of — over a period of several months following the close, that would probably be our objective, although I don't think — at this point I would like to be pinned down on those specific timeframe.

#### Bryan Gallot - Gunnison - Analyst

That makes sense. It just sounds like that would be more in large transactions as opposed to kind of an ongoing buyback, which could take time in, the market may view may or may not happen.

# George Harad - Boise Cascade Corporation - Chairman & CEO

I don't think that at this point I want to comment any further than I already did.

## Bryan Gallot - Gunnison - Analyst

Okay, great. The last question was just in response to the earlier question on pension. The pension liability for retired employees, did that stay at OfficeMax?

#### George Harad - Boise Cascade Corporation - Chairman & CEO

Yes it does.

## Bryan Gallot - Gunnison - Analyst

What would the size of that be?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Well, the pension assets and liabilities for both retired employees and active employees whose benefits have been frozen under the existing plan will result at the closing in roughly \$150m to \$160m under funded position. And I say roughly because again we are guessing at the timing and what the returns and the plan maybe between now and then.

# Bryan Gallot - Gunnison - Analyst

Okay. And that's what you outlined on page 8 as debt?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Exactly.

## Bryan Gallot - Gunnison - Analyst

Okay. Thank you.

# Operator

Your next question comes the line of Rod Levy with Leeborg Limited Partners.

## Rod Levy - Leeborg Limited Partners - Analyst

Yes, good morning. You already answered the question in part, but I guess I was looking from a shareholder perspective — as a BCC shareholder, what do you think I could expect to receive, in as much detail as you can outline, after the close of the transaction from a cash in OfficeMax shares perspective?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Well, obviously your Boise shares will convert, if that's the right word, they will become OfficeMax shares. Essentially it's the same corporation, we're simply changing the name. In terms of direct proceeds in distributions, I really can't comment on that other than the guidance we've given and say that at the end of the day, we hope to have between \$800m and \$1b to return to shareholders, whether preferred or common shareholders.

# Rod Levy - Leeborg Limited Partners - Analyst

Right, okay. Thank you very much.

## Operator

Your next question comes from the line of Nathaneal August with Kay Capital. (ph)

## Nathaneal August - Kay Capital - Analyst

Hello, good morning and congratulations. To echo few other callers, I think this is great news. I wanted to ask a few questions on this return to capital to shareholders. Have you considered a tender offer?

13

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, actually we haven't gone that far in our thinking. I think that what we will do is, as the time draws closer, we will figure out what the best way is and the best form is to return that money to shareholders in a very efficient manner.

#### Nathaneal August - Kay Capital - Analyst

Okay. And on the \$65m of Class A stock that you are going to own, are you going to own that side by side with MDP or is it that just a special class that's created just for you?

#### George Harad - Boise Cascade Corporation - Chairman & CEO

That is a special class.

## Nathaneal August - Kay Capital - Analyst

Okay, and then to understand you are going to get another \$150m of cash from the timber notes that you are going to monetize at their close, is that right?

## George Harad - Boise Cascade Corporation - Chairman & CEO

No. If I understand your question correctly, Ted you want to?

## **Theodore Crumley**

No, actually what we've shown on the schedules is the non-cash portion that we would expect on the front-end from the timber notes. In other words that's our retained interest in the timber notes. We would expect to monetize the remainder of those, as Stewart said, approximately 90%.

#### Nathaneal August - Kay Capital - Analyst

Okay, so the \$165m, is the 10% of that?

# **Theodore Crumley**

Yes, it is the retained interest.

## Nathaneal August - Kay Capital - Analyst

Okay, and the \$200m to \$300m of transaction costs, does that include debt tender costs and premiums that you'd need to pay on the debt?

# **Theodore Crumley**

No, but on that particular schedule, that \$200m to \$300m includes transaction related fees; there's some state and local taxes; excise taxes in some states, there are obviously some human resource costs in that. So that does not include the tender costs.

## Nathaneal August - Kay Capital - Analyst

Okay, and have you retained any building or forest product type assets like, I don't know, the wood and plastic mill that you own or anything like that?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Yes, there are two Building Solutions assets that are not part of the sale. One is, our architectural products business located in Elma, Washington. That business will be managed by Boise Cascade LLC under a management contract, and as those of you who followed the Company know that facility is still on the process of start-up, and as it starts up and starts to run at expected production levels, we'll want to see what alternatives we have available to us. But that would not be a long-term asset for office management. The other asset is a series of timberland parcels in Idaho, where we were in the process of monetizing those parcels as higher and better used lands, and we simply excluded them from the sales so that we'll go ahead and complete those transactions.

# Nathaneal August - Kay Capital - Analyst

And could you give me an approximate value on the HBU land?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, we've got about 6000 acres, we have bids outstanding, and we haven't gone through the process of bidding and closing all the bids. So, I think it would be premature to announce a number.

# Nathaneal August - Kay Capital - Analyst

Okay, great. Again congratulations. We think this is a terrific transaction with the Company.

Thank you.

#### Operator

Your next question comes from the line of Al Marino (ph) with Citigroup.

## Al Marino - Citigroup - Analyst

Hi. Will any existing Boise debt consumed by the acquiring LLC?

## George Harad - Boise Cascade Corporation - Chairman & CEO

If I heard your question Al, it was whether any existing Boise debt will be acquired by LLC? And the answer is no.

#### Al Marino - Citigroup - Analyst

Okay. Thank you.

# Operator

Your next question comes from the line of Peter Ruschmeier with Lehman Brothers.

## Peter Ruschmeier - Lehaman Brothers - Analyst

Thanks. Good morning. Not to be redundant, but congratulations. You've really been working on a lot of transactions. Most of my questions are answered. I was curious, is there a break-up fee and if so how much?

## George Harad - Boise Cascade Corporation - Chairman & CEO

There is a break up fee, under certain conditions related to what you would know as a fiduciary out, which is to say, if we got a competing offer and the board decides that it is superior, we would pay a fee. There is also a liquidated damages clause in certain events of MDP does not fulfill its obligations to close.

#### Peter Ruschmeier - Lehaman Brothers - Analyst

Okay. And just along the lines of a potential bid from another competitor, can you comment, George, on your thought process, as you look at this, and obviously competitors had plenty of time to step forward but, may be just elaborate on why this is going to a financial buyer versus a strategic buyer that could have arguably offered more in the way synergies?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Well, we had as we've told people over the course of this process a number of different alternatives that we evaluated and on balance we think this is the best transaction for our shareholders. It certainly has the value of a single transaction and creating certainty for both shareholders as well as other constituencies including our customers. And I think that, that influenced the decision as well as the fact that we think it is a very good valuation for the assets that we're selling.

# Peter Ruschmeier - Lehaman Brothers - Analyst

Okay, I want to come back to the \$800m to \$1b that you'll return to shareholders. I know you can't comment on specifics, but can you help us to better understand the flexibilities that you have, is it fair to assume that you could go to one extreme and dividend the whole amount out and you could go to the other extreme and do a Dutch tender or sometime of purchase of shares and that there is not any limitations that you know of to prevent you from swinging between those possible outcomes?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, at this point we think that all the alternatives are open to us and obviously what we're looking for is as we get closer to the time to figure out what the most effective way is to deliver that cash both for the shareholders receiving it and for the benefit of the shareholders continuing as investors?

## Peter Ruschmeier - Lehaman Brothers - Analyst

Okay, and lastly if I could, can you elaborate on either a selection process of Tom Stevens I guess that was an Madison Dearborn decision and whether there are other meaningful management changes anticipated or is it expected that this is going to be a fairly seemless transition?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Well, the selection of Tom Stevens obviously was Madison Dearborn's choice and you should talk with them about that. We obviously think that Tom is very experienced executive in the business and will do a good job. We do expect that the majority of our management will be hired by L.L.C. and will continue on in roles, but obviously individuals will be able to make their own choices, as we go forward, and at this point we really can't announce exactly what the management structure will look like.

15

# Peter Ruschmeier - Lehaman Brothers - Analyst

Is it possible to comment on whether there are any types of parachutes, so to speak, that would make it — provide an extra layer cost that could be out there?

Well, there are retention agreements that we will be entering into with executives and both at the officer level and the key executive level. That are also included in that.

## **Theodore Crumley**

These are franchise fee taxes.

# George Harad - Boise Cascade Corporation - Chairman & CEO

And those are the three major buyers (ph).

#### Peter Ruschmeier - Lehaman Brothers - Analyst

Thank you.

#### Operator

Your next question comes from the line of Bill Hoffman with UBS.

#### George Harad - Boise Cascade Corporation - Chairman & CEO

Hello, Bill.

## Operator

Mr. Hoffman your line is open. He is not responding. We will go to the next question, sir. And that is from the line of Jeremy Lester (ph) with SAC Capital.

## Jeremy Lester - SAB Capital - Analyst

Yes, that's actually SAB Capital. Hi, guys. Just a few questions. On the tax side, what you've said is not supposed to be significant, is that kind of under a few \$100m or is that under \$50m, just kind of in the realm of what sort of expectations are there?

# **Theodore Crumley**

Yes. This is Ted. When I talk about the tax and talking about the initial cash cost of the transaction, and we see that is being more or like the second — under a hundred million and, remember that we also — as I said we have well over \$100m of NOLs, we also won an A&G positions, so obviously, there will be some cash tax due on the transaction, we believe but we — it's barely minimal compared to the size of the transaction.

# Jeremy Lester - SAB Capital - Analyst

Do you think net of the NOL it's under \$100m, is that a fair statement?

## **Theodore Crumley**

That's probably in the range, I don't have it, actually I have calculation that, is that accurate at this point of time, but I think that's a fair estimate.

# Jeremy Lester - SAB Capital - Analyst

Okay, and then on the pension side, sorry, I know you talked about this, but, the numbers that are out there, you put out there for OfficeMax funds, OfficeMax trying about \$150m, \$160m. I know you are going to be funding \$200m into it. The unfunded amount though as of the end of the year was, it was like \$460m—?

## **Theodore Crumley**

Right.

# Jeremy Lester - SAB Capital - Analyst

So what happens to that other \$100m, I guess—?

## **Theodore Crumley**

Well, let me give you just the little more of the detail on there. You are correct. The difference between the projected benefit obligation at the end of the year and the assets was approximately \$460m, sure above. We have been making contributions during the first part of the year 2004, and obviously, we've been paying benefits. In the George's point, we've made some changes in the plan on a go forward basis. And so, if you take that year-end number adjusted for results of the first half of the year, we are making a contribution in this plan of approximately \$200m and basically the time that the deal is consummated the liability will be on the ABO basis, accumulated basis in the post projected basis and the resulting approval of assets against that adjusted for the

16

amounts that are transferred to the buyer will leave us with the \$150m to \$160m among the underfunded position. The point on that is though that there are adequate assets in the plan and the benefit that accruals are frozen and so the plan assets should definitely have the ability to earn an adequate return to cover that underfunding.

# Jeremy Lester - SAB Capital - Analyst

Well, that the new — I'm sorry, say that question again?

#### Jeremy Lester - SAB Capital - Analyst

Yes. The other benefits, I would assume is mostly health care constants going forward that was \$110m underfunded by the end of '03?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Yes, I think some of that is actually accrued liabilities for active employees, it will be transferred to the Boise Cascade, L.L.C.

#### Jeremy Lester - SAB Capital - Analyst

So there will be a separation?

# George Harad - Boise Cascade Corporation - Chairman & CEO

And there will be a separation of that liability.

## Chris Milliken - Boise Cascade Corporation - President & CEO, Boise Office Solutions

So, the amount remaining on OfficeMax will be less?

## Jeremy Lester - SAB Capital - Analyst

Some portion of that \$110m.

## George Harad - Boise Cascade Corporation - Chairman & CEO

Yes. That's correct. They will be in lieu of long-term incentives that the company normally would grant at this time. On the Office Solutions, OfficeMax side of the business, there will also be a long-term incentive offered to the employees, but it will be of a more traditional sort because that group of employees is in an ongoing and stable situation.

#### Alan Hunt - Provident Investment - Analyst

Okay. Thanks and congratulations.

#### George Harad - Boise Cascade Corporation - Chairman & CEO

Thanks.

# Operator

Your next question comes from Mark Wilde with Deutsche Bank.

## Mark Wilde - Deutsche Bank - Analyst

Good morning and again congratulations. I wanted to talk a little bit about the timber installment notes. Can you first of all give us any sense of tenure on them?

# **Theodore Crumley**

Yes. This is Ted again. I wouldn't give you the specifics, because we have not finalized those nor offered them obviously at this point. Installment note — timber installment notes that have been done in the past have been typically 10 to 15 year tenure. And so I would see that what we will do — would accomplish would be very much in line with what has been happening in the past.

# Mark Wilde - Deutsche Bank - Analyst

Okay. And Ted just kind of backing into from your 90% number and then that \$165m, is it fair to assume that Timberland and the deal is valued at somewhere around \$1.8b?

## **Theodore Crumley**

No, 10% divided into \$165m is 1.65, but that number is also I think misleading Mark in a sense that you have to view this as an overall evaluation and obviously Madison Dearborn and ourselves were looking for opportunities to benefit in terms of allocation of value on their side and tax minimization on our side.

17

## Mark Wilde - Deutsche Bank - Analyst

Okay. And just a follow-up on that tax minimization George, if Madison Dearborn monetizes some of this Timberland over the next couple of years whether it's in a straight sale or maybe they are looking at an alternative ownership structure, can you talk about what tax consequences that you may have for OMX?

# **Theodore Crumley**

Yes. This is Ted again. If you think about Boise's or OfficeMax's retained ownership — that retained ownership is in the Timberland notes not in the Timber itself. The Madison Dearborn, Boise Dual (ph) L.L.C., actually owns the Timberland.

In other way we believe that post-closing and post-monetization of the notes, actions by MDP will not affect the tax status of the notes.

#### Mark Wilde - Deutsche Bank - Analyst

That's the key question. Okay. Thank you very much.

## Operator

Your next question comes from the line of Alan Hunt (ph) with Provident Investment.

#### Alan Hunt - Provident Investment - Analyst

Hi, this is Alan. I had a question on the transaction-related issues and you've talked about that briefly before, but could you give a little more detail? It seems like kind of a large number. Do you have an approximate break down of what the fees and HR costs are?

#### **Theodore Crumley**

This is Ted again. I would just not go through all the pieces of that I touched on the pieces, but not the quantification. Their fees in the transaction, I think you will find when we finish this transaction. They are very typical for a transaction of this size, but obviously there are certain primarily HR costs that because of the separation of the employees that those will in effect be have to be settled after closing. And then also as I said there are some fairly significant taxes at the local level primarily on the transfer of Timberland — of real property.

## Alan Hunt - Provident Investment - Analyst

Okay, great. Thanks a lot. Congratulations.

#### Operator

Your next question comes from the line of Craig Brandon (ph), with Eden Bank (ph).

## Craig Brandon - Eden Bank - Analyst

Hi. My questions have already been answered. Thanks.

## Operator

Your next question comes from the line of Brad Carlson with Idaho Business Review.

# Brad Carlson - Idaho Business Review - Analyst

Thank you. At the headquarters, how much has employment changed recently, and how much will it change as the transition is completed. Also, can you give any details about any severance packages and the cost of those? Thank you.

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, let me try the first part of that. I don't think that we've had anything other than normal turnover over the course of the last year. As we've looked at these alternatives, I think that is frankly a credit to our employees and their loyalty to the Company and dedication to the Company. On the close of this transaction, Boise Cascade LLC expects to hire virtually all of the employees. I think there may be some few who will transfer to, or stay with, most precisely said, OfficeMax and some who will be here only for the transition and then their positions will be eliminated but I think that will be a relatively small number. And in terms of the cost of all that, I think it is hard to predict until we know exactly what the numbers would be, but we will be talking with employees today about what severance policies will be in place for employees whose jobs might be affected by the transaction.

## Brad Carlson - Idaho Business Review - Analyst

Thank you.

18

## Operator

Your next question comes from the line of Noah Eclaise (ph) with Globus capital.

# Noah Eclaise - Globus capital - Analyst

Can you guys talk about what the corporate expense will look like for OfficeMax a year or two out, not having to have the Boise headquarters on its P&L. Thanks.

## George Harad - Boise Cascade Corporation - Chairman & CEO

Yes, Noah, I am not sure what it would be a year or two out, but I think we said earlier that if, some costs are already allocated to the segment. So that if you took the segment expense and added roughly \$10m plus or minus on an annual basis, we think that is a pretty good estimate of the incremental cost that this segment would bear when it converts to being BoiseMax Corporation. Well, excuse me, OfficeMax Corporation.

# Noah Eclaise - Globus capital - Analyst

Thanks.

#### Operator

Your next question comes from the line of Michael Baker with Deutsche Bank.

#### Michael Baker - Deutcshe Bank - Analyst

Hi thanks. I was looking at it from the retails perspective on the OfficeMax side. Does this transaction change your outlook at all with respect to the ultimate growth opportunity in the retail business, number of stores that are said, out to open and are there any more stores to close?

## **Theodore Crumley**

No, I don't think it changes the outlook at all. In fact, I think it strengthens the outlook, and we don't anticipate any additional closures other than our normal planning cycle that we have in place right now.

## Michael Baker - Deutcshe Bank - Analyst

So when you say it strengthened the outlook, the potential to reaccelerate the number of stores that you opened, is that a fair statement?

#### Theodore Crumley

Yes.

# Michael Baker - Deutcshe Bank - Analyst

Okay. We talked about what your square footage percent maybe in terms of growth over the next couple of years or is that still—?

## **Theodore Crumley**

We haven't decided that yet. That is out for debate as we talk right now.

#### Michael Baker - Deutcshe Bank - Analyst

Okay, and thanks, and if I could one more, just a quick question. Is this offering profit in EBITDA numbers that you outlined earlier in the presentation, are those before or after the expected synergies integration cost and the pamlink (ph) corporate overhead addition that you spoke about?

## George Harad - Boise Cascade Corporation - Chairman & CEO

The numbers that Chris talked about, included the synergies and the integrations that we have described previously for the business on its own, but it did not include any additional corporate overhead costs.

# Michael Baker - Deutcshe Bank - Analyst

Thanks, thanks a lot.

# Operator

Your next question comes from the line of Tom Yanko (ph) with Morgan Stanley.

# Tom Yanko - Morgan Stanley - Analyst

Hi guys, most of my questions have been answered. I was wondering if you could tell me what the book tax rate on OfficeMax, is going to be, going forward?

19

# **Theodore Crumley**

This is Ted again. I don't have a numbers exactly, but given their mix of foreign and domestic, I think that rate that we are using currently in the 35% to 36% rate, is probably still a good estimate.

# Tom Yanko - Morgan Stanley - Analyst

Okay, and on the Boise LLC capitalization, I think you said that your \$110m investment would be a little less than 20% of the pro forma comment?

# **Theodore Crumley**

Correct.

# Tom Yanko - Morgan Stanley - Analyst

And, that would imply that there is going to be about 15% common equity in that deal, which to me looks a little skinny and I was wondering if you can give us a little detail on the pro forma capitalization of the LLC?

# **Theodore Crumley**

Well, I'd really refer that question to Madison Dearborn. I think that they obviously are working hard on the capital structure. But remember that in addition to the common equity, there would be our Class-A equity.

Okay.
Theodore Crumley
Which would change the ratio a little bit from the number, you just calculated.
Tom Yanko - Morgan Stanley - Analyst
Okay, and what would you tax basis be, in your equity investment in the LLC?
Theodore Crumley
It would be a 100% of the investment, base value, to start certainly.
Tom Yanko - Morgan Stanley - Analyst
Okay.
Theodore Crumley
Because, it's a reinvestment. We get the cash first and put the money back in.
Tom Yanko - Morgan Stanley - Analyst
I understand. I didn't know if you somehow transferred some basis, send some other assets into the equity investment, which would give you the lower tax basis?
Theodore Crumley
No, but let me clarify, obviously, I mean, getting in the technical and the tax, but, that you go through the transaction, the gains and loss on the transactional basis and then there is the deferral of any gain or loss on the portion that we retain in ownership, that has — but I think face value is the way to look at the investment base in those assets.
Tom Yanko - Morgan Stanley - Analyst
Great, thank you.
Operator
The next question comes from the line of Era Sock (ph) with Severan Revolt Capital (ph) .
Era Sock - Severen Revolt Capital - Analyst
Hello, just a question regarding your \$1.8b to \$1.9b of debt retirement and repurchases. Is that inclusive of totalities and premiums, or is that just the principle amount of debt?
George Harad - Boise Cascade Corporation - Chairman & CEO
That is basically the principle amount of debt.
Era Sock - Severen Revolt Capital - Analyst
Okay, thank you.
20
Operator
Your next question comes from Henry Levan (ph) with John. A Levan (ph) and Company.  Henry Levan John A Levan and Company. Analyst
Henry Levan - John A. Levan and Company - Analyst  Thank you I just some online in middle of the pension question. I want to just confirm that in fact (ph) there is some of your obligations, which are being assumed.
Thank you. I just came online, in middle of the pension question. I want to just confirm that in fact (ph) there is some of your obligations, which are being assumed, attention wise then you will left with roughly a \$160m of cumulative benefit obligation, in which and that's after the repayment of the \$200m or so, and you say, have asset — sound like they're assets sufficient to cover that. Is that correct?
Theodore Crumley
No, let me —

Tom Yanko - Morgan Stanley - Analyst

Henry Levan - John A. Levan and Company - Analyst

I apologize.

**Theodore Crumley** 

No that's fine. I have to make to you (ph) exactly clear. The 150 to a 160, is the under funded position. There will be obviously be more, there will be an ABL liability and their assets, that will be retained by OfficeMax, the 150 to 160 as an under funded, but, I said with a frozen plan, we feel comfortable at the plan assets that give the ability to earn, to cover the cumulative benefit obligation.

## Henry Levan - John A. Levan and Company - Analyst

I'm — to the — you are paying down — in the slide show there is page eight, there is \$200m of pension funding that's got to be taken care of?

## **Theodore Crumley**

Right. In other words what I was reconciling a little earlier before you came on was the — at the end of the year — last year we had approximately a little over \$1.7b projected benefit obligation for all of the employees and we had \$1.273b of assets. During these transactions we made additional contributions to the plan that we talked about, we will separate the active employees, the assets and liabilities that move with them. At the end of the day we will retain unaccumulated benefit obligation for retirees and the remaining employees with OfficeMax and the assets to go with that and the difference we expect, we're estimating will be about a \$150m to \$160m of underfunding at the end of the transaction.

## Henry Levan - John A. Levan and Company - Analyst

OfficeMax as all expect is only on in the [Inaudible] basis?

## **Theodore Crumley**

Yes, correct.

# Henry Levan - John A. Levan and Company - Analyst

Okay, so that's it. After you pay down there is a \$150m \$160m left?

# **Theodore Crumley**

Correct.

## Henry Levan - John A. Levan and Company - Analyst

Okay. Thanks a lot.

#### Operator

You have a follow up question from the line of David Cohen with Farallon Capital.

## David Cohen - Farallon Capital - Analyst

Yes, just two quick clarifications. In the press release when you talk of OfficeMax debt being \$250m to \$300m, I assume that is not net of the \$125m of cash you mentioned, Ted?

# Theodore Crumley

Yes, that's correct. It is not net.

## David Cohen - Farallon Capital - Analyst

It is not net?

# **Theodore Crumley**

Right.

# David Cohen - Farallon Capital - Analyst

And so it's actual debt of \$250m to \$300m, so net debt maybe half that amount?

21

# **Theodore Crumley**

It could be although I — you are exactly right. That's the way from a credit standpoint to look at it, from my standpoint on an operating basis, we do need a \$125m or so of cash to operate, that's correct.

## David Cohen - Farallon Capital - Analyst

Got it, And then the second question, just wanted to clarify, on the installment note, if the \$165m is up 10% interest, is there a big tax liability on the full amount that comes with OfficeMax that's due in the 10 to 15 years that goes or is that not going to be an issue?

# **Theodore Crumley**

Well there is a significant issue — there will be a significant for tax liability. There is a tax gain obviously, once the notes are settled and so there will be a significant deferred tax liability that will be in the 10 to 15 year range before it's due.

## David Cohen - Farallon Capital - Analyst

Thank you very much.

## Operator

You have a follow up question from the line of Donald Lipkin with Bear Stearns.

## Donald Lipkin - Bear Stearns - Analyst

Hi. What do you expect your — what happened to your ratings after the transaction?

#### **Theodore Crumley**

Well, again we've just — are in the process of informing the rating agencies of this transaction. As Ted said earlier we had talked to them at the time we entered into the OfficeMax transaction and told them that our plan was to deliver quickly. We don't expect although we would like to see. We don't expect an upward revaluation of our ratings initially, but we do think that this is significant delivering and that within some reasonable period of time we'd like to get back to an investment grade credit.

#### Donald Lipkin - Bear Stearns - Analyst

Okay. Thanks a lot.

# Operator

You have a follow up question from the line of Kevin Cohen with CSFB.

# Kevin Cohen - CSFB - Analyst

I just wanted to follow up, in regard to the newer bonds that were issued back in October the notes due in 2010 and 2013, does this transaction fall under the asset sale test, which would permit the company to call in those two bonds at par as opposed to a premium or do bond holders have the option put those bonds at par behind but not to the company or is it just not applicable? Thanks.

## **Theodore Crumley**

No, you are correct, I mean this is the asset sale provisions of the bond. This will fall under those.

# Kevin Cohen - CSFB - Analyst

Okay. Thanks.

# Operator

You have a follow up question from Mark Hebbel with Barclays.

## Mark Hebbel - Barclays - Analyst

It's been answered, thank you.

## Operator

You have a follow up question from Mark Weintraub with Buckingham Research.

# Mark Weintraub - Buckingham Research Group - Analyst

Thank you, a small detail — but you typically allocate pension expense to the various segments. Can you tell us how — were you allocating any pension expense to the OfficeMax what you are calling in the OfficeMax slide?

# **Theodore Crumley**

Yes, this is Ted again, yes the OfficeMax because of the historical benefit of the pension to their employees was incurring a FAS 87, if you will, expense. We would expect with the rebalancing, if you will, of the pension plan and the freezing of pension benefits going

22

forward, that these rebalanced plans will mean that the FAS 87 expense at office products on a go forward basis would be very minimal, if not zero.

## Mark Weintraub - Buckingham Research Group - Analyst

Okay and is there anything included in the — that \$210m, \$240m guidance you give for 04'?

## **Theodore Crumley**

In terms of the change in the ongoing pension expense, the answer is no we didn't build anything into that.

# Mark Weintraub - Buckingham Research Group - Analyst

Thank you.

#### Operator

Your next question comes from the line of Aaron Cocke (ph), private investor.

#### Aaron Cocke Analyst

Yes, thank you for taking my call. I was just curious about the thought process on the timberland, seems to me that one option that you obviously would have to consider was monetizing the timberland and having a timber supply agreement with your converting facilities. Can you just give me some brief knowledge as to why you chose one option over the other?

## **Theodore Crumley**

Well, I don't think we want to get into what all the options were in detail, but think that clearly doing a single transaction makes the issue of arranging timber supply agreements and transfer agreements and support agreements moot. And from our standpoint that was an advantage of a single buyer.

#### Aaron Cocke Analyst

Just one brief follow up. So, I guess the question is in terms of that transaction, you don't feel like if you would have been able to get an increased price per acre on the deals in relation to doing it as a single transaction, for example if you monetize the timberland individually and you got to say it 900 to a 1000 an acre on average, you don't think that would have outweighed doing the single transaction with the Madison Dearborn?

#### **Theodore Crumley**

Well, I think again without getting into details, we had a number of alternatives available to us. I think it's fair to say that we had through those alternatives a good idea of the range of values that might be available to us for the Timberland, and then all in the matters in Dearborn transaction with a better alternative for shareholders.

# Alan Hunt - Provident Investment - Analyst

Sure. All right thank you for taking my call.

#### Operator

You have a follow-up question from the line of Tom Yanko (ph) with Morgan Stanley.

## Tom Yanko - Morgan Stanley - Analyst

Yes, on that point that you just mentioned, if something would've happened to Madison bid Dearborn deal, I assume you're still committed to re-exploring this whole process, and a streamline in the Company and then you'd go back to exploring the type of alternatives that you've just finished exploring. I just wonder if you just comment on what would happen in your event that the deal with Madison Dearborn terminated for any reason?

# George Harad - Boise Cascade Corporation - Chairman & CEO

Well, we — obviously that's not an alternative that we anticipate, but I think that we're committed to a plan that maximizes the shareholder value, and that would include eventually the separation of the two businesses.

## Tom Yanko - Morgan Stanley - Analyst

Okay. Thank you.

# Operator

Your next question is a follow-up question from Nathaneal August with Kay Capital.

## Nathaneal August - Kay Capital - Analyst

Hi, hello to follow up on the pension expense. How much pension expense was included as an allocation to the office product side in the \$210m to \$240m of operating profit guidance that you given?

23

# **Theodore Crumley**

I don't know how to answer that question.

# George Harad - Boise Cascade Corporation - Chairman & CEO

I don't have it off the top of my head for the year 2004, and so it's probably in the low-teens, it maybe \$11m to \$13m.

# Nathaneal August - Kay Capital - Analyst

Okay, and then going forward that will be reduced by 80%, call it?

# George Harad - Boise Cascade Corporation - Chairman & CEO

I would think so. I would think, we will be able to — again it depends on the performance of the plan assets and so forth, but it would be reduced — should be reduced dramatically, given the funding position that we plan to have — the plans later this year.

## Nathaneal August - Kay Capital - Analyst

Okay, great. Thank you.

#### Operator

There are no further question at this time. Do you have any closing remarks, sir?

## George Harad - Boise Cascade Corporation - Chairman & CEO

Well, only to again thank people for their time and attention and to tell you that we are excited about our ability to complete successfully the review, strategic alternatives, and to move forward with OfficeMax as a world-class competitor once this transaction is completed. Thank you.

#### Operator

Thank you for participating in today's Boise conference call. You may now disconnect.

## DISCLAIMER

Thomson Financial reserves the right to make changes to documents, content, or other information on this web site without obligation to notify any person of such changes.

In the conference calls upon which Event Transcripts are based, companies may make projections or other forward-looking statements regarding a variety of items. Such forward-looking statements are based upon current expectations and involve risks and uncertainties. Actual results may differ materially from those stated in any forward-looking statement based on a number of important factors and risks, which are more specifically identified in the companies' most recent SEC filings. Although the companies may indicate and believe that the assumptions underlying the forward-looking statements are reasonable, any of the assumptions could prove inaccurate or incorrect and, therefore, there can be no assurance that the results contemplated in the forward-looking statements will be realized.

THE INFORMATION CONTAINED IN EVENT TRANSCRIPTS IS A TEXTUAL REPRESENTATION OF THE APPLICABLE COMPANY'S CONFERENCE CALL AND WHILE EFFORTS ARE MADE TO PROVIDE AN ACCURATE TRANSCRIPTION, THERE MAY BE MATERIAL ERRORS, OMISSIONS, OR INACCURACIES IN THE REPORTING OF THE SUBSTANCE OF THE CONFERENCE CALLS. IN NO WAY DOES THOMSON FINANCIAL OR THE APPLICABLE COMPANY OR THE APPLICABLE COMPANY ASSUME ANY RESPONSIBILITY FOR ANY INVESTMENT OR OTHER DECISIONS MADE BASED UPON THE INFORMATION PROVIDED ON THIS WEB SITE OR IN ANY EVENT TRANSCRIPT. USERS ARE ADVISED TO REVIEW THE APPLICABLE COMPANY'S CONFERENCE CALL ITSELF AND THE APPLICABLE COMPANY'S SEC FILINGS BEFORE MAKING ANY INVESTMENT OR OTHER DECISIONS.

 $\ensuremath{\mathbb{C}}$  2004, Thomson StreetEvents All Rights Reserved.